SECURITIES AND EXCHANGE COMMISSION Washington, D.C. 20549

FORM 10-Q

[X] QUARTERLY REPORT PURSUANT TO SECTION 13 OR 15 (d) OF THE SECURITIES EXCHANGE ACT OF 1934

For the quarterly period ended May 26, 2001

[] TRANSITION REPORT PURSUANT TO SECTION 13 OR 15 (d) OF THE SECURITIES EXCHANGE ACT OF 1934

For the transition period from ___

Commission file no. 1-11107



FRANKLIN COVEY CO. Incorporated pursuant to the Laws of the State of Utah

Internal Revenue Service - Employer Identification No. 87-0401551

2200 West Parkway Boulevard, Salt Lake City, Utah 84119 (801) 817-1776

Indicate by check mark whether the registrant (1) has filed all reports required to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 during the preceding 12 months (or for such shorter period that the registrant was required to file such reports), and (2) has been subject to such filing requirements for the past $90\ days$.

Yes [X] No []

The total number of shares of the registrant's Common Stock outstanding on July 3 was 19,856,245

PART I. FINANCIAL INFORMATION ITEM 1. FINANCIAL STATEMENTS

FRANKLIN COVEY CO.

CONSOLIDATED CONDENSED BALANCE SHEETS

(in thousands, except share and per share amounts) May 26,

August 31,

	2001	2000
	(unau	dited)
ASSETS		
Current assets: Cash and cash equivalents Accounts receivable, less allowance for doubtful	\$ 14,303	\$ 21,242
accounts of \$2,392 and \$3,350, respectively	31,860	84,747
Inventories Income taxes receivable	53,106 7,259	53,599
Other current assets	30,964	33,447
Total current assets	137,492	
Property and equipment, net	123,898	121,556
Goodwill and other intangible assets, net Investment in unconsolidated subsidiary	229,890 18,344	258,475
Other long-term assets	21,382	19,413
	\$ 531,006 ======	
LIABILITIES AND SHAREHOLDERS' EQUITY		
Current liabilities:	r 6.063	¢ 17.004
Lines of credit Accounts payable	\$ 6,962 16,162	28,251
Income taxes payable Current portion of long-term debt and capital lease obligations	F F70	4,645
Other current liabilities	56,921	7,413 61,504
Total current liabilities	85,623	119,697
		,
Line of credit Long-term debt and capital lease obligations, less current portion	60,000 3.648	55,000 7.505
Deferred compensation	3,648 3,740	7,505 3,285
Deferred income taxes	32,939	32,939
Total liabilities	185,950	218,426
Shareholders' equity:		
Preferred stock - Series A, no par value; convertible into common stock at \$14 per share; 4,000,000 shares authorized, 811,088		
shares issued	80,967	80,967
Common stock, \$0.05 par value; 40,000,000 shares authorized, 27,055,894 shares issued	1,353	1,353
Additional paid-in capital	223,895	225,748
Retained earnings	166,567	186,711
Notes and interest receivable from sales of common stock Restricted stock deferred compensation	(2,330)	(894) (58)

Total shareholders' equity 345,056 37	=======
Treasury stock at cost, 7,223,248 and 6,439,329 shares, respectively (124,445) (12.4.445)	\$ 592,479
	374,053
	(122) (119,652)

See Notes to Consolidated Condensed Financial Statements.

FRANKLIN COVEY CO.

CONSOLIDATED CONDENSED STATEMENTS OF OPERATIONS

(in thousands, except share and per data)

	Quarte	r Ended	Nine Months Ended			
	May 26, 2001	May 27, 2000	May 26, 2001	May 27, 2000		
	(unau		(unaı			
Sales	\$ 90,610	\$ 113,732	\$ 353,098	\$ 411,976		
Cost of sales (exclusive of stock option purchase costs of \$978 and \$2,113, respectively, in fiscal 2000)	40,939	56,022	151,406	186,113		
Gross margin	49,671	57,710	201,692	225,863		
Selling, general and administrative (exclusive of stock option purchase and relocation costs of \$7,785 and \$8,808, respectively, in fiscal 2000) Stock option purchase and relocation costs Restructuring costs Depreciation Amortization		65,399 8,763 (402) 5,655 5,387		10,921		
Loss from operations	(25,541)	(27,092)	(23,094)	(5,912)		
Equity in earnings of unconsolidated subsidial Interest income Interest expense Other income		296 (1,301) 396	2.042	020		
Loss before income taxes	(26,522)	(27,701)	(25,401)	(9,177)		
Benefit for income taxes	(11,935)	(8,867)	(11,340)	(350)		
Net loss	(14,587)					
Preferred stock dividends	2,028	2,028	6,083	5,978		
Net loss attributable to common shareholders	\$ (16,615) ======	\$ (20,862) ======	\$ (20,144) ======	\$ (14,805) ======		
Net loss per share: Basic and diluted	\$ (.84) ======	\$ (1.02) =======	\$ (.99) ======	\$ (.73) ======		
Weighted average number of common and common equivalent shares: Basic and diluted	19,872 ======	20,413	20,323 ======	20,377 ======		

See Notes to Consolidated Condensed Financial Statements.

FRANKLIN COVEY CO.

CONSOLIDATED CONDENSED STATEMENTS OF CASH FLOWS

(in thousands)

	Nine Mont	hs Ended
	May 26, 2001	
	(unaud	ited)
Cash flows from operating activities: Net loss Adjustments to reconcile net loss to net cash provided by operating activities:	\$ (14,061)	\$ (8,827)
Depreciation and amortization Equity in earnings of unconsolidated subsidiary	34,643 (2,042)	34,223
Cash distribution of earnings from unconsolidated subsidiary Other Changes in assets and liabilities, net of effects from acquisitions:	1,874 822	11
Decrease in accounts receivable Decrease (increase) in inventories	52,578 371	45,520 (1,739)
Decrease (increase) in other assets and deferred compensation liability, net Decrease in accounts payable and accrued liabilities Increase (decrease) in income taxes payable	(14,919) (11,903)	(2,760) (5,984) 745
Net cash provided by operating activities		61,189
Cash flows from investing activities: Formation of joint venture, acquisition of businesses, and earnout payments Proceeds from sale of commercial printing division assets	(4,432)	(20,853) 6,406
Purchases of property and equipment	(23,563)	
Net cash used for investing activities	(27,995)	(27, 324)
Cash flows from financing activities: Net decrease in short-term borrowings	(10,922)	(1,396)

Proceeds from long-term debt and line of credit Payments on long-term debt and capital lease obligations Proceeds from issuance of preferred stock, net Payment of preferred dividends Purchases of common stock for treasury Payment of interest on management loan program Proceeds from treasury stock issuance Net cash used for financing activities	23,040 (23,731) (6,083) (7,332) (1,435) 889 (25,574)	71,244 (108,667) 4,092 (3,950) (5,308) 5,877
Effect of foreign exchange rates	(830)	114
Net decrease in cash and cash equivalents	(6,939)	(4,129)
Cash and cash equivalents at beginning of period	21,242	26,781
Cash and cash equivalents at end of period	\$ 14,303 =======	\$ 22,652 ======
Supplemental disclosure of cash flow information: Interest paid	\$ 4,263 =======	\$ 5,488 =======
Income taxes paid	\$ 643 ======	\$ 5,422 ======
Fair value of assets acquired Cash paid for net assets	\$ 4,432 (4,432)	\$ 20,853 (20,853)
Liabilities assumed from acquisitions	\$ - ======	\$ - ======
Non-cash investing and financing activities: Net assets contributed to form joint venture, net of cash contributed Accrued preferred dividends Preferred dividends paid with additional shares of preferred stock Notes receivable issued from sale of common stock Notes payable issued for the acquisition of business	\$ 18,176 2,028	\$ 2,028 1,875 894 6,000

See Notes to Consolidated Condensed Financial Statements.

FRANKLIN COVEY CO.

NOTES TO CONSOLIDATED CONDENSED FINANCIAL STATEMENTS

(unaudited)

NOTE 1 - BASIS OF PRESENTATION

Franklin Covey Co. (the "Company") provides integrated training and performance solutions to organizations and individuals in productivity, leadership, sales performance, communication, and other areas. Each integrated solution may include components for training and consulting, assessment, and other application tools that are generally available in electronic or paper-based formats. The Company's products and services are available through professional consulting services, public workshops, catalogs, retail stores, and the Internet at www.franklincovey.com and <a href="https://www.franklincovey

The accompanying unaudited consolidated condensed financial statements reflect, in the opinion of management, all adjustments (which include only normal recurring adjustments) necessary to present fairly the financial position and results of operations of the Company as of the dates and for the periods indicated. Certain information and footnote disclosures normally included in financial statements prepared in accordance with accounting principles generally accepted in the United States have been condensed or omitted pursuant to Securities and Exchange Commission ("SEC") rules and regulations. The Company suggests that the information included in this Report on Form 10-Q be read in conjunction with the financial statements and related notes included in the Company's Annual Report to Shareholders on Form 10-K for the fiscal year ended August 31, 2000.

The Company utilizes a modified 52/53 week fiscal year that ends on August 31. Corresponding quarterly periods generally consist of 13-week periods that ended on November 25, 2000, February 24, 2001, and May 26, 2001 during fiscal 2001. Under this modified 52/53 week reporting format, the quarter ended May 26, 2001 had the same number of business days as the quarter ended May 27, 2000, however, the nine months ended May 26, 2001 had two fewer business days than the corresponding period of the prior year.

The results of operations for the quarter and nine months ended May 26, 2001 are not necessarily indicative of the results for the entire fiscal year ending August 31, 2001.

During September 2000, the Emerging Issues Task Force ("EITF") of the Financial Accounting Standards Board ("FASB") released Issue No. 00-10, "Accounting for Shipping and Handling Fees and Costs." This standard requires that all amounts billed to a customer in a sale transaction related to shipping and handling be classified as sales. Previously, the Company recorded amounts billed to customers for shipping and handling as a component of cost of sales to offset the corresponding shipping and handling expense. Based upon the EITF release, amounts charged to customers for shipping and handling have been reclassified as sales in the accompanying condensed consolidated statements of income for all periods presented.

NOTE 2 – INVENTORIES

Inventories were comprised of the following (in thousands):

	May 26, 2001	2000			
Finished goods Work in process Raw materials	\$ 36,250 2,421 14,435	\$ 38,363 2,803 12,433			
	\$ 53,106	\$ 53,599			
	========	=========			

NOTE 3 - RESTRUCTURING COSTS

During the fourth quarter of fiscal 1999, the Company's Board of Directors approved a plan to restructure the Company's operations, reduce its workforce, and formally exit the majority of its leased office space located in Provo, Utah. These changes were intended to align the Company's products, services, and distribution channels in a manner that focuses Company resources on providing integrated training and performance solutions to organizations and individuals. The restructuring was also intended to lay strategic, operational, organizational, and financial foundations for profitable growth. Included in the restructuring charge were costs to provide severance and related benefits as well as costs to formally exit the leased office space. As of August 31, 2000, the Company's restructuring plan was substantially complete. The components of the accrued restructuring charge and the remaining accrual balances at May 26, 2001 were as follows (in thousands):

	Severance Costs	Leased Office Space Exit Costs	Total
Accrued restructuring costs at August 31, 2000	\$ 2,415	\$ 2,745	\$ 5,160
Restructuring costs paid	(1,789)	(446)	(2,235)
Accrued restructuring costs as of May 26, 2001	\$ 626 ======	\$ 2,299 =======	\$ 2,925 ======

As of May 26, 2001, accrued severance costs consisted of expected remaining severance and benefit payments for terminated employees. Remaining accrued leased office space exit costs represent the difference between base rental charges and the offsetting expected sublease revenue receipts. The Company expects that the remaining restructuring accrual will be sufficient to complete its restructuring plan.

NOTE 4 – SHAREHOLDERS' EQUITY

At November 25, 2000, the Company's Board of Directors had approved various plans for the purchase of up to 8,000,000 shares of the Company's common stock. Through November 25, 2000, the Company had purchased 7,705,000 shares under these Board authorized purchase plans.

On December 1, 2000, the Company's Board of Directors approved an additional plan to purchase up to \$10.0 million of the Company's common stock. Through May 26, 2001, the Company had purchased a total of 888,000 shares for \$7.1 million under the terms of this plan. In addition, the Company purchased 24,000 shares of its common stock with a cost of \$0.2 million for exclusive distribution to participants enrolled in the Employee Stock Purchase Plan.

NOTE 5 - COMPREHENSIVE INCOME (LOSS)

Comprehensive income (loss) includes net income and other revenues, expenses, gains, and losses that are excluded from net income (loss) but are included as components of shareholders' equity. Comprehensive income (loss) for the Company was as follows (in thousands):

	Quarte	r Ended	Nine Months Ended				
	May 26, 2001	May 27, 2000	May 26, 2001	May 27, 2000			
Net loss attributable to common shareholders	\$ (16,615)	\$ (20,862)	\$ (20,144)	\$ (14,805)			
Other comprehensive loss: Foreign currency translation adjustments	(148)	(409)	(829)	(108)			
Comprehensive loss	\$ (16,763) ======	\$ (21,271) ======	\$ (20,973) ======	\$ (14,913) =======			

NOTE 6 - NET LOSS PER COMMON SHARE

Basic earnings per share ("EPS") is calculated by dividing net income (loss) attributable to common shareholders by the weighted-average number of common shares outstanding for the period. Diluted EPS is calculated by dividing net income (loss) by the weighted-average number of common shares outstanding plus the assumed exercise of all dilutive securities using the treasury stock or the "as converted" method as appropriate. The diluted EPS calculation for the periods presented excludes the impact of stock options and preferred stock because they are antidilutive. Significant components of the numerator and denominator used for basic and diluted EPS were as follows (in thousands, except per share amounts):

	Quarto	er Ended	Nine Months Ended				
	May 26,	May 27,	May 26,	May 27,			
	2001	2000	2001	2000			
Net loss	\$ (14,587)	\$ (18,834)	\$ (14,061)	\$ (8,827)			
Preferred dividends	2,028	2,028	6,083	5,978			
Net loss attributable to common shareholders	\$ (16,615)	\$ (20,862)	\$ (20,144)	\$ (14,805)			
	=======	======	=======	======			
Basic and diluted weighted-average shares outstanding	19,872	20,413	20,323	20,377			
	=======	======	=======	======			
Net loss per share: Basic and diluted	\$ (.84)	\$ (1.02)	\$ (.99)	\$ (.73)			
Antidilutive shares excluded from the diluted EPS calculation: Incremental shares from assumed exercises of stock options Preferred stock on an	65	94	74	90			
"as converted" basis	5,794	5,794	5,794	5,794			

NOTE 7 – SEGMENT INFORMATION

As part of its restructuring initiatives during fiscal 2000, the Company adopted a channel-based view of its operations and has aligned its business operations into the following business segments:

Retail Stores – Includes the sales and operational results of the Company's 159 retail stores. Although retail store sales mainly consist of products such as planners and handheld electronic devices, virtually any component of the Company's leadership and productivity solutions can be purchased through the retail store channel.

Catalog/eCommerce — This operating segment includes the sales and operating results of the Company's catalog operation and its Internet web-site at www.franklincovey.com. Nearly all of the Company's products and services can be purchased through these channels.

Organizational Sales Group – The organizational sales group ("OSG") is primarily responsible for the sale and delivery of leadership, productivity, sales performance, and communication training seminars to corporations and certain other organizational clients.

Educational – The educational channel includes the sales and operations of Premier, a subsidiary that focuses on productivity and effectiveness tools for students, and includes sales of both products and training to educational institutions from elementary schools to colleges and universities. Operating results of this channel are primarily dependent upon the seasonal sales pattern of Premier, which recognizes the majority of its sales during the Company's fourth fiscal quarter.

International – The international channel is responsible for the delivery of products and services to clients located outside of the United States.

Other – The "other" channel consists primarily of wholesale, government, personal coaching, and commercial printing operations at Publishers Press (now Franklin Covey Printing). Financial results from this channel were affected by the contribution of personal coaching assets to form Franklin Covey Coaching LLC, effective September 1, 2000, and the sale of commercial printing operations at Franklin Covey Printing, which occurred during fiscal 2000.

The Company's chief operating decision maker is the Chief Executive Officer ("CEO"). Each of the reportable segments and corporate support departments has an executive vice-president who reports directly to the CEO. The primary measurement tool in segment performance analysis is earnings before interest, taxes, depreciation, and amortization ("EBITDA"), which also approximates cash flows from the operating segments and may not be calculated as similarly titled amounts presented by other companies. The calculation of EBITDA includes the effects of stock option purchase and relocation costs and restructuring costs in fiscal 2000 as well as the equity in earnings of Franklin Covey Coaching, LLC, a newly formed joint venture that began operations on September 1, 2000.

The Company accounts for its segment information on the same basis as the accompanying consolidated condensed financial statements. Prior year information has been restated to conform with current year classifications.

SEGMENT INFORMATION (in thousands)

ent
E

Quarter ended May 26, 2001	Retail Stores	Catalog/ eCommerce	0SG	Educational	International	Other	Adjus	orate, tments and inations	Con	solidated
Sales to external customers	\$ 26,952	\$ 15,184	\$ 19,458	\$ 3,808	\$ 11,256	\$ 13,952	•	(2.690)	\$	90,610
Intersegment sales Gross margin	12,441	8,636	13,253	2,319	7,334	3,689 7,115	\$	(3,689) (1,427)		49,671
EBITDA	(1,586)	2,726	3,396	(5,776)	679	(2,533)		(9,723)		(12,817)
Depreciation and amortization	2,559	213	1,096	1,441	362	939		5,507		12,117

Quarter ended May 27, 2000									
Sales to external customers Intersegment sales	\$ 33,	480	\$ 20,613	\$ 21,979	\$ 3,495	\$ 10,113	\$ 24,052 6,731	\$ (6,731)	\$ 113,732
Gross margin Stock option purchase and relocation costs	14,	601	7,745	14,815	2,096	6,292	11,489	672 8,763	57,710 8,763
Restructuring costs								(402)	(402)
EBITDA	2,	979	974	3,953	(5,087)	(1,345)	(1,682)	(15,842)	(16,050)
Depreciation and	,			,	(-,,	(, ,	(, ,	(-, - ,	(,,,,,,,
amortization	1,	630	159	837	1,366	397	1,150	5,503	11,042
Nine months ended May 26, 2001									
Sales to external customers	\$ 121,	363	\$ 75,115	\$ 59,557	\$ 11,731	\$ 40,505	\$ 44,827		\$ 353,098
Intersegment sales	•		,	•	,	•	13,309	\$ (13,309)	•
Gross margin	62,		42,197	41,188	7,115	26,768	23,631	(1,926)	201,692
EBITDA	22,	203	23,476	11,064	(16,849)	4,642	(4,368)	(27,848)	12,320
Depreciation and	-	004	340	0.700	4 054	4 465	0.070	45 604	00 070
amortization Segment assets	31,	824 148	471	2,793 15,416	4,254 76,531	1,165 24,554	3,372 63,134	15,624 319,752	33,372 531,006
Nine months ended May 27, 2000									
Sales to external customers	\$ 126,	390	\$ 87,593	\$ 63,125	\$ 12,123	\$ 39,331	\$ 83,414		\$ 411,976
Intersegment sales			,	•	,	•	19,397	\$ (19,397)	•
Gross margin Stock option purchase	68,	087	44,134	42,440	7,079	25,205	36,571	2,347	225,863
and relocation costs Restructuring costs								10,921 (402)	10,921 (402)
EBITDA Depreciation and	34,	297	21,885	12,137	(13,821)	3,256	(2,520)	(29,685)	25,549
amortization		982	296	2,442	3,911	1,102	3,798	14,930	31,461
Segment assets	24,	442	414	16,189	73,340	23,963	73,504	347,054	558,906

A reconciliation of reportable segment EBITDA to consolidated EBITDA is presented below (in thousands):

	Quarter Ended					Nine Months Ended				
	May 26, 2001		May 27, 2000		May 26, 2001			y 27, 2000		
Reportable segment EBITDA Corporate expenses Stock option purchase and relocation expenses Restructuring costs Intercompany rent charges Other	(Ì1	3,034) 1,149) 1,574 (208)	\$	(208) (9,086) (8,763) 402 1,711 (106)	\$	40,168 (30,584) 4,530 (1,794)	\$	55,237 (24,507) (10,921) 402 5,133 205		
Consolidated EBITDA	\$ (12	2,817)	\$	(16,050)	\$	12,320	\$	25,549 =======		

Corporate assets such as cash, accounts receivable, and other assets are not generally allocated to reportable business segments for business analysis purposes. However, inventories, certain identifiable intangible assets, and fixed assets are classified by segment.

NOTE 8 – PERSONAL COACHING JOINT VENTURE

Effective September 1, 2000, the Company entered into a joint venture with American Marketing Systems, Inc. ("AMS"), a major customer of the Company's Personal Coaching division. The new company, Franklin Covey Coaching, LLC, will continue to provide personal coaching services for the Company's customers. Under terms of the agreement, the Company and AMS each own 50 percent of Franklin Covey Coaching, LLC and are equally represented in the management of the new company. The Company expects that the new venture will broaden the curriculum and services currently offered in order to grow the personal coaching business over the long term, while maintaining a substantial portion of the Company's current earnings from coaching activities. The Company contributed substantially all of the net assets of the Personal Coaching division to form the new entity. The contribution of net assets to the joint venture produced a contribution basis difference totaling \$9.1 million, which will be amortized using the straight-line method over a 20-year period. The Company accounts for the joint venture using the equity method of accounting and reported its share of the joint venture's earnings as "equity in earnings of unconsolidated subsidiary" in the accompanying consolidated condensed statements of operations for fiscal 2001.

NOTE 9 – ACCOUNTING FOR DERIVATIVE INSTRUMENTS

Effective September 1, 2000, the Company adopted Statement of Financial Accounting Standards ("SFAS") No. 133, "Accounting for Derivative Instruments and Hedging Activities", as amended by SFAS No. 138. The new standard requires that all derivative instruments be recorded on the balance sheet as either an asset or liability measured at fair value, and that changes in the derivative's fair value be recognized as a component of earnings from current operations unless specific hedge criteria are met. The cumulative effect of adopting SFAS No. 133 was not material to the Company's financial statements.

The Company utilizes certain derivative financial instruments to enhance its ability to manage risk, primarily from foreign currency exposures that exist as part of the Company's ongoing business operations. Derivative instruments are entered into for periods consistent with related underlying exposures and do not constitute positions that are independent of those exposures. In addition, the Company does not enter into derivative contracts for speculative purposes, nor is the Company party to any leveraged derivative instrument.

During the normal course of operations, the Company is exposed to foreign currency exchange risks as a result of transactions that are denominated in currencies other than the United States dollar. As of May 26, 2001, the Company utilized a foreign currency forward contract to manage the volatility of certain intercompany financing transactions that are denominated in Japanese Yen. This contract did not meet certain hedge accounting requirements and corresponding gains and losses have been recorded as a component of current operations, which offset gains and losses have been recorded as a component of currency forward contract was approximately \$6.0 million at May 26, 2001.

NOTE 10 – SUBSEQUENT EVENTS

Subsequent to May 26, 2001, the Company entered into a new credit agreement with its lenders. The new credit agreement is comprised of a \$69.0 million term loan, payable in quarterly installments over three years, and a \$45.0 million revolving credit facility, which expires in July 2004. Included in the \$69.0 million term loan, is \$33.6 million due from the participants of the management common stock loan program, which was previously guaranteed by the Company. Under terms of the new credit agreement, the Company will now be the primary lender of these non-recourse notes from the participants of the management common stock loan program. Accordingly, the Company will recognize a corresponding receivable in the shareholders' equity section of its consolidated balance sheet. The new credit agreement is secured by receivables, inventory, and certain real estate, and requires the Company to maintain certain financial ratios and minimum net worth levels. As of May 26, 2001, the Company was in compliance with the terms of the new agreement.

As part of the new credit agreement, the Company paid \$1.4 million to the lender for accrued interest on the management common stock loan program. The amount will be repaid to the Company from the loan participants when the loans mature in March 2005. Accordingly, this transaction was recorded as a component of notes and interest receivable from sales of common stock in the shareholders' equity section of the accompanying condensed consolidated balance sheet as of May 26, 2001.

The Company entered into an outsourcing agreement with Electronic Data Systems ("EDS") to provide warehousing, distribution, information systems, and call center operations. Under terms of the agreement, EDS will operate the Company's primary call center, provide warehousing and distribution services, and support the Company's information systems. The Company believes that the agreement will reduce operating costs and improve asset utilization. In a related transaction, the Company also entered into an agreement to sell its warehouse and distribution facilities in Salt Lake City, Utah. The sale of these facilities is expected to generate a nominal gain to the Company.

ITEM 2. MANAGEMENT'S DISCUSSION AND ANALYSIS OF FINANCIAL CONDITION AND RESULTS OF OPERATIONS

The following discussion and analysis should be read in conjunction with the Consolidated Financial Statements, the Notes thereto and Management's Discussion and Analysis included in the Company's Annual Report to Shareholders on Form 10-K for the year ended August 31, 2000.

RESULTS OF OPERATIONS

Net loss attributable to common shareholders for the quarter ended May 26, 2001 was \$16.6 million compared to a loss of \$20.9 million in the prior year. The loss in fiscal 2000 included \$8.4 million of non-recurring charges for stock option purchase, relocation, and restructuring costs. Reported results for the third quarter of fiscal 2001 were adversely affected by decreased comparable retail store sales, a decline in corporate on-site training sales, and decreased demand for certain products sold through the Company's wholesale channel. The Company also attributes declining performance to prevailing economic conditions in the United States that have continued since late 2000. These issues, combined with factors described in the Company's Reports on Form 10-Q for previous quarters in fiscal 2001, have adversely affected overall performance compared to the prior year.

The following table sets forth selected data concerning sales of the Company's operating segments (dollars in thousands):

		Q 	uarter	Ended		Nine Months Ended					
	May 26, 2001				Var. %	M	May 26, 2001		May 27, 2000	Var. %	
	(unaudited)					(unaudited)					
Retail stores Catalog/eCommerce OSG Educational International Other	\$	26,952 15,184 19,458 3,808 11,256 13,952	\$	33,480 20,613 21,979 3,495 10,113 24,052	(19) (26) (11) 9 11 (42)	\$	121,363 75,115 59,557 11,731 40,505 44,827	\$	126,390 87,593 63,125 12,123 39,331 83,414	(4) (14) (6) (3) 3 (46)	
	\$	90,610	\$ ==	113,732	(20)	\$	353,098	\$	411,976	(14)	

Quarter Ended May 26, 2001 Compared with the Quarter Ended May 27, 2000

Sales

Retail store sales decreased compared to the prior year primarily due to a 28 percent decline in comparable store sales. Additional sales volume from 31 new stores was insufficient to offset unfavorable sales performance from existing stores. At May 26, 2001, the Company was operating 159 retail stores compared to 128 stores at May 27, 2000. The decline in comparable store sales was primarily due to reduced consumer traffic in the stores during the quarter and strong sales in the prior year due to the introduction and market acceptance of various PalmTM handheld products. The Company attributed decreased store traffic to a general softening in retail sales due to economic conditions that have continued in the United States since late 2000, the delayed introduction of a new line of Palm products, and canibalization of existing store sales by newly opened stores. Catalog/eCommerce sales decreased primarily due to reduced call volume in the Company's catalog operations. However, sales of products and training through the Company's web site at www.franklincovey.com continued to increase compared to the prior year. In an effort to revialize product sales through its retail and catalog/eCommerce channels, the Company has introduced new electronic devices, such as the VisorTM series of handheld organizers from HandspringTM, wireless communications service, desktop organization tools, such as Franklin Covey OnePlaceTM, and expanded its line of paper-based planners. Sales through the Company's Organizational Sales Group ("OSG") declined during the quarter primarily due to reduced on-site corporate leadership, productivity, and communications training seminars. The Company attributes the decline in corporate seminars to general economic conditions, which appear to have adversely affected corporate training budgets and resulted in canceled seminar presentations during the quarter. Educational sales, which include Premier, increased primarily due to improved commercial pr

Gross Marain

In general, the Company's gross margin as a percent of sales was affected by the reclassification of shipping and handling revenue from cost of sales to sales as required by EITF 00-10. All periods presented in the accompanying condensed consolidated statements of operations have been restated according to the requirements of EITF 00-10. Gross margin for the Company was 54.8 percent of sales for the quarter, compared to 50.7 percent in the prior year. The Company's gross margin improved primarily due to changes in the Company's sales mix, decreased obsolete inventory write-offs resulting from improved forecasting and production procedures, and price increases on certain planner products. Sales of handheld electronic devices, which have lower margins than the majority of the Company's other products and services, declined as a percent of total sales compared to the prior year. Partially offsetting these factors was the formation of Franklin Covey Coaching LLC, a joint venture formed to provide coaching services, which reduced the Company's overall gross margin due to financial reporting requirements that exclude the favorable gross margins of the personal coaching between the company of the personal coaching between the coaching

Operating Expenses

Selling, general and administrative ("SG&A") expenses decreased by \$2.3 million, net of stock option purchase and relocation costs in fiscal 2000, but due to decreased sales volume, increased to 69.6 percent of sales, compared to 57.5 percent in the prior year. The decrease was primarily due to the formation of Franklin Covey Personal Coaching LLC, reduced associate expenses, decreased catalog promotional costs, reduced international operating expenses, and expense reduction efforts in various operating areas of the Company. Partially offsetting these decreases, was increased retail store operating costs, ongoing development and marketing of new products and electronic commerce channels, and increased consulting costs. Due to accounting guidelines associated with the formation of the personal coaching joint venture, the Company no longer includes the operating costs of the Personal Coaching division in its consolidated results. In order to improve the profitability of its catalog operation, the Company reduced catalog production and mailing costs and eliminated certain promotional activities that produced only marginal value. The Company is currently developing a comprehensive marketing plan, which is expected to produce better customer focus for its catalogs and other channels, and promotional materials and may incur increased costs in future periods to achieve its goals. Due to decreased sales during the quarter, the Company also implemented various expense reduction initiatives to improve operating performance and profitability. Partially offsetting these efforts were the additional operating costs resulting from 31 new retail stores. In addition, the Company has also incurred and expensed pre-opening costs during the quarter associated with the new stores. The Company continues to increase to develop and market new products and services, such as the new series of training courses entitled "Productivity in the Digital Age" and the desktop organizational tool, Franklin Covey OnePlace. The Company also continu

Subsequent to May 26, 2001, the Company entered into an outsourcing agreement with Electronic Data Systems ("EDS") to provide warehousing, distribution, information systems, and call center operations. Under terms of the agreement, EDS will operate the Company's primary call center, provide warehousing and distribution services, and support the Company's information systems. The Company believes that the agreement with EDS will reduce operating costs for these activities.

Depreciation charges increased by \$1.6 million compared the prior year, primarily due to the addition of leasehold improvements and fixtures in new and recently remodeled retail stores, the purchase of computer hardware and software, and the purchase of new manufacturing equipment. Amortization charges decreased by \$0.6 million, primarily due to the contribution of Personal Coaching intangible assets to Franklin Covey Coaching LLC, a newly formed joint venture that is recorded on the Company's books as an unconsolidated subsidiary under the guidelines of equity method accounting.

Income tax benefit was recorded at an effective rate of 45.0 percent for the quarter ended May 26, 2001. The effective tax rate was based upon expected taxable income for the remainder of fiscal 2001 and the effects of non-deductible goodwill amortization. Non-deductible goodwill amortization from previous acquisitions and related contingent earnout payments generally has an unfavorable impact on the Company's effective tax rate.

Nine Months Ended May 26, 2001 Compared to the Nine Months Ended May 27, 2000

Sales

The decrease in retail store sales was primarily due to reduced consumer traffic in the stores and strong sales during the prior year due to demand for PalmTM handheld electronic products and related accessories. The Company attributed decreased traffic to a general softening in retail sales due to economic conditions that began in late 2000 and which have continued during 2001, cannibalization of existing store sales by newly opened stores, and fewer business days than the comparable period of the prior year. These factors combined to produce a 14 percent decrease in comparable store sales, which was partially offset by sales from new stores. Catalog/eCommerce sales decreased primarily due to reduced call volume in the Company's catalog operations. However, sales through the Company's web site at www.franklincovey.com/continued to increase compared to the prior year. Sales through OSG decreased primarily due to reduced corporate on-site productivity, leadership, and communications training. The Company believes that current prevailing economic conditions adversely affected sales performance in its catalog/eCommerce operations and OSG as corporations reduce training and related spending. Educational sales, which include Premier, decreased due to the timing of agenda shipments in fiscal 2000. The Company was unable to ship all of the 1999/2000 agenda order during the fourth quarter of fiscal 1999. As a result, some of those agendas were shipped during the first quarter of fiscal 2000. However, the Company was able to ship nearly all of its 2000/2001 agenda order in the fourth quarter of fiscal 2000, which minimized spill-over sales in fiscal 2001. International sales increases in Mexico, Europe, Brazil, and licensee operations were partially offset by sales decreases in Japan, Canada, Australia, and New Zealand. In addition, unfavorable foreign currency exchange rates compared to the prior year had an adverse effect on translated sales for the nine months e

Gross Margin

Gross margin improved to 57.1 percent of sales, compared to 54.8 percent in the prior year. The Company's gross margin improved primarily due to the sale of the commercial division of Publishers Press, decreased obsolete inventory write-offs resulting from improved forecasting and production procedures, sales mix changes, and price increases on certain planner products and seminar programs. Commercial printing sales from Publishers Press had significantly lower gross margins than the majority of the Company's other products, and services, and was sold effective February 28, 2000. Although sales of handheld electronic devices, which have lower margins than the majority of the Company's other products, have increased compared to the prior year, sales of higher margin accessory and other products have increased during fiscal 2001. Partially offsetting these factors was the formation of Franklin Covey Coaching LLC, which reduced the Company's overall gross margin due to financial reporting requirements that exclude the favorable gross margins of the personal coaching business.

Operating Expenses

Selling, general and administrative ("SG&A") expenses increased \$1.6 million, net of stock option purchase and relocation costs in fiscal 2000, to 54.2 percent of sales, compared to 46.1 percent in fiscal 2000. The increase was primarily due to increased retail store operating costs, ongoing development and marketing of new products and electronic commerce channels, a compensation plan change in one of the Company's operating divisions, and increased consulting costs. These increases were partially offset by reductions in SG&A expenses resulting from the formation of the personal coaching joint venture, reduced catalog promotional costs, and decreased corporate overhead costs resulting from corporate expense reduction initiatives and reduced headcount. At May 26, 2001, the Company was operating 31 additional retail stores compared to the prior year, for which incremental sales performance was not strong enough to offset the

corresponding increase in operating costs. In addition, the Company has incurred and expensed pre-opening costs associated with theses new stores. The Company also continues to incur costs to develop and market new products and services, and to improve its electronic commerce infrastructure to meet consumer preferences. A compensation plan change related to sales commissions at one of the Company's operating divisions changed the timing of commission expense recognized, but will not adversely affect total commission expense for the fiscal year. As the Company positions itself for future growth and opportunities, consulting costs related to the completion of several key initiatives were incurred during the period. Partially offsetting these increases was the formation of the personal coaching joint venture. Due to accounting guidelines associated with the formation of the personal coaching joint venture, the Company no longer includes the operating costs of the Personal Coaching division in its consolidated results. In order to improve the profitability of its catalog operation, the Company reduced catalog production and mailing costs. The Company continues to review its operating cost structure and may implement additional initiatives to further reduce related operating expenses.

Depreciation charges increased by \$2.6 million over the prior year, primarily due to the addition of leasehold improvements and fixtures in new stores, the purchase of computer hardware and software, and the purchase of new manufacturing equipment. Amortization charges decreased by \$0.7 million, primarily due to the contribution of personal coaching intangible assets to the newly formed Franklin Covey Coaching LLC joint venture.

Income tax benefit was recorded at an effective rate of 44.6 percent, which was based upon expected taxable income for the remainder of fiscal 2001 and the effects of non-deductible goodwill amortization. Non-deductible goodwill amortization from previous acquisitions and related contingent earnout payments generally has an unfavorable impact on the Company's effective tax rate. In addition, the Company's effective income tax rate may be adjusted during the remainder of fiscal 2001, if necessary, to reflect changes in expected taxable income.

LIQUIDITY AND CAPITAL RESOURCES

Historically, the Company's primary sources of capital have been net cash provided by operating activities, long-term borrowings and line of credit financing. Working capital requirements have also been financed through short-term borrowing. In addition to these sources, the Company issued preferred stock for cash to a private investor and through a subscription offering to existing shareholders that closed during fiscal 2000.

Net cash provided by operating activities during the nine months ended May 26, 2001 was \$47.5 million, compared to \$61.2 million in the prior year. In general, cash flow from operating activities was primarily affected by decreased operating performance during fiscal 2001. Adjustments to net loss included \$33.6 million of depreciation and amortization charges and \$1.9 million of cash distributed from Franklin Covey Coaching LLC, an unconsolidated subsidiary of the Company. The main source of cash from operating activities was the collection of accounts review primary use of cash for operating activities was payment of accounts payable and accrued liabilities at Premier, due to normal seasonal operating activity, and at the Company's core operations.

Net cash used for investing activities totaled \$28.0 million during the first nine months of fiscal 2001 compared to \$27.3 million in fiscal 2000. Cash used to purchase property and equipment totaled \$28.6 million and was used primarily for leasehold improvements and fixtures in new retail stores, computer hardware and software, and manufacturing equipment. Cash used for contingent earnout payments declined to \$1.9 million in fiscal 2001 from \$16.3 million during fiscal 2000, primarily due to the completion of the Premier earnout agreement in fiscal 2000. The fiscal 2001 payment of \$1.9 million represents the final contingent earnout payment to the former of Personal Coaching. As of May 26, 2001, the Company had no remaining contingent earnout agreements from the acquisition of substition of substition fixed in activity during 2001 was limited primarily to the Project Consulting Group, which was acquired during the quarter ended May 26, 2001 for \$1.5 million in cash. The Company also contributed \$0.3 million of cash to form Franklin Covey Coaching LLC, a new joint venture that began operations on September 1, 2000.

Net cash used for financing activities during the first nine months of fiscal 2001 was \$25.6 million compared to \$38.1 million in the prior year. The primary source and use of financing cash was proceeds from and payments on the Company's lines of credit. Additionally, the Company paid \$6.1 million for preferred stock dividends and spent \$7.3 million to acquire 923,000 shares of its common stock.

Subsequent to May 26, 2001, the Company entered into a new credit agreement with its lenders. The new credit agreement is comprised of a \$69.0 million term loan, payable in quarterly installments over three years, and a \$45.0 million revolving credit facility, which expires in July 2004. Included in the \$69.0 million term loan, is \$33.6 million due from the participants of the management common stock loan program, which was previously guaranteed by the Company. Under terms of the new credit agreement, the Company will now be the primary lender of these non-recourse notes from the participants of the management common stock loan program. Accordingly, the Company will recognize a corresponding receivable in the shareholders' equity section of its consolidated balance sheet. The new credit agreement is secured by receivables, inventory, and certain real estate, and requires the Company to maintain certain financial ratios and minimum net worth levels. As of May 26, 2001, the Company was in compliance with the terms of the new agreement.

Going forward, the Company will continue to incur costs necessary for the current and future operations of the business. Cash provided by operations, available lines of credit and other financing alternatives will be used for these expenditures. Management believes that its existing capital resources should be sufficient to enable the Company to maintain its current level of operations and its planned internal growth for the foreseeable future. The Company also continues to pursue additional financing alternatives as it positions itself for future opportunities.

"Safe Harbor" Statement Under the Private Securities Litigation Reform Act of 1995

With the exception of historical information (information relating to the Company's financial condition and results of operations at historical dates or for historical periods), the matters discussed in this Management's Discussion and Analysis of Financial Condition and Results of Operations and elsewhere are forward-looking statements that necessarily are based on certain assumptions and are subject to risks and uncertainties. Such uncertainties include, but are not limited to, unanticipated developments in any one or more of the following areas: the integration of acquired or merged businesses, management of costs in light of decreasing revenues, availability of financing sources, dependence on products or services, the rate of consumer acceptance of new product introductions, competition, the number and nature of customers and their product orders, pricing, pending and threatened litigation, and other risk factors which may be detailed from time to time in the Company's press releases, reports to shareholders and in filings with the SEC.

While the Company has a broad customer base, it is subject to variables over which it has no direct control such as innovations in competing products, the general transition from paper-based products to electronic or Internet based products, changing corporate policies on the part of the Company's customers, and competition from others in the industry. In addition, the Company is subject to changes in costs of supplies necessary to produce its products and distribution of those products. The Company's business is subject to seasonal variations. The Company is also subject to risks associated with sales occurring in countries other than the United States. Sales outside the United States potentially present additional risks such as political, social, and economic instability.

The market price of the Company's common stock has been and may remain volatile. In addition, the stock markets in general have recently experienced increased volatility. Factors such as quarter-to-quarter variations in revenues and earnings or the failure of the Company to meet analysts' expectations could have a significant impact on the market price of the Company's common stock. In addition, the price of the common stock can change for reasons unrelated to the performance of the Company.

The forward-looking statements contained in this report are based on management's expectations as of the date hereof, and the Company does not undertake any responsibility to update any of these statements in the future. Actual future performance and results will differ and may differ materially from that contained in or suggested by these forward-looking statements as a result of the factors set forth in this Management's Discussion and Analysis of Financial Condition and Results of Operations, the business risks described in the Company's Annual Report on Form 10-K for the year ended August 31, 2000, and elsewhere in the Company's filings with the SEC.

ITEM 3. QUANTITATIVE AND QUALITATIVE DISCLOSURES ABOUT MARKET RISK

MARKET RISK OF FINANCIAL INSTRUMENTS

The principal risks to which the Company is exposed are changes in interest rates and foreign currency exchange rates. The Company manages its exposure to interest rates by optimizing the use of variable-rate and fixed-rate debt instruments. If interest rates had increased by one percent during the nine months ended May 26, 2001, the Company's total interest expense would have increased by \$0.5 million. To manage the volatility related to currency exchange rates, the Company has entered into limited derivative transactions to manage well-defined foreign currency exchange risks. Foreign exchange forward contracts are entered into for periods consistent with underlying exposures and do not constitute positions independent of those exposures. The Company does not purchase foreign exchange or other derivative instruments for speculative purposes and does not use leveraged instruments. As the Company continues to expand internationally, the Company's use of foreign exchange contracts may grow in order to manage the foreign currency risks to the Company. At May 26, 2001, the Company had one foreign exchange forward contract with a bank to sell Japanese Yen and receive United States Dollars with a notional amount of approximately \$6.0 million.

EURO CONVERSION

On January 1, 1999, the European Monetary Union ("EMU"), which is comprised of 11 out of the 15 member countries of the European Union, introduced a new common currency, the "Euro." During the transition period between January 1, 1999 and January 1, 2002, both the Euro and national currencies will coexist. The national currencies will remain legal tender until at least January 1, 2002, but not later than July 1, 2002. The Company currently transacts business in EMU countries using the national currencies and translates the financial results of those countries in accordance with current accounting pronouncements. Further, the Company has not experienced, nor does it expect to experience, a material adverse impact on its financial condition, results of operations or liquidity as a result of the Euro conversion.

PART II. OTHER INFORMATION

Item 1— Legal Proceedings:

Not applicable.

Item 2—Changes in Securities:

Not applicable

Item 3— Defaults upon Senior Securities:

Not applicable.

Item 4— Submission of Matters to a Vote of Security Holders:

Not applicable

Item 5— Other information:

Not applicable.

Item 6— Exhibits and Reports on Form 8-K:

10.1	Agreement for Information Technology Services between each of Franklin Covey Co., Electronic Data Systems Corporation and EDS Information Services LLC, dated April 1, 2001 (filed herewith).
10.2	Additional Services Addendum #1 to Agreement for Information Technology Services between each of Franklin Covey Co., Electronic Data Systems Corporation and EDS Information Services LLC, dated June 30, 2001 (filed herewith).
10.3	Amendment #2 to Agreement for Information Technology Services between each of Franklin Covey Co., Electronic Data Systems Corporation and EDS Information Services LLC, dated June 30, 2001 (filed herewith).
10.4	Interest Transfer Agreement between Bank One, NA and Franklin Covey Co., dated May 3, 2001 (filed herewith).
10.5	Fourth Amendment to Facility and Guaranty Agreement among Franklin Covey Co., Bank One, NA as Agent and the Financial Institutions Signatory Hereto, (filed herewith).

(B) Reports on Form 8-K:

(A) Exhibits:

Not applicable.

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned thereunto duly authorized.

FRANKLIN COVEY CO.

By: ROBERT A. WHITMAN

ROBERT A. WHITMAN Chief Executive Officer Dated July 10, 2001

By: STEPHEN D. YOUNG

STEPHEN D. YOUNG Senior Vice President and Controller Dated July 10, 2001

Exhibit List

- 10.1 Agreement for Information Technology Services between each of Franklin Covey Co., Electronic Data Systems Corporation and EDS Information Services LLC, dated April 1, 2001
- 10.2 Additional Services Addendum #1 to Agreement for Information Technology Services between each of Franklin Covey Co., Electronic Data Systems Corporation and EDS Information Services LLC, dated June 30, 2001
- 10.3 Amendment #2 to Agreement for Information Technology, Services, between each of Franklin Covey Co., Electronic Data Systems Corporation and EDS Information Services LLC, dated June 30, 2001
- 10.4 <u>Interest Transfer Agreement</u> between Bank One, NA and Franklin Covey Co., dated May 3, 2001
- 10.5 Fourth Amendment to Facility and Guaranty Agreement among Franklin Covey Co., Bank One, NA as Agent and the Financial Institutions Signatory Hereto

Table of Contents

AGREEMENT FOR INFORMATION TECHNOLOGY SERVICES

between each of

FRANKLIN COVEY CO.,

ELECTRONIC DATA SYSTEMS CORPORATION

and

EDS INFORMATION SERVICES L.L.C.

April 1, 2001

Article	I. Relationship Management
1.1 1.2 1.3 1.4 1.5	EDS Account Executive
	(vi) Executive Summary 2 (b) Migration Plan 2 (c) Updates 3 (d) Services 3
Article	II. Personnel
2.1	Transition of Employees
2.2	Use and Replacement of Personnel
2.3	(b) Key Personnel 5 Qualifications and Retention of EDS Personnel 5 (a) Qualifications 5 (b) Turnover Rate 5 (c) Training 5
2.4 2.5 2.6 2.7	Üse of Subsidiaries and Subcontractors .5 Hiring of Employees .6 Additional Benefits .6 No Re-Assignment to Competitor .6
Article	III. EDS Services
3.1 3.2 3.3 3.4	Term; Renewal. 6 EDS Services. 6 Services Generally; Performance Criteria. 7 Service Levels. 7 (a) Procedures. 7 (b) Reports and Remediation. 7 (i) Investigation. 7 (ii) Describe. 7 (iii) Correct the Error. 7
3.5	(iv) Report 8 (c) Meeting 8 (d) Service Credits; Other Remedies 8 (e) Technological Advancements 8 (f) Efficient Use of Resources 8 (g) Customer Surveys 8 Additional Services 8 (a) From EDS 8 (b) From Third Parties 9 (c) EDS Responsibilities to Other Service Providers 9
3.6 Article	(d) Franklin Covey Conditions to Using Third Party Services9 Reliance on Instructions
4.1	Franklin Covey's Role10
Article	V. Maintenance Contracts and Additional Property
5.1 5.2 5.3 5.4 5.5	Maintenance Contracts
Article	VI. Warranties and Additional Covenants
6.1	Warranties and Additional Covenants 11 (a) Performance 11 (b) Service Schedule Information 12 (c) Viruses 12 (d) Disabling Codes 12 (e) Pass-Through Warranties and Indemnities 12 (f) Software Development 12
6.2	Disclaimer of Warranties
Article	VII. Proprietary Rights
7.1	Definitions

	(a) EDS SOftWare. (b) EDS-Vendor Software. (c) Franklin Covey Software. (d) Franklin Covey-Vendor Software. (e) Developed Software. (f) Other Deliverables.	13 13 13 14
7.2	(g) EDS Development Tools; Residual Technology(h) Software	14
7.3 7.4	EDS-Vendor Software. Franklin Covey Software.	14
7.5 7.6	Franklin Covey-Vendor Software	15
7.7	EDS Development Tools; Residual Technology	15
7.8 Article	Further Assurances	16
8.1	Data of Franklin Covey	16
8.2	Safeguarding Data. (a) Restricted Access. (b) Security System.	16
8.3 8.4	(c) Loss of Data	16 17
0.4	(a) Scope of Obligation	17
8.5	(b) Exceptions	18
	(a) General. (b) Procedures.	18
Article	(c) Results	18
9.1	Charges for EDS Services	19
9.2	Out-of-Pocket Expenses. Charges for Additional Servcies.	19
9.4	Time of Payment; Disputed Amounts. Taxes.	19
0.0	(a) General. (b) Reports and Returns	19
	(c) Minimization of Taxes(d) Tax Audits	20
9.5	Shared Cost Savings. (a) Franklin Covey Costs.	21
9.6	(b) Shared Costs	21
9.0	(a) Benchmarking	21
	(b) Benchmarking Schedule	22
Article	(d) Corrective Action for Charges	22
10.1	Dispute Escalation	22
10.2	Mediation. Arbitration	22
10.3	(a) Demand for Arbitration; Location	23
	(b) Identification of Arbitrator(c) Conduct of Arbitration	23
	(d) Scope of Discovery	23
	(f) Joinder of Parties	23
10.4 10.5	Exclusive Remedy Continued Performance; Escrow Account	
Article	XI. Termination	
11.1 11.2	Termination for Cause	
11.2 11.3 11.4	Termination for Bankruptcy and Related Events Termination for Convenience	24
11.5	Effect of Expiration or Termination	25
	(b) Transition	25
	(i) Transition Plan. (ii) Specifications (iii) Acquisition of Equipment	25
	(iv) Licensed Programs	26
	(vi) Training(vii) Hiring of Employees	26
	(c) Disclosure of Information(d) Charges	26
	(e) Franklin Covey's Right to Extend the Termination Date	
Article	XII. Indemnities and Liability	
12.1	Indemnities	
	(b) Claims Relating to Personal Injury and Property Damage (i) General	
	(ii) Waiver of Subrogation(c) Infringement Claims	
	(i) General(ii) Exclusions	
	(iii) Additional Remedy(d) Claims Relating to Internet Usage	28
	(e) Third Party Indemnification of EDS(f) Procedures	29
	(i) Notice	29
12.2	Liability(a) General Limitation	30
	(b) Limitation on Other Damages. (c) Exceptions to Limitations	30
	(d) Duty to Mitigate	30
	(e) Contractual Statute of Limitations(f) Acknowledgment	
Article	XIII. Miscellaneous	
13.1 13.2	Right to Engage in Other Activities	
13.3 13.4	Entire Agreement; Survival	31
13.4 13.5 13.6	Amendments, walver Binding Nature; Assignment. Export Regulations.	31
13.6 13.7 13.8	Approvals and Similar Actions	32
13.8	NOTICES Excused Performance	

13.10	Public Relations and Marketing References	32
13.11	Construction Rules	32
13.12	Accounting Terms	33
13.13	Governing Law	33

Schedule 2.1 - Employee Offerees

Schedule 2.2(b) - Key Positions

Schedule 2.7(b) - Franklin Covey Competitors

Schedule 3.2 - Statement of Work

Schedule 3.2(a) - Service Levels

Schedule 3.2(b) - Responsibility Matrix

Schedule 3.2(c) - Supported Locations

Schedule 5 (a) - Equipment Leases

Schedule 5 (b) - Software Licenses Schedule 5 (c)- Maintenance Contracts

Schedule 5 (d)- Franklin Covey-Owned Equipment and Franklin Covey-Leased Equipment Schedule 7.1(a) - EDS Software

Schedule 7.1(b) - EDS-Vendor Software

Schedule 7.1(c) - Franklin Covey Software

Schedule 7.1(d) - Franklin Covey-Vendor Software Schedule 7.1(f) - Other Deliverables

Schedule 9.1 - Charges

Schedule 9.5(a) - Tax Mapping Schedule 11.4 - Termination for Convenience Charges

Schedule 13.8 - Addresses for Notices

Agreement for Information Technology Services

This Agreement for Information Technology Services (the "Agreement"), dated as of April 1, 2001 (the "Effective Date"), is between each of Franklin Covey Co. a Utah corporation ("Franklin Covey"), Electronic Data Systems Corporation, a Delaware corporation ("EDS"), and EDS Information Services L.L.C., a Delaware limited liability company ("EIS"). By way of background, Franklin Covey desires that certain of the technology and information systems operations and functions currently performed by Franklin Covey for itself be performed and managed by a third party experienced in performing and managing such operations. EDS is a provider of various types of information technology services, including those desired by Franklin Covey, and Franklin Covey has selected EDS as its vendor of choice to provide the technology and information systems services during the term of this Agreement. This Agreement documents the terms and conditions under which Franklin Covey agrees to purchase, and EDS agrees to provide, such services. The obligations of EDS set forth in this Agreement will be performed by EDS, itself and through its direct and indirect wholly owned subsidiaries, including EIS. All references to EDS in this Agreement will be deemed to include all such subsidiaries, and EDS and Franklin Covey may be referred to in this Agreement individually as a "Party" and together as the "Parties". For convenience of reference, the Schedule of Definitions attached hereto sets forth the capitalized terms that are used in this Agreement and identifies the sections hereof in which the definitions for such terms appear.

Article I. Relationship Management

- EDS Account Executive. During the term of this Agreement, EDS will provide an individual who will be dedicated to Franklin Covey's account and who will maintain an office at Franklin Covey's facilities located in Salt Lake City, UT (the "EDS Account Executive"). The EDS Account Executive (a) will be acceptable to Franklin Covey, (b) will be the primary contact for Franklin Covey in dealing with EDS under this Agreement, (c) will have overall responsibility for managing and coordinating the delivery of the EDS Services, (d) will meet regularly with the Franklin Covey Representative and (e) will have the power and authority to make decisions with respect to actions to be taken by EDS in the ordinary course of day-to-day management of Franklin Covey's account in accordance with this Agreement.
- Franklin Covey Representative. During the term of this Agreement, Franklin Covey will designate a senior level individual who will be authorized to act as Franklin Covey's primary contact for EDS in dealing with Franklin Covey under this Agreement and who will have the power and authority to make decisions with respect to actions to be taken by Franklin Covey under this Agreement (the "Franklin Covey Representative"). The Franklin Covey Representative may designate in writing a reasonable number of additional Franklin Covey employees to be points of contact for EDS with respect to particular matters of expertise relating to this Agreement.
- Committees. Within (30) days of the Effective Date, each of EDS and Franklin Covey will provide the other with the names of three members of its management staff (inclusive of the EDS Account Executive and the Franklin Covey Representative) who will serve on a steering committee (the "Steering Committee"), with the chairman of the Steering Committee being designated by Franklin Covey from one of its committee members. The Steering Committee will be responsible for overseeing the overall relationship between the Parties, including those activities pertaining to other agreements between the Parties. In addition, the Parties have agreed to establish an Operating Committee, which will be responsible for (a) ensuring sufficient and continued communications between Franklin Covey and EDS, including the provision of reasonable advance notice of significant business planning initiatives or changes in architecture relating to the provision of the Services; (b) to attempt to resolve disputes in accordance with <u>Article X</u> by mutual agreement of both Parties; (c) to conduct quarterly progress reviews during the Term; (d) to review any price changes and amendments to this Agreement; (e) to ensure that all audit findings, security and control issues are effectively resolved; (f) to annually review the responsibilities of Franklin Covey and EDS hereunder; and (g) to undertake such other responsibilities as Franklin Covey and EDS may mutually agree upon from time to time. Although the EDS Account Executive and the Franklin Covey Representative will remain as members of the Steering Committee, either Party may change either or both of its other two representatives from time to time upon written notice to the other. In addition, the Parties may mutually agree to increase or decrease the size, purpose or composition of the Steering Committee in an effort for EDS to better provide, and for Franklin Covey to better utilize, the Services. The location of the Steering Committee meetings will be at the Franklin Covey headquarters in Salt Lake City, Utah. The Steering Committee may, from time to time, create additional committees, the purposes and meeting schedules therefore to be determined as mutually agreeable to the parties.
- Meetings: Reports. Within 30 days of the identification of the Steering Committee, the Steering Committee will determine (a) an appropriate set of periodic meetings to be held by them or other representatives of the Parties involved with the performance of the Services and the procedures to be followed for such meetings, including the preparation of agenda and minutes, and (b) an appropriate set of periodic reports to be issued by EDS to Franklin Covey in addition to those, if any, listed in Schedule 3.2. At a minimum, the meetings will consist of an annual Steering Committee meeting to conduct a quality review of the EDS Services and monthly Steering Committee meetings to discuss, among other things, the responsibilities described above and the strategic objectives of the Parties, and to engage in long-range planning.

1.5 Procedures Manual; Migration Plan.

- Procedures Manual. Within two weeks of the Effective Date EDS will provide Franklin Covey an example procedures manual (the "Procedures Manual") and the parties shall agree to a written plan, including a timetable, for the completion of the Procedures Manual; provided, however, that EDS shall complete and deliver to Franklin Covey the final Procedures Manual no later than 4 months after the Effective Date. The final version of the Procedures Manual shall be in form and content reasonably satisfactory to Franklin Covey and shall include, but not be limited to:
 - Periodic Reports. Periodic reports and detailed requirements for the periodic reports that shall be provided to Franklin Covey by EDS regarding EDS 's provision of the Services; (i)
 - (ii) Goals. Additional goals, specifications and documentation (e.g., operations manuals, user guides, specifications and management reporting) consistent with the provisions of this Agreement;
 - Program Management. A description of the management tools and techniques that EDS shall use to perform the Services, including, but not limited to, EDS' proprietary Service Excellence Dashboard, (iii) Systems Life Cycle Methodology, and Project Management Methodology;
 - Disaster Recovery Procedures. Disaster recovery procedures, together with problem escalation policies and procedures and such other matters as Franklin Covey and EDS may agree; (iv)
 - Change Control Procedures. A description of the change control procedures and implementation techniques to be used therefore, including policies and procedures relating to hardware, software, facility (v) changes, system problem management and emergency changes; and
 - Executive Summary. An executive summary of the operational obligations of EDS under this Agreement. (vi)
- Migration Plan. Within 30 days of the Effective Date, EDS shall prepare and submit to Franklin Covey for review and approval a migration plan detailing the transition of certain Services to leveraged facilities, and (b) implementation of EDS procedures, methodologies and disciplines consistent with the Service Schedules ("the Migration Plan").
- Updates. EDS shall periodically update the Procedures Manual (no less frequently than twice during each contract year of the term) to reflect changes in the operations, policies and procedures described therein. All (c) updates to the Procedures Manual will be submitted to Franklin Covey for review, and will be reasonably satisfactory to Franklin Covey.
- Services. It is the Parties' intention that the Parties' respective obligations under this Agreement will be consistently performed in accordance with the Procedures Manual. Any failure by either Party to consistently (d) perform its obligations under this Agreement in accordance with the Procedures Manual may be referred by the other Party for resolution to the Operating Committee.

- Transition of Employees. EDS will extend offers of at-will employment to each Franklin Covey employee listed in Schedule 2.1 (each, an "Employee Offeree") in accordance with EDS' normal employment policies, subject to the allowances and provisions set forth in this Article II, at a base salary at least equal to the base salary such employee received from Franklin Covey immediately prior to the Effective Date. Each Employee Offeree who accepts EDS' offer of employment pursuant to EDS' normal employment policies and this Article II will become an EDS employee (each, a "Transitioned Employee") as of April 1, 2001 (the "Employment Date") and will be eligible to participate in all employee benefit plans or employment policies and programs available to similarly situated EDS employees.
 - (a) Past Service Credits. For purposes of vesting, participation and benefit accruals for time periods commencing as of the Employment Date only, but not for benefit accrual during periods prior to the Employment Date, in any "employee benefit plan" (as such term is defined in Section 3(3) of the Employee Retirement Income Security Act of 1974, as amended) maintained by EDS, EDS will recognize and give past service credit for service earned by each Transitioned Employee during the period of his or her employment with Franklin Covey ending on the Employment Date. Such past service credit will be established entirely on the information as recorded by Franklin Covey and promptly communicated to EDS, and EDS will not be responsible to audit, validate or confirm the accuracy of any such information.
 - (b) Waiver of Certain Conditions. EDS will waive drug tests for each Transitioned Employee as a condition of EDS' offer of employment. For all Transitioned Employees, the Electronic Data Systems Corporation Health Benefit Plan and the Electronic Data Systems Corporation Long Term Disability Plan provide that any Transitioned Employee is not subject to any pre-existing condition limitations or exclusions; provided, however, that there may be some conditions that are covered under Franklin Covey's health benefit plans that either are not covered or are covered in a different manner under the Electronic Data Systems Corporation Health Benefit Plan, to the extent applicable, will recognize each such Transitioned Employee's (and his or her eligible dependents') un-reimbursed expenses to the extent that such paid expenditures are recognized under the Franklin Covey Health Benefit Plan as a deductible applicable to the particular Transitioned Employee or (eligible dependent) for the same calendar year. All Transitioned Employees receive the maximum "flexible benefit credits" (as such term is defined in the Electronic Data Systems Corporation Flexible Benefit Plan) for medical coverage elected under the Electronic Data Systems Corporation Flexible Benefit Plan. Proof of insurability will be required for certain amounts of Group Universal Life coverage as specified in the applicable policies.
 - (c) Franklin Covey Cooperation. Franklin Covey will cooperate with EDS in connection with the making by EDS of the offers of at-will employment contemplated by this Section 2.1 and the distribution of employment information to the Employee Offerees. Franklin Covey will provide all necessary or appropriate employee data or benefits information in its custody. Franklin Covey has not and will not make any representation, promise or other communication, whether written or oral, to the Employee Offerees regarding employment with EDS or the employment benefits, plans or practices of EDS without obtaining the prior written consent of EDS. Franklin Covey has not taken and will not take any action that diminishes the right of EDS to dismiss, subject to applicable law, any Transitioned Employee at any time and for any or no reason. EDS may request at any time, and Franklin Covey agrees to furnish to EDS or its designated agent, in such format as is mutually agreed, all reasonably required information pertaining to any and all Employee Offerees' employment and benefits as is reasonable for EDS to provide or administer its benefits or payroll or as may be required by any law or governmental agency.
 - (d) <u>Payroll Accommodation</u>. Should EDS request that Franklin Covey continue to make payment to any Transitioned Employees on an interim basis, Franklin Covey will do so as an administrative convenience until such Transitioned Employees can be integrated into the EDS payroll system. In such event, Franklin Covey will be acting solely as an accommodation to EDS, and EDS will reimburse Franklin Covey for all salary paid and employer's contributions made by Franklin Covey in connection therewith.
 - (e) Transfer of Benefits. The Parties acknowledge that this Agreement is not made in connection with a sale of assets of Franklin Covey, and the employment of the Transitioned Employees by EDS pursuant to this Agreement does not constitute a separation from service for purposes of Section 401(k) of the Internal Revenue Code of 1986, as amended (the "Code"). Accordingly, any benefits contributed into the Franklin Covey 401(k) Plan pursuant to cash or deferred arrangements under Code Section 401(k) are not eligible for distribution. Consequently, all benefits and participant loans attributable to Transitioned Employees participating in the Franklin Covey 401(k) Plan will be transferred into the EDS 401(k) Plan in accordance with the terms and conditions of Code Section 414(l) and the Transition Plan listed in Schedule 11.4.
 - (f) Liabilities for Benefit Claims. Franklin Covey, or the appropriate employee benefit plan or insurer of Franklin Covey, will be liable for all employee welfare benefit plan claims of each Transitioned Employee incurred through the employee's last date of employment with Franklin Covey. Franklin Covey will be responsible to make all necessary disclosures to those Employee Offerees or Transitioned Employees who separate employment from Franklin Covey as well as any required filings. Transitioned Employees will experience a qualifying event (as such term is defined under Code Section 4980B). Consequently, Franklin Covey is responsible to extend continuation coverage to each Transitioned Employee and otherwise to comply with all notice and other requirements of Code Section 4980B resulting from the Employee Offeree's or Transitioned Employee's termination of employment with Franklin Covey. Franklin Covey will be solely liable for claims by any Employee Offeree or Transitioned Employee for wages, benefits, third party fees, assessments, duties, permits or other similar charges and for unemployment compensation, severance benefits, workers' compensation and any other employee benefit to which, in each such case, such Employee Offeree or Transitioned Employee or employee's employment or termination of employment with Franklin Covey.

2.2 Use and Replacement of Personnel.

- (a) Account Team. In the event that Franklin Covey is dissatisfied for any significant reason with any EDS personnel providing the Services (including the EDS Account Executive and any Transitioned Employee collectively, the "Account Team"), Franklin Covey will give written notice to EDS of such dissatisfaction and the specific reasons for such dissatisfaction. EDS will have 30 days from the receipt of such notice in which to remedy such problem to the reasonable satisfaction of Franklin Covey. In the event that Franklin Covey remains dissatisfied (in Franklin Covey's reasonable discretion) with the EDS personnel in question after such 30-day period, EDS will promptly replace that EDS person.
- (b) Key Personnel. EDS agrees that each of the EDS personnel listed in Schedule 2.2(b) (the "Key Positions") will devote his or her full time and effort to the performance of the Services. Before assigning a replacement individual to any of the Key Positions, EDS will notify Franklin Covey of the proposed assignment, will provide Franklin Covey with a resume and any other job-related information about the individual reasonably requested by Franklin Franklin Covey in person) the individual for an interview by the appropriate Franklin Covey representatives. If Franklin Covey in good faith objects to the proposed assignment, the Parties shall attempt to resolve Franklin Covey's concerns on a mutually agreeable basis. If the Parties have not been able to resolve Franklin Covey's concerns within 5 business days, EDS shall not assign the individual to that position and shall propose to Franklin Covey the assignment of another individual of similar and suitable ability and qualifications. Personnel filling Key Positions may not be transferred or re-assigned until Franklin Covey has approved a suitable replacement. EDS agrees to use reasonable efforts to provide Franklin Covey with at least 30 days advance written notice prior to transferring or reassigning any personnel filling EDS Key Positions.

3 Qualifications and Retention of EDS Personnel.

- (a) Qualifications. EDS shall assign an adequate number of personnel to perform the Services. The Account Team shall be: (i) Good name and character, and without, to the extent permitted by applicable law, a history of misconduct or dishonesty; (ii) appropriately competent and experienced in performing the tasks, works, functions or obligations to be performed by such personnel; and (iii) if required to perform the applicable task, work, function or obligation, qualified with appropriate U.S. qualifications or qualifications which are recognized in or equivalent to qualifications recognized in the U.S.
- (b) <u>Turnover Rate</u>. Franklin Covey and EDS both agree that it is in their best interests to keep the turnover rate of the Account Team to a reasonably low level. Accordingly, if Franklin Covey believes that EDS 's turnover rate may be excessive and so notifies EDS, EDS shall provide data concerning its turnover rate, meet with Franklin Covey to discuss the reasons for, and impact of, the turnover rate and otherwise use reasonable efforts to keep such turnover rate to a reasonably low level. If appropriate, EDS shall submit to Franklin Covey its proposals for reducing the turnover rate, and the Parties shall mutually agree on a program to bring the turnover rate down to an acceptable level. In any event, notwithstanding transfer or turnover of personnel, EDS shall at all times during the Term remain obligated to perform the Services without degradation and in accordance with this Agreement.
- (c) Training. EDS shall implement and maintain a policy of ongoing training for the Account Team members to ensure that all Account Team members have the requisite skills and knowledge required to perform the tasks, works, functions and obligations assigned to them by EDS.
- Use of Subsidiaries and Subcontractors. The Services will be performed by EDS, itself and through its direct and indirect wholly owned subsidiaries, including EIS. Subject to the other provisions of this Section 2.4, EDS may assign this Agreement to EIS and may subcontract any portion of the Services to unaffiliated third parties; provided, however, that to the extent more than 15% of the work hereunder is to be subcontracted out (exclusive of telecom services and desktop break/fix services), such subcontract will only be entered into with Franklin Covey's prior written consent. EDS will not disclose any confidential information of Franklin Covey to any such subcontractor unless and until the subcontractor has agreed in writing to protect the confidentiality of such confidential information in the manner required by Section 8.4 and then only to the extent necessary for the subcontractor to perform those Services subcontracted to it. EDS will remain responsible for the obligations performed by any of its subsidiaries or subcontractors to the same extent as if EDS performed such obligations.
- 2.5 <u>Hiring of Employees</u>. During the term of this Agreement and for a period of 12 months thereafter, neither Party will solicit, directly or indirectly, for employment any employee of the other Party who is or was actively involved in the performance, consumption or evaluation of the Services without the prior written consent of the other. Notwithstanding the foregoing, the Parties acknowledge and agree that this Agreement will not prohibit (a) any executive search or similar business controlled by EDS or Franklin Covey or by any of their respective affiliates from engaging in its business in the ordinary course in a manner consistent with past practices on behalf of clients other than EDS or Franklin Covey or (b) solicitations through advertising or other publications of general circulation.
- 2.6 <u>Additional Benefits</u>. Franklin Covey shall provide, at no additional cost to EDS, for any Transitioned Employee or EDS employee dedicated to the Franklin Covey account and assigned to work at Franklin Covey's Salt Lake City, Utah headquarters, at prices or discounts comparable to those offered to regular Franklin Covey employees: (a) access to the Franklin Covey Wellness Center, campus cafeteria, and the campus credit union; and (b) a discount equal to the employee discount on Franklin Covey products sold through Franklin Covey retail channels.
- 2.7 No Re-Assignment to Competitor. EDS agrees that it will not assign or otherwise provide Key Personnel to perform Restricted Activities during the Restricted Period in support of Franklin Covey Competitors' United States-based Learning and Performance Solutions Industry operations. For purposes of this Section 2.7, the following terms shall have the meaning as set forth below:
 - (a) "Learning and Performance Solutions Industry" means the business of designing, developing, and/or providing consulting services, training seminars, workshops and personal coaching services and manufacturing and/or distributing products designed to improve individual and organizational effectiveness and productivity through, among other things, effective leadership, time and project management, sales effectiveness, assessment and measurement of return on investment in organizational training and process improvement, and improved oral and written communications and presentations.
 - (b) "Franklin Covey Competitors" means the companies listed on Schedule 2.7(b), attached hereto. Franklin Covey may update the list from time to time, subject to EDS approval.
 - (c) "Key Employees" means the EDS employees set forth in Section 2, which may be updated by written agreement of the Parties to add such additional EDS employees who may be assigned to perform program management and consulting functions in the future.
 - (d) "Restricted Activities" means the performance by Key Employees of program management and consulting services.
 - (e) "Restricted Period" means a period of one year following the last day on which a Key Employee performs program management or consulting services under this Agreement.

Article III. EDS Services

<u>Term; Renewal</u>. Unless terminated earlier in accordance with the provisions of this Agreement, the term of this Agreement will commence on the Effective Date and will expire on March 31, 2011 (the "Expiration Date"). The period of time between the Effective Date and the Expiration date shall be referred to herein as "the Term." Thereafter, the Term of this Agreement may be extended by mutual written agreement of the Parties.

- EDS Services. The EDS Services will consist of: (a) the Statement of Work, attached hereto as Schedule 3.2 (b) Schedule 3.2 (c) the services described in the Migration Plan; (d) the Responsibility Matrix, attached hereto as Schedule 3.2 (b): the supported locations as described in Schedule 3.2 (c) attached hereto; and (e) such other information technology services an ay be agreed upon in writing by the Parties and by reference to this Agreement made a part hereof. Collectively, the services identified in (a), (b), (c) (d), and (e) shall be referred to as the "EDS Services" and will be provided to Franklin Covey in Salt Lake City, Utah, and additional locations specified in the Service Schedules. The EDS Services and any Additional Services are collectively referred to in this Agreement as the "Services". All schedules attached hereto describing the Services shall be referred to herein as the "Service Schedules." EDS agrees to undertake project work at no additional charge to Franklin Covey to the extent that such work can be accomplished without interfering with the performance of Services consistent with agreed-to Service Levels. EDS agrees to work flexibly with Franklin Covey, consistent with the Service Schedules and Procedures Manual, to escalate, prioritize, and manage projects as requested by Franklin Covey.
- 3.3 Services Generally; Performance Criteria. The Services shall be performed by EDS in a manner (including, the degree of accuracy, quality, completeness, timeliness, responsiveness and efficiency) equivalent to or better than the manner in which such Services were performed by Franklin Covey and effective sixty days following the Migration Completion Date (as defined in the Migration Plan), shall meet or exceed the specifications and the Service Levels referred to in or established in the Service Schedules, which specifications and Service Levels which shall be consistent with methods and standards satisfied by well-managed operations performing services similar to the Services. Except as otherwise expressly provided in this Agreement, EDS has the sole right and obligation to supervise, manage, contract, direct, procure, perform or cause to be performed, all Services to be provided pursuant to this Agreement
- 3.4 Service Levels. The performance base lines, performance measurements and performance standards to be used for each Party's performance of its respective obligations under this Agreement are set forth in the Service Schedules. The Parties, through the Steering Committee, agree to annually review, at a minimum, the Service Levels and reports and reports and reports should be adjusted. No such adjustments may be made without the prior written consent of the Parties. If any services, functions, or responsibilities of a Party, not specifically described in this Agreement are reasonably required for the Party's proper performance of its obligations under the Agreement, including without limitation those necessary to satisfy the Service Levels, such services, functions, and responsibilities shall be deemed to be included within the scope of the Agreement to the same extent and manner and subject to the same terms and conditions as if specifically identified in this Section 3.4. Except as otherwise expressly provided in this Agreement, EDS shall provide the facilities, personnel, and other resources necessary to provide the Services.
 - (a) Procedures. EDS and Franklin Covey will jointly implement and utilize written change control procedures, approval procedures and software development methodologies in connection with the performance of their respective obligations under this Agreement. The Parties may modify or supersede such change control procedures and approval procedures by mutual written agreement in the Procedures Manual. EDS may modify or replace such development methodologies at any time after consulting with and receiving approval from the Franklin Covey Representative, except that such modified or replacement methodologies may not increase or otherwise alter in any material respect Franklin Covey's obligations under this Agreement without the prior written consent of Franklin Covey.
 - (b) Reports and Remediation. If EDS fails to attain one or more Service Levels during a particular month, EDS shall
 - (i) Investigate. Promptly investigate the root cause of the failure;
 - (ii) <u>Describe</u>. Provide a description of the failure, the root cause of the failure and a plan for the corrective action to be taken by EDS in connection therewith, all of which shall be contained in the performance report to be delivered by EDS to Franklin Covey pursuant to the Service Schedules in respect of the month during which such failure occurred;
 - (iii) Correct the Error. Use commercially reasonable efforts to correct the problem giving rise to such failure and begin attaining the Service Level as soon as practicable; and
 - (iv) Report. Keep the Franklin Covey Representative apprised of the status and efficacy of the remedial efforts being undertaken by EDS to correct such problem.
 - (c) Meeting. If after pursuing avenues to address Service Level failures pursuant to the Procedures Manual or otherwise under this Agreement, Franklin Covey is concerned about the standard of service being provided by EDS, the Franklin Covey Representative may request a meeting with senior executives of EDS to discuss such concerns. The meeting shall be held at Franklin Covey's corporate headquarters in Salt Lake City, Utah at a mutually convenient time on a day which is not more than 10 Business Days following the day on which the request for the meeting is made by Franklin Covey. Notwithstanding the foregoing, in the event EDS's performance fails to meet the applicable Services Levels, at any time during the Term, Franklin Covey may seek all remedies available to it in under the terms of this Agreement.
 - (d) Service Credits; Other Remedies. If EDS fails to meet the Service Levels for reasons other than the wrongful actions of Franklin Covey or circumstances that constitute force majeure under Section 13.9, then Franklin Covey shall be entitled to receive Accrued Service Credits as defined in Service Schedules. Receipt of Accrued Service Credits, if any, shall be in addition to, and not in lieu of, any other remedies available to Franklin Covey under this Agreement or at law or equity.
 - (e) Technological Advancements. EDS agrees to use commercially reasonable efforts, without an increase in cost or charges to Franklin Covey, to keep the technology utilized in providing the Services to Franklin Covey at a level that is comparable with the level of technological advancement generally attained in the professional services/distribution industry.
 - (f) Efficient Use of Resources. So long as there is no material negative impact on the specifications or Service Levels, EDS will use commercially reasonable efforts, without additional charge or expense to Franklin Covey, to efficiently use the resources chargeable to Franklin Covey (either as part of the base monthly fees or otherwise) used in performing the Services, including, without limitation, (a) tuning or optimizing the system (e.g., optimizing the performance of the system software), (b) consolidating hardware, (c) consolidating software, (d) optimizing the telecommunications network and (e) making schedule adjustments (consistent with Franklin Covey's priorities and schedules for the services and EDS's obligations to attain and maintain the Service Levels) and delaying the performance of non-critical jobs within established limits.
 - (g) <u>Customer Surveys</u>. At least once, but no more than twice during any 12 month period during the term of this Agreement, EDS and Franklin Covey will jointly conduct surveys of Franklin Covey executives to determine the level of Franklin Covey's satisfaction with the Services and the EDS relationship. The Parties will review the survey results and work together to mutually agree on any appropriate adjustments to EDS's practices and procedures as they relate to the provision of the Services and the Parties' relationship.

Additional Services

- (a) From EDS. In addition to the Services, EDS will provide to Franklin Covey such other information technology services as Franklin Covey may reasonably request in writing from time to time during the term of this Agreement and with respect to which the Parties have reached an agreement regarding the nature and scope of such services, the period of time during which such services will be provided and the basis upon which EDS will be compensated therefor (the "Additional Services"). Such agreement with respect to any Additional Services will be set forth in a written letter agreement which will contain the following information, as applicable: (i) a reference to this Agreement, which reference will be deemed to incorporate all of the provisions of this Agreement; (ii) the date as of which the provisions of the letter will become effective and, if applicable, the term or period of time during which the services or resources referenced therein will be provided; (iii) a description of the services or resources to be provided by EDS pursuant to the letter and any performance standards applicable to such services or resources; (iv) a description of Franklin Covey's responsibilities relating to the letter, including any facilities, hardware, software or other support that will be required; (v) the amounts payable for the services or resources provided pursuant to the letter and the schedule on which such amounts will be invoiced; and (vi) any additional provisions applicable to the services or resources to be provided pursuant to the letter that are not otherwise set forth in this Agreement or that are exceptions to the provisions set forth in this Agreement.
- (b) From Third Parties. Nothing in this Section 3.6 or elsewhere in this Agreement will be deemed to limit Franklin Covey's right to solicit or use providers other than EDS (including any Franklin Covey in-house capabilities) to render information technology services that are in addition to, or outside the scope of, the Services; provided. however, that Franklin Covey will request from EDS a new or additional services proposal for any new information technology and information systems management services contemplated (a "New Services Proposal"). Franklin Covey may at its sole discretion also obtain New Services Proposals from other third party service providers. Franklin Covey shall retain the right to select any third party provider it may choose (or to use Franklin Covey personnel), for any reason whatsoever in Franklin Covey 's sole and absolute discretion. Franklin Covey may select any third party provider to perform any new services, including but not limited to, application systems development, contract programming, performance of systems planning projects, operation of new platforms, and performance of other information technology services.
- (c) EDS Responsibilities to Other Service Providers. To the extent Franklin Covey performs any of the new services itself, or retains a third party provider to do so, EDS shall cooperate fully with Franklin Covey or any such third party provider, all at Franklin Covey's reasonable direction.
- (d) Franklin Covey Conditions to Using Third Party Services. Franklin Covey agrees that if it uses a third party to perform services related to its information technology needs, Franklin Covey will use commercially reasonable efforts to ensure that such services will not affect the ability of, or cost to, EDS to perform its obligations under this Agreement, and EDS agrees to reasonably cooperate with any such third party. As a condition to any such cooperation, the third party must agree in writing to the security and confidentiality obligations and procedures reasonably required by EDS, ranklin Covey will keep EDS apprised of any material changes to Franklin Covey's information technology environment that are made by Franklin Covey or any third party. EDS will not be responsible or penalized for any adverse impact on Franklin Covey, the Services or the Service Levels resulting from the acts or omissions of Franklin Covey or third party, but will use commercially reasonable efforts to mitigate it. EDS shall immediately notify Franklin Covey if an act or omission of EDS or a third party may cause a problem or delay in providing the Services and shall diligently work with Franklin Covey or the third party to prevent or circumvent the problem or delay. Franklin Covey shall reimburse EDS for any documented costs incurred by EDS as the result of Franklin Covey using a third party service provider.
- Reliance on Instructions. Unless otherwise specifically provided in the Procedures Manual, in performing its obligations under this Agreement, EDS will be entitled to rely upon any routine instructions, authorizations, approvals or other information provided to EDS by the Franklin Covey Representative or, as to areas of competency specifically identified in writing by the Franklin Covey Representative, by any other Franklin Covey personnel identified by the Franklin Covey Representative, from time to time, as having authority to provide the same on behalf of Franklin Covey in such person's area of competency. Unless EDS knew of any error, incorrectness or inaccuracy in such instructions, approvals or other information. EDS will incur no liability or responsibility of any kind in relying on or complying with any such instructions or information.

Article IV. Franklin Covey's Role

<u>Franklin Covey's Role</u>. During the term of this Agreement and in connection with EDS' performance of its obligations hereunder, Franklin Covey will, at its own cost and expense, have the obligations to EDS, and retain the responsibilities, described in the Service Schedules. Franklin Covey acknowledges and agrees that EDS' ability to perform the Services in accordance with this Agreement is contingent upon Franklin Covey's timely performance of those obligations assigned to Franklin Covey hereunder.

Franklin Covey shall, throughout the Term, provide to EDS personnel assigned to perform Services at Franklin Covey facilities, adequate access to facilities, office supplies, and basic office infrastructure and services.

Article V. Maintenance Contracts and Additional Property

any Maintenance Contracts. EDS shall not renew, modify, terminate or cancel, or request or grant any consents or waivers under any Maintenance Contracts without the prior written consent of Franklin Covey. Any modification, termination or cancellation of, or consent or waiver under, the Maintenance Contracts, obtained or given without Franklin Covey's prior written consent, shall be paid by EDS.

EDS shall (a) receive any invoice submitted by third parties in connection with the Maintenance Contract (the "Maintenance Contract Invoices"), (b) review the Maintenance Contract Invoices in a timely manner and (c) submit such Maintenance Contract Invoices to Franklin Covey for payment within a reasonable period of time after receipt thereof, but in no event in less time than necessary to enable Franklin Covey to timely pay such Maintenance Contract Invoices in accordance with Franklin Covey's normal payment policies. Franklin Covey shall be responsible for any late fees in respect to the Maintenance Contract Invoices, unless such late fees were incurred as a result of EDS's failure to submit the Maintenance Contract Invoices in a timely manner (as set forth in the preceding sentence), in which event EDS shall be solely responsible for such late fees.

EDS shall promptly notify Franklin Covey of any breach of, or misuse or fraud in connection with, any Maintenance Contracts of which EDS becomes aware and shall cooperate with Franklin Covey to prevent or stay any such breach, misuse or fraud. EDS shall not be liable for failures by third parties to perform in accordance with the applicable Maintenance Contract requirements except to the extent such failure is attributable to EDS's failure to properly perform management services hereunder.

Access to Franklin Covey Equipment. During the term of this Agreement, Franklin Covey will allow, or will obtain the right to allow, EDS (and any subcontractors of EDS engaged in accordance with this Agreement) to Access at no charge any equipment owned or leased by Franklin Covey that is reasonably necessary for EDS to perform Services hereunder (the "Franklin Covey Equipment"). For purposes of this Agreement, the term "Access" means the enjoyment of physical and legal use and operation of a specified item of property in order for EDS to perform the Services. EDS agrees that it will use the Franklin Covey Equipment leased by Franklin Covey in a manner consistent with the terms and conditions of such lease.

Except as otherwise provided by this Agreement, EDS will, at its cost and expense, manage the support and maintenance of the Franklin Covey Equipment in accordance with the specifications herein and ensure that all Franklin Covey equipment is kept in good working order. EDS will be responsible for refreshing Franklin Covey desktop equipment, including but not limited to, personal computers, laptop computers, printers, and personal communication devises used by the Transitioned Employees as of the Effective Date.

- Use of Franklin Covey Software and Franklin Covey-Vendor Software. During the term of this Agreement, Franklin Covey will allow, or will obtain the right to allow, EDS (and any subcontractors of EDS engaged in accordance with this Agreement) to Access at no charge (a) the Franklin Covey Software and (b) subject to Franklin Covey having obtained any required consents, the Franklin Covey-Vendor Software. Franklin Covey will pay all costs and expenses with respect to the Franklin Covey Software and the Franklin Covey-Vendor Software. Franklin Covey represents and warrants to EDS that (a) it is not (and, to its knowledge, the licensor is not) in default in any material respect under any of the licenses applicable to the Franklin Covey-Vendor Software and (b) Franklin Covey has delivered to EDS complete copies of those license terms and conditions relating to the use and operation of the Franklin Covey-Vendor Software (including all amendments thereto) as in effect on the Effective Date. EDS agrees that it will use the Franklin Covey-Vendor Software in a manner consistent with such terms and conditions. Franklin Covey agrees not to allow any of such terms and conditions to be amended in any material respect without the prior written consent of EDS.
- Additional Items of Property. If any additional items of property, whether in the form of equipment, software and upgrades thereto or otherwise, are required by EDS to perform the Services or otherwise to meet its obligations hereunder, whether in addition to or in replacement of any then existing property, EDS will inform Franklin Covey and, with Franklin Covey's prior written consent, will obtain such additional items and will provide them to Franklin Covey at Franklin Covey's expense or on such other terms and conditions as the Parties mutually agree. If Franklin Covey, on its own, decides to acquire any additional items of property for use in Franklin Covey's information technology environment, whether in addition to or in replacement of any then existing property, Franklin Covey will consult with, and obtain the consent of, EDS prior to acquiring such property. If Franklin Covey does not approve an acquisition proposed by EDS or does not obtain EDS' consent to Franklin Covey's own acquisition of property, EDS will not be responsible or penalized for any adverse impact on Franklin Covey, the Services or the Service Levels resulting therefrom, but will use commercially reasonable efforts to mitigate it. If EDS' responsibilities in connection with this Section 5.4 (such as site preparation, transportation, installation or maintenance) require resources in addition to those then being used to perform the Services, such resources will be provided as Additional Services.
- Consents; Further Assurances. EDS will work with and assist Franklin Covey in Franklin Covey's efforts to obtain such consents as are required for the transactions contemplated by this Article V, with Franklin Covey paying such transfer, upgrade or other fees as are necessary to obtain a required consent. Franklin Covey will provide EDS with written evidence of such consents upon Franklin Covey's receipt thereof, if EDS has not otherwise received satisfactory evidence thereof. If a required consent is not obtained, (a) unless and until such required consent is obtained, EDS will determine and adopt, subject to Franklin Covey's prior written consent, such alternative approaches as are necessary and sufficient to perform the Services without such required consent and (b) the Parties will mutually agree in writing on any appropriate adjustments to this Agreement, whether with respect to the scope of the Services, the Service Levels, EDS' charges or otherwise. In addition, EDS and Franklin Covey agree to execute and deliver such other instruments and documents as either Party reasonably requests to evidence or effect the transactions contemplated by this Article V.

Article VI. Warranties and Additional Covenants

5.1 <u>Warranties and Additional Covenants.</u>

- (a) <u>Performance</u>. EDS represents and warrants that all Services will be performed in a professional and workmanlike manner, consistent with the standards in the industry. Further, EDS covenants that all Services will be performed consistent with (i) the specifications and conditions set forth in the Service Schedules, and (ii) any and all representations or in this Agreement or any attachment thereto.
- (b) <u>Service Schedule Information</u>. Each party represents to the other that it has taken reasonable care to ensure that the information furnished by said party to the other on which the Service Schedules are based and the charges to be paid by Franklin Covey for the Services are determined, as set forth in this Agreement, is accurate and complete in all material respects. To the extent that such information turns out to be inaccurate or incomplete in any material respect, the Parties agree to negotiate in good faith adjustments to the applicable Service Schedule(s) and / or the charges to be paid by Franklin Covey for the Services, as appropriate.
- (c) Viruses. Each Party will use commercially reasonable measures to screen any software provided or made available by it to the other Party hereunder for the purpose of avoiding the introduction of any "virus" or other computer software routine or hardware components which are designed (i) to permit access or use by third parties to the software of the other Party not authorized by this Agreement, (ii) to disable or damage hardware or damage, erase or delay access to software or data of the other Party or (iii) to perform any other similar actions.
- (d) <u>Disabling Codes</u>. EDS will not, without informing the Franklin Covey Representative, knowingly insert into the software used by it hereunder any code or other device which would have the effect of disabling, damaging, erasing, delaying or otherwise shutting down all or any portion of the Services or the hardware, software or data used in providing the Services. EDS will not invoke such code or other device at any time, including upon expiration or termination of this Agreement for any reason, without Franklin Covey's prior written consent.
- (e) Pass-Through Warranties and Indemnities. EDS agrees that it will pass through to Franklin Covey any rights it obtains under warranties and indemnities given by its third party subcontractors or suppliers in connection with any services, software, equipment or other products provided by EDS pursuant to this Agreement to the extent permitted by the applicable subcontractors or suppliers. If pass-through warranties and indemnities reasonably acceptable to Franklin Covey are not available from a particular subcontractor or supplier, EDS will discuss the matter with Franklin Covey prior to engaging the particular subcontractor or supplier, and the Parties will mutually determine to either accept the terms available from such subcontractor or supplier, in which case EDS will enforce the applicable warranty or indemnity on behalf of Franklin Covey as provided below, or deal with another vendor of comparable services, software, equipment or other products that will provide warranties and indemnities reasonably acceptable to Franklin Covey. In the event of a third party software or equipment nonconformance, EDS will coordinate with, and be the point of contact for resolution of the problem through, the applicable vendor and, upon becoming aware of a problem, will notify such vendor and will use commercially reasonable efforts to cause such vendor to promptly repair or replace the nonconforming item in accordance with such vendor's warranty. If any warranties or indemnities may not be passed through, EDS agrees that it will, upon the request of Franklin Covey, take reasonable action to enforce any applicable warranty or indemnity that is enforceable by EDS in its own name. However, EDS will have no obligation to resort to litigation or other formal dispute resolution procedures to enforce any such warranty or indemnity unless EDS chooses to do so and Franklin Covey agrees to reimburse EDS for all costs and expenses incurred in connection therewith, including reasonable attomeys' fees and expenses.
- (f) Software Development. For a period of 180 days following acceptance of any Developed Software (as defined in Section 7.1(e)) in accordance with the approval procedures adopted by the Parties pursuant to Article 3. EDS warrants that each item of Developed Software will conform in all material respects to the written technical specifications agreed to by the Parties pursuant to Article 3. As soon as reasonably practicable after discovery by Franklin Covey or EDS of a failure of the Developed Software to so conform (a "Nonconformance"), Franklin Covey or EDS, as applicable, will deliver to the other a statement and supporting documentation describing in reasonable detail the alleged Nonconformance. If EDS confirms that there is a Nonconformance, then EDS will use commercially reasonable efforts to correct such Nonconformance. The methods and techniques for correcting Nonconformances will be at the sole discretion of EDS. The foregoing warranty will not extend to any Nonconformances caused (i) by any change or modification to software without EDS' prior written consent or (ii) by Franklin Covey operating software otherwise than (a) in accordance with the applicable documentation, (b) for the purpose for which it was designed or (c) on hardware not recommended, supplied or approved in writing by EDS. Furthermore, if, after undertaking commercially reasonable efforts to remedy a breach by EDS of the foregoing warranty, EDS, in the exercise of its reasonable business judgment, determines that any repair, adjustment, modification or replacement is not feasible, or in the event that the Developed Software subsequent to all repairs, adjustments, modifications and replacements continues to fail to meet the foregoing warranty, Franklin Covey will return the Developed Software to EDS, and EDS will credit to Franklin Covey, in a manner and on a schedule agreed to by the Parties and as Franklin Covey's sole and exclusive remedy for such failure, an amount equal to the charges actually paid by Franklin Covey to EDS for the
- Disclaimer of Warranties. EXCEPT AS OTHERWISE EXPRESSLY PROVIDED IN SECTION 6.1 AND ARTICLE V. EDS MAKES NO REPRESENTATIONS OR WARRANTIES, EXPRESS OR IMPLIED, REGARDING ANY MATTER, INCLUDING THE MERCHANTABILITY, SUITABILITY, ORIGINALITY, FITNESS FOR A PARTICULAR USE OR PURPOSE, OR RESULTS TO BE DERIVED FROM THE USE, OF ANY INFORMATION TECHNOLOGY SERVICE, SOFTWARE, HARDWARE OR OTHER MATERIALS PROVIDED UNDER THIS AGREEMENT, OR THAT THE OPERATION OF ANY SUCH SERVICE, SOFTWARE, HARDWARE OR OTHER MATERIALS WILL BE UNINTERRUPTED OR ERROR-FREE.

Article VII. <u>Proprietary Rights</u>

7.1 <u>Definitions</u>. For purposes of this Agreement:

- (a) <u>EDS Software</u>. The term "EDS Software" means any Software that is owned or developed by or on behalf of EDS before the Effective Date, or is acquired or developed by or on behalf of EDS after the Effective Date without reference to or use of the intellectual property of Franklin Covey. The EDS Software that is expected, as of the Effective Date, to be used initially by EDS in performing the Services is listed in <u>Schedule 7.1(a)</u>.
- (b) EDS-Vendor Software. The term "EDS-Vendor Software" means any Software that is proprietary to any party other than EDS or Franklin Covey and is licensed to EDS. The EDS-Vendor Software that is expected, as of the Effective Date, to be used initially by EDS in performing the Services is listed in Schedule 7.1(b).
- (c) Franklin Covey Software. The term "Franklin Covey Software" means any Software that is owned or developed by or on behalf of Franklin Covey before the Effective Date, or is acquired or developed by or on behalf of Franklin Covey (other than by EDS) after the Effective Date without reference to or use of the intellectual property of EDS. The Franklin Covey Software that is expected, as of the Effective Date, to be used initially by EDS in performing the Services is listed in Schedule 7.1(c).
- (d) Franklin Covey-Vendor Software. The term "Franklin Covey-Vendor Software" means any Software that is proprietary to any party other than Franklin Covey or EDS and is licensed to Franklin Covey. The Franklin Covey.

Covey-Vendor Software that is expected, as of the Effective Date, to be used initially by EDS in performing the Services is listed in Schedule 7.1(d).

- (e) <u>Developed Software</u>. The term "Developed Software" means any Software that is developed and delivered by EDS under this Agreement, and paid for by Franklin Covey, (i) including any Software Changes made by EDS to Franklin Covey Software or Franklin Covey-Vendor Software as part of the Services, but (ii) excluding (A) any Software Changes made by EDS to EDS Software or EDS-Vendor Software, (B) any EDS Software or EDS Development Tools that are used in developing, modifying or enhancing any Developed Software hereunder and (C) any Residual Technology.
- (f) Other Deliverables. The term "Other Deliverables" means those tangible, viewable items other than Software listed in Schedule 7.1(f) that are developed and delivered by EDS under this Agreement, and paid for by Erapklin Cover.
- (g) EDS Development Tools; Residual Technology. The term "EDS Development Tools" means all know-how, intellectual property, methodologies, processes, technologies, algorithms, software or development tools used in performing the Services which (i) are based on trade secrets or proprietary information of EDS, (ii) are developed or created by or on behalf of EDS without reference to or use of the intellectual property of Franklin Covey or (iii) are otherwise owned or licensed by EDS. The term "Residual Technology" means the ideas, concepts, methodologies, processes and know-how which are developed or created by EDS in the course of performing the Services and may be retained by EDS' employees in intangible form.
- (h) Software. The term "Software", as used in the terms that are defined in this Section 7.1, means computer programs, together with input and output formats, source and object codes, program listings, data models, flow charts, outlines, narrative descriptions, operating instructions and supporting documentation, and includes the tangible media upon which such programs and documentation are recorded, including all authorized reproductions of such programs. Except as otherwise expressly provided in this Agreement, Software includes any corrections, enhancements, translations, modifications, updates, new releases, new versions and other changes thereof or thereto (collectively, "Software Changes").
- EDS Software. All EDS Software, including all Software Changes made thereto, will be and remain EDS' property, and Franklin Covey will have no rights or interest therein. EDS will grant to Franklin Covey a perpetual (subject to compliance with the applicable license), nontransferable, nonexclusive license to use, after the Expiration Date or the effective date of termination by Franklin Covey, whichever is applicable, the object code form of any application software programs, including existing documentation, of the EDS Software (if any) then being used by EDS in performing the Services, including those items of EDS Software being used at one or more EDS data centers as part of the Services as may be reasonably necessary for Franklin Covey's continued use of Franklin Covey Vendor Software and any other Software to which Franklin Covey has continuing rights, solely for the purpose of Franklin Covey (or the Third Party Provider assuming the obligations of EDS hereunder) performing the information and technology systems services previously performed by EDS hereunder (the "Licensed Programs"), subject to EDS and Franklin Covey entering into an agreement, in form and substance reasonably satisfactory to EDS and Franklin Covey, containing such terms and conditions as may be appropriate. Notwithstanding anything to the contrary in this Agreement, such license will not include the right to use any Software Changes with respect to the EDS Software other than those in use at the time the license is granted.
- 7.3 EDS-Vendor Software. All EDS-Vendor Software will be and remain the property of the applicable third party vendor(s), and, as between EDS and Franklin Covey, all Software Changes made by EDS thereto will be owned by EDS. EDS will obtain all consents necessary to permit EDS (and any subcontractors of EDS engaged in accordance with this Agreement) to Access the EDS-Vendor Software in connection with the performance of the Services and will pay all costs and expenses associated therewith. During the term of this Agreement, EDS will pay all required license, installation, maintenance and upgrade fees with respect to the EDS-Vendor Software.
- 7.4 Franklin Covey Software. All Franklin Covey Software, excluding any Software Changes made by EDS thereto as part of the Services (which will be Developed Software and will be owned as set forth in Section 7.4), will be and remain Franklin Covey's property. The Franklin Covey Software will be made available to EDS in such form and on such media as EDS may reasonably request, together with appropriate documentation.
 - Franklin Covey-Vendor Software. All Franklin Covey-Vendor Software, excluding any Software Changes made by EDS thereto as part of the Services (which will be Developed Software and, subject to any Franklin Covey-Vendor rights or restrictions, will be owned as set forth in Section 7.6), will be and remain the property of the applicable third party vendor(s). The Franklin Covey-Vendor Software will be made available to EDS in such form and on such media as EDS may reasonably request.
 - Developed Software and Other Deliverables. Each party will retain all rights, including trademarks, patents, trade secrets and copyrights ("IP Rights"), in any software, ideas, concepts, know-how, development tools, techniques or any other proprietary material or information that it owned or developed prior to the date of this Agreement, or acquired or developed after the date of this Agreement without reference to or use of the intellectual property of the other party. Subject to any third party rights or restrictions and the other provisions of this paragraph 7.6. Franklin Covey. Notwithstanding anything to the contrary in this Agreement, EDS (i) will retain all IP Rights in and to all software development tools, know-how, methodologies, processes, technologies or algorithms used in performing the Services which are based on previously developed trade secrets or proprietary information of EDS or are otherwise owned or licensed by EDS (collectively, "tools"), (ii) will be free to use the ideas, concepts and know-how which are developed or created in the course of performing the Services and may be retained by EDS' employees in intangible form, all of which constitute substantial rights on the part of EDS in the technology developed as a result of the Services performed under this Agreement, and (iii) will retain ownership of any prior-developed EDS-owned software or tools ("EDS Tools") that are used in producing the deliverables and become embedded in the deliverables. EDS hereby grants to Franklin Covey a perpetual (subject to compliance with this sentence), royalty-free, nontransferable, nonexclusive license to use such embedded EDS Tools (if any) solely in connection with Franklin Covey's internal use and exploitation of the deliverables and only so long as such software and tools (if any) remain embedded in the deliverables and are not separated therefrom. EDS will own patent rights with respect to processes and menthodologies developed by EDS in connection with deliverables other than the copyright owners
- 7.7 EDS Development Tools; Residual Technology. Notwithstanding anything to the contrary in this Agreement, EDS will retain all right, title and interest in and to, and will be free to use, (a) the EDS Development Tools and (b) subject to the confidentiality obligations set forth in Section 8.4, the Residual Technology. The Parties acknowledge and agree that EDS' right, title and interest in and to the Residual Technology constitute substantial rights in the technology developed as a result of the Services performed under this Agreement. No licenses will be deemed to have been granted by either Party to any of its patents, trade secrets, trademarks or copyrights, except as otherwise expressly provided in this Agreement. Nothing in this Agreement will require EDS or Franklin Covey to violate the proprietary rights of any third party in any software or otherwise.
- 7.8 <u>Further Assurances</u>. EDS and Franklin Covey agree to execute and deliver such other instruments and documents as either Party reasonably requests to evidence or effect the transactions contemplated by this <u>Article VII</u>. The provisions of this <u>Article VII</u> will survive the expiration or termination of this Agreement for any reason.

Article VIII. <u>Data, Confidentiality and Audit Rights</u>

- B.1 Data of Franklin Covey. As between EDS and Franklin Covey, information relating to Franklin Covey or its customers (the "Franklin Covey Data") is confidential, will be subject to Section 8.4 and will be and remains the property of Franklin Covey. EDS (and any subcontractors of EDS engaged in accordance with this Agreement) is hereby authorized to have access to and to make use of the Franklin Covey Data for the term of this Agreement as is appropriate for the performance by EDS of its obligations hereunder. Upon expiration or termination of this Agreement for any reason, subject to any record retention requirements of EDS, EDS will, at Franklin Covey's expense, return to Franklin Covey all of the Franklin Covey Data in EDS' possession and in EDS' then existing machine-readable format and media. EDS will not use the Franklin Covey Data for any purpose other than providing the Services. At no time shall any of the computer or other files or other materials or information containing Franklin Covey Data be stored or held in a form or manner not reasonably accessible to Franklin Covey through the Franklin Covey Representative. Subject to restrictions contained in any relevant contracts, EDS shall provide the Franklin Covey Representative or such other person or persons as may be designated in writing by Franklin Covey all such files and other materials promptly upon the reasonable request of Franklin Covey, including hardware and software keys and such information as to format encryption (if any) and any other specification or information necessary for Franklin Covey to retrieve, read, revise and/or maintain such files and information.
- 8.2 <u>Safeguarding Data</u>. EDS will maintain safeguards against the destruction, loss or alteration of the Franklin Covey Data in the possession of EDS, which safeguards are consistent with those written procedures established and in use by Franklin Covey as of the Effective Date and provided to EDS. To the extent that any such procedures have not been established, EDS will maintain safeguards that are no less rigorous than those maintained by EDS for its own information of a similar nature. Franklin Covey will have the right to establish backup security for the Franklin Covey Data and to keep backup data and data files in its possession if it so chooses; provided, however, that EDS will have access to such backup data and data files as is reasonably required by EDS.

Without limiting the above:

- (a) Restricted Access. EDS Personnel shall not attempt to access, or allow access to, any data, files or programs to which they are not entitled under this Agreement. If such Access is attained, EDS shall immediately report the incident to the Franklin Covey Representative, describe in detail any accessed materials and the method of access and, upon request, provide to the Franklin Covey Representative copies of any accessed materials; and
- (b) Security System. EDS shall institute reasonable system security measures to guard against the unauthorized access, alteration or destruction of software and Franklin Covey Data. Such measures shall include the installation of software which (i) requires all users to enter a user identification code and password prior to gaining access to the system, (ii) controls and tracks the addition and deletion of users, and (iii) controls user access to areas and features of the system.
- (c) <u>Loss of Data</u>. If any Franklin Covey Data is lost or damaged due to EDS' failure to perform Services as required under this Agreement, EDS shall use commercially reasonable efforts to replace or regenerate such lost or damaged Franklin Covey Data without additional charge or expense and shall not require Franklin Covey to pay for any increased resource usage from the replacement or regeneration of such lost or damaged data.
- Privacy Laws. The Parties acknowledge and agree that Franklin Covey will be and remain the controller of the Franklin Covey Data for purposes of all applicable laws relating to data privacy, trans-border data flow and data protection (collectively, the "Privacy Laws"), and nothing in this Agreement will restrict or limit in any way Franklin Covey's rights or obligations as owner and/or controller of the Franklin Covey Data for such purposes. The Parties also acknowledge and agree that EDS may have certain responsibilities prescribed by applicable Privacy Laws as a processor of the Franklin Covey Data, and EDS hereby acknowledges such responsibilities to the extent required thereby for processors of data and agrees that such responsibilities will be considered as a part of the Services to be provided by EDS under this Agreement; provided, however, that in the event that Privacy Laws to which the activities contemplated by this Agreement are subject are modified, EDS will work with Franklin Covey in an effort to continue to comply with such Privacy Laws, as so modified, but to the extent that such modifications expand the scope of the activities previously undertaken by EDS pursuant to this Section 8.3. EDS will, at Franklin Covey's reasonable request, provide such additional activities as Additional Services.

3.4 <u>Confidentiality.</u>

(a) Scope of Obligation. Except as otherwise expressly provided in this Agreement, EDS and Franklin Covey each agrees that (i) all information communicated to it by the other and identified as confidential, whether before or after the Effective Date, (ii) all information identified as confidential to which it has access in connection with the Services, whether before or after the Effective Date, and (iii) this Agreement and the Parties' rights and obligations hereunder, will be and will be desen to a proper to the proper or after the Effective Date, and (iii) this Agreement and the Parties' rights and obligations hereunder, will be and will be desen precived in confidence and will be used only for purposes of this Agreement, and each of EDS and Franklin Covey agrees to use the same means as it uses to protect its own confidential information, but in no event less than reasonable means, to prevent the disclosure and to protect the confidentiality thereof. No such information will be disclosed by the recipient Party without the prior written consent of the other Party; provided, however, that each Party may disclose this Agreement and the other Party's confidential information to those of the recipient Party, so long as the recipient Party requires, in the case of its attorneys, auditors and insurers, that each of them execute a confidentiality agreement containing terms and conditions no less restrictive than those set forth in this Section 8.4 and advises, in the case of its subcontractors and employees, each such subcontractor and employees of the confidentiality obligations set forth in this Section 8.4 will remain the responsibility of the Party employing or engaging such persons. Notwithstanding the foregoing, EDS may disclose this Agreement and other confidential information to which it has access hereunder to professional advisers, financial institutions and other third parties in connection with any transaction entered into to

provide financing related to this Agreement or the obligations of EDS hereunder, so long as each of them execute a confidentiality agreement containing terms and conditions no less restrictive than those set forth in this Section 8.4.

(b) Exceptions. The foregoing will not prevent either Party from disclosing information that belongs to such Party or (i) is already known by the recipient Party without an obligation of confidentiality other than under this Agreement, (ii) is publicly known or becomes publicly known through no unauthorized act of the recipient Party, (ii) is rightfully received from a third party, (iv) is independently developed without use of the other Party's confidential information or (v) is disclosed without similar restrictions to a third party by the Party owning the confidential information. If confidential information is required to be disclosed pursuant to a requirement of a governmental authority, such confidential information may be disclosed pursuant to such requirement so long as the Party required to disclose the confidential information to the extent possible, provides the other Party with timely prior notice of such requirement and coordinates with such other Party in an effort to limit the nature and scope of such required disclosure; provided, however, that, in the event of a tax audit, (A) notice of a disclosure requirement in connection therewith will not be given, and (B) the Parties will use commercially reasonable efforts to ensure that any confidential information that is subject to a valid request for delivery of a copy of such information (including a copy of this Agreement) to the taxing authority is not subject to further disclosure by it (such as by marking such information as a trade secret). If confidential information is required to be disclosed in connection with the conduct of any mediation or arbitration proceeding carried out pursuant to Article X, such confidential information may be disclosed pursuant to and in accordance with the approval and at the direction of the mediator or arbitrator, as the case may be, conducting such proceeding. Upon written request of the disclosing Party at the expiration or termination of this Agreement for any reason, all such documented confidential informat

8.5 Audit Rights

- (a) General. Employees of Franklin Covey and its auditors who are from time to time designated by Franklin Covey and who agree in writing to the security and confidentiality obligations and procedures reasonably required by EDS will be provided with reasonable access to any facility at which the Services are being performed to enable them to conduct audits of EDS' performance of the Services and other matters relevant to this Agreement, including (i) verifying the accuracy of EDS' charges to Franklin Covey and (ii) verifying that the Services are being provided in accordance with this Agreement, including any Service Levels.
- (b) Procedures. Such audits may be conducted once a year during reasonable business hours; provided, however, that the Parties may agree to more frequent audits as deemed reasonably necessary. Franklin Covey will provide EDS with prior written notice of an audit. EDS will cooperate in the audit, will make the information reasonably required to conduct the audit available on a timely basis and will assist the designated employees of Franklin Covey or its auditors as reasonably necessary. If Franklin Covey requests resources beyond those resources then assigned to the account team under this Agreement who are able to provide reasonable assistance of a routine nature in connection with such audit, such resources will be provided as Additional Services. EDS will retain records that support EDS' performance of the Services and other matters relevant to this Agreement in accordance with EDS' retention guidelines. Notwithstanding anything to the contrary in this Agreement, EDS will not be required to provide access to the proprietary data of EDS or other EDS customers. All information learned or exchanged in connection with the conduct of an audit, as well as the results of any audit, is confidential and will be subject to Section 8.4.
- (c) Results. Following an audit, Franklin Covey will conduct an exit conference with EDS to discuss issues identified in the audit that pertain to EDS, and Franklin Covey will give EDS a copy of any portion of the audit report pertaining to EDS. The Parties will review each EDS audit issue and will determine (i) what, if any, actions will be taken in response to such audit issues, when and by whom and (ii) which Party will be responsible for the cost of taking the actions necessary to resolve such issues. Any such determination will be based on the following criteria: (A) who the owner of the original deficiency is; (B) who has contractual responsibility for the improvement of internal controls; and (C) who owns the standards against which the audit is done. EDS will not be responsible for the cost of an audit, except to the extent the audit expenses are incurred in connection with a specific audit limited to EDS and the an original material deficiency discovered in the audit is attributable to EDS, or otherwise agreed to in writing by the Parties. For the purposes of this Agreement a "material deficiency" is any discrepancy between charges billed and actual billable charges of greater than five percent (5%).

Article IX. Payments

- 9.1 Charges for EDS Services. In consideration for the performance of the Services, Franklin Covey will pay to EDS the charges set forth in Schedule 9.1, plus any and all Collected Taxes (as defined in Section 9.4) and other amounts described in this Article IX. EDS will invoice Franklin Covey for such charges on a monthly basis in arrears, with each invoice setting forth the charges related to the previous month. EDS either will include on each such invoice, or will bill Franklin Covey separately for, the Collected Taxes for which Franklin Covey is responsible hereunder.
- Out-of-Pocket Expenses. Franklin Covey will pay, or reimburse EDS for, the reasonable out-of-pocket expenses, including travel and travel-related expenses, incurred by EDS with the prior written consent of Franklin Covey, in connection with EDS' activities requested by Franklin Covey and not within the general scope of the Services, including any purchases by EDS on behalf of Franklin Covey of additional items of property pursuant to Section 5.7. EDS will invoice Franklin Covey separately for all such out-of-pocket expenses, which invoice will be sent by EDS to Franklin Covey after EDS incurs such expenses and will contain an itemized listing of the applicable expenses.
- .3 Charges for Additional Services. In consideration for any agreement by EDS to provide new or additional Services, Franklin Covey will pay to EDS (a) the amounts mutually agreed to in writing by EDS and Franklin Covey for the new or additional Services, and (b) any and all collected Taxes and (c) the out-of-pocket expenses incurred by EDS (in accordance with Section 9.2) in connection with the performance by EDS of the new or additional Services.
- Time of Payment; Disputed Amounts. Except as otherwise expressly provided in this Agreement, any amount due to EDS under this Agreement and not disputed in good faith by Franklin Covey (as provided below) will be due and payable on the 30th day following receipt by Franklin Covey of the invoice from EDS therefor. All amounts will be payable to EDS by check or by ACH (for amounts less than \$1,000,000) or more), in accordance with payment instructions provided by EDS from time to time, so as in each case to constitute immediately available funds by 12 noon, Plano, Texas time, on the payment date no matter what the method of payment. Any amount not paid when due will bear interest until paid at a rate of interest equal to the lesser of (a) the prime rate established from time to time by Citibank of New York plus one and one-half percent or (b) the maximum rate of interest allowed by applicable law. If any portion of an amount due to EDS under this Agreement is subject to a bona fide dispute between the Parties, Franklin Covey will pay to EDS on the date such amount is due all amounts not disputed in good faith by Franklin Covey. Within 15 days of Franklin Covey's receipt of the invoice on which a disputed amount appears, Franklin Covey will notify EDS in writing of the specific items in dispute, will describe in detail Franklin Covey's reason for disputing each such item and will deposit such disputed amount into an escrow account established in the manner set forth in Section 10.5. Within 10 days of EDS' receipt of such notice, the Parties will negotiate in good faith pursuant to the provisions of Article X to reach settlement on any items that are the subject of such dispute. If Franklin Covey does not notify EDS of any items in dispute within such 15-day period of time, Franklin Covey will be deemed to have approved and accepted such invoice.

9.5 <u>Taxes</u>.

- (a) General. Unless Franklin Covey provides EDS with a valid and applicable exemption certificate, there will be added to any charges under this Agreement, or there will be separate billing for, and Franklin Covey will bear the cost of and will pay to EDS, or reimburse EDS for, any and all Collected Taxes arising from or imposed on this Agreement, the transactions arising hereunder, the charges payable hereunder, the Services or the provision, license or sale thereof or the Software, equipment, materials, property (tangible or intangible) or other resources provided hereunder or used in connection herewith; provided, however, that Franklin Covey's obligations hereunder with respect to Collected Taxes will not apply to any income taxes that are based on or measured by EDS' net income. The Parties acknowledge and agree that EDS shall accurately and correctly invoice or bill Franklin Covey for the Services by separating the taxable and non-taxable Services in accordance with the SOW, applicable federal and state law, and as described on the Taxing Matrix attached hereto as Schedule 9.5(a). For purposes of this Agreement, the term "Collected Taxes" means any present or future taxes, assessments, duties, permits, tariffs, fees and other charges of any kind, however designated, assessed, charged or levied now or hereafter, including without limitation sales, use, state, local, provincial, property, gross receipts, excise, transaction, goods and services, value-added or similar taxes, other taxes or amounts in whatever nature or in lieu thereof, and all fines, penalties, interest, additions to taxes (penalty in nature) or similar liabilities imposed in connection therewith. Subject to Section 12.1(f), Franklin Covey will indemnify and defend EDS from any and all Collected Taxes, including without limitation any additions to tax, penalties, interest, fines, fees, reasonable attorneys' and accountants' fees and other expenses arising out of, under or in connection with any claim (i) that any Collected Taxes are not paid
- (b) Reports and Returns. All reports and returns required to be filed with regard to Collected Taxes will be prepared and timely filed by the Party required by the applicable taxing authority to file such reports and returns. If EDS takes a position on any form, in any return, on audit or otherwise on behalf of Franklin Covey, or at Franklin Covey's request, Franklin Covey will bear the cost of and pay EDS, or reimburse EDS for the payment of, any and all expenses and liabilities, including without limitation Collected Taxes, costs, reasonable attorneys' and accountants' fees and other expenses or disbursements arising out of, attributable to or in connection with any audit or contest of such Collected Taxes, or any claim or assessment claimed by any taxing authority.
- (c) Minimization of Taxes. The Parties will reasonably cooperate in the minimization of taxes to the extent legally permissible. The Parties agree that, for state and local sales and use tax purposes, the amounts payable under this Agreement will be audited, where permissible, by the applicable taxing authority as part of EDS' state and local sales and use tax audits, but Franklin Covey will pay, or reimburse EDS for, any and all Collected Taxes assessed under the audits.
- (d) Tax Audits. If any claim is made or liability asserted by notice to or commencement of proceedings against EDS (or any of its affiliates) for any Collected Taxes as to which Franklin Covey has a payment obligation pursuant to this Section 9.5, EDS will notify Franklin Covey of such claim or liability in writing and will furnish Franklin Covey with copies of the claim or notice of liability and all other applicable writings received from the applicable taxing authority; provided, however, that EDS' failure to so notify or furnish such copies to Franklin Covey will not operate to relieve Franklin Covey of its obligation to indemnify EDS under this Agreement. Within 15 days of Franklin Covey's receipt of notice from EDS of such claim or liability, but not later than three business days before the date required to timely file any documents with the applicable taxing authority, Franklin Covey may request that EDS contest such claim or liability. If (i) requested by Franklin Covey, (ii) a good faith, reasonable basis (as defined in Section 6662(d)(2)(B)(ii)(II) of the Internal Revenue Code of 1986, as amended) exists for such contest and (iii) EDS determines that such contests will not result in any adverse impact to EDS (or any of its affiliates), EDS will, in good faith and at Franklin Covey to EDS of all reasonable costs and expenses incurred, including reasonable attorneys' and accountants' fees), contest the claim or liability, mountain the proceedings, or both, (B) whether any such other contests such claim or liability, will be initially by way of judicial or administrative proceedings, or both, (B) whether any such claim or liability, including (A) whether any action to contest such claim or liabilities will be contested by resisting payment therefor by paying the same and seeking a refund thereof and (C) if EDS undertakes judicial action with respect to such claim or liability, the court or other judicial body before which such action will be commenced. EDS will have full control over any contest pursuan
- Shared Cost Savings. In the event either party proposes a change to the way Services are provided hereunder or requests new services and such change or new services results in a readily quantifiable out-of-pocket savings in cost to either EDS or Franklin Covey, the Parties agree that upon the realization of any such savings, such savings shall be allocated between the Parties as follows:
 - (a) Franklin Covey Costs. If Franklin Covey proposes the change and pays for the entire cost of such change or new services, all savings associated therewith shall be allocated to Franklin Covey. If EDS proposes the change and Franklin Covey pays for the entire cost of such change or new services, then EDS shall be allocated 25% of the savings associates therewith.
 - (b) Shared Costs. If the cost of such change or new services is shared by Franklin Covey and EDS, the savings associated therewith shall be allocated between the Parties in accordance with the percentages mutually agreed upon by the Parties at the time such change or new services were agreed upon by the Parties.

9.6 Cost and Quality Benchmarking

- (a) Benchmarking. At Franklin Covey's request, the Parties will from time to time during the Term measure the quality and cost-effectiveness of the Services through the use of independent third party benchmarking services (each, a "Benchmark"). Comparisons will be to determine whether, in the aggregate, the Services being delivered are comparable to similarly bundled price and service offerings of other best-in-class information technology service providers, taking into account financial and operational risks, volume, term and any minimum commitments. The party used to perform the benchmarking services (the "Benchmarker") will be reasonably acceptable to both Parties, and will exclude any entity that is a competitor of either party, or that would be conflicted by its status as a corporate auditor of either Party. Agreement upon a Benchmarker shall be contingent upon such Benchmarker's agreement to such non-disclosure obligations as may be reasonably required by either Party. The fees and services charged by the Benchmarker shall be borne by Franklin Covey. EDS shall fully cooperate with such efforts, investigate unfavorable variances, and take corrective action as necessary.
- (b) Benchmarking Schedule. At Franklin Covey's s request, the initial Benchmark will be performed prior to the second anniversary of the Effective Date. Thereafter, Benchmarks will be performed on a mutually agreed upon schedule, but in no event more than every two years.
- (c) Corrective Action for Quality. If the Benchmark shows that any individual Service quality is not in the upper quartile of the group used in the Benchmark, EDS shall take such corrective action as necessary to promptly bring the particular Service into the upper quartile within 30 days from receipt of the Benchmark or such other time period as may otherwise be mutually agreed upon by the Parties. If as a result of technological changes, the cost of EDS performing Services within the upper quartile of the Benchmark would exceed the then current costs of providing such services by more than ten percent (10%), then the costs of such corrective action shall be the subject of good faith negotiations between the Parties.
- (d) Corrective Action for Charges. If the Benchmark shows that the aggregate charges paid by Franklin Covey to EDS for the Services are not in the third or lower quartile of the group used in the Benchmark (i.e., lower in cost than 50% of the companies in the group), EDS shall downwardly adjust the charges herein in order to bring the aggregate charges into the third quartile. In the event EDS fails to take such corrective action (which failure shall not constitute a breach of this Agreement), an Executive Vice President of Franklin Covey Co. (or equivalent) and a Client Executive of EDS (or equivalent) (the "Executives") shall meet and attempt in good faith, to resolve any dispute between the Parties relative to this Section 11.4.

Article X. Dispute Escalation, Mediation and Arbitration

- Dispute Escalation. In the event of any dispute, controversy or claim of any kind or nature arising under or in connection with this Agreement (including disputes as to the creation, validity, interpretation, breach or termination of this Agreement) (a "Dispute"), then upon the written request of either Party, each of the Parties will appoint a designated senior business executive (other than the EDS Account Executive or the Franklin Covey Representative) whose task it will be to meet for the purpose of endeavoring to resolve the Dispute. The designated executives will meet as of fear as the Parties reasonably deem necessary in order to gather and furnish to the other all information with respect to the matter in issue which the Parties believe to be appropriate and germane in connection with its resolution. Such executives will discuss the Dispute and will negotiate in good faith in an effort to resolve the Dispute without the necessity of any formal proceeding relating thereto. The specific format for such discussions will be left to the discretion of the designated executives but may include the preparation of agreed upon statements of act or written statements of position furnished to the other Party. No formal proceedings for the resolution of the Dispute under Section 10.2 or 10.3 may be commenced until the earlier to occur of (a) a good faith mutual conclusion by the designated executives that amicable resolution through continued negotiation of the matter in issue does not appear likely or (b) the 30th day after the initial request to negotiate the Dispute.
- 10.2 Mediation. Any Dispute that the Parties are unable to resolve through informal discussions or negotiations pursuant to Section 10.1 will be submitted to non-binding mediation, which will be held in Plano, Texas if initiated by Franklin Covey and in Salt Lake City, Utah if initiated by EDS. The Parties will mutually determine who the mediator will be from a list of mediators obtained from the American Arbitration Association office located in the city in which the proceeding will take place, determined as set forth above (the "AAA"). If the Parties are unable to agree on the mediator, the AAA will select the mediator. Each Party will bear its own costs and expenses with respect to the mediaton, including one-half of the fees and expenses of the mediator.
- 10.3 Arbitration. Any Dispute that the Parties are unable to resolve through mediation pursuant to Section 10.2 will be submitted to arbitration in accordance with the following procedures:
 - (a) <u>Demand for Arbitration; Location</u>. Either Party may demand arbitration by giving the other Party written notice to such effect, which notice will describe, in reasonable detail, the facts and legal grounds forming the basis for the filing Party's request for relief and will include a statement of the total amount of damages claimed, if any, and any other remedy sought by that Party. The arbitration will be held before one neutral arbitrator in Plano, Texas if the proceedings are initiated by Franklin Covey and in Salt Lake City, Utah if the proceedings are initiated by EDS.
 - (b) <u>Identification of Arbitrator</u>. Within 30 days after the other Party's receipt of such demand, the Parties will mutually determine who the arbitrator will be. If the Parties are unable to agree on the arbitrator within that time period, the AAA will select the arbitrator. In any event, the arbitrator will have a background in, and knowledge of, the information technology services industry and will be an appropriate person based on the nature of the Dispute. If a person with such industry experience is not available, the arbitrator will be chosen from the large and complex case panel or, if an appropriate person is not available from such panel, the retired federal judges pool.
 - (c) Conduct of Arbitration. The Commercial Arbitration Rules of the AAA will govern the arbitration, except as expressly provided in this Section 10.3. However, the arbitration will be administered by any organization mutually agreed to in writing by the Parties. If the Parties are unable to agree on the organization to administer the arbitration, it will be administered by the AAA under its procedures for large and complex cases.
 - (d) Scope of Discovery. Discovery will be limited to the request for and production of documents, depositions and interrogatories. Interrogatories will be allowed only as follows: a Party may request the other Party to identify by name, last known address and telephone number (i) all persons having knowledge of facts relevant to the Dispute and a brief description of that person's knowledge, (ii) any experts who may be called as an expert witness, the subject matter about which the expert is expected to testify, the mental impressions and opinions held by the expert and the facts known by the expert (regardless of when the factual information was acquired) which relate to or form the basis for the mental impressions have been used for consultation, but who are not expected to be called as an expert witness, if such consulting expert's opinions or impressions have been reviewed by an expert witness. All discovery will be guided by the Federal Rules of Civil Procedure. All issues concerning discovery upon which the Parties cannot agree will be submitted to the arbitrator for determination.
 - (e) Authority of Arbitrator. The arbitrator will determine the rights and obligations of the Parties according to the laws of the State of Utah. The arbitrator will not have authority to award damages in excess of the amount or other than the types allowed by Section 12.2 and may not, in any event, make any ruling, finding or award that does not conform to the terms and conditions of this Agreement.
 - (f) <u>Joinder of Parties</u>. Each of EDS and Franklin Covey agree that it will use commercially reasonable efforts to join (and will allow the other Party to join) any third party that the Parties have agreed is indispensable to the arbitration. If any such third party does not agree to be joined, the arbitration will proceed nonetheless.
 - (g) Award. The decision of, and award rendered by, the arbitrator will be final and binding on the Parties. Upon the request of a Party, the arbitrator's award will include written findings of fact and conclusions of law.

 Judgment on the award may be entered in and enforced by any court of competent jurisdiction. Each Party will bear its own costs and expenses (including filing fees) with respect to the arbitration, including one-half of the fees and expenses of the arbitrator.
- 10.4 Exclusive Remedy. Other than those matters involving injunctive or other extraordinary relief or any action necessary to enforce the award of the arbitrator, the Parties agree that the provisions of this Article X are a complete defense to any suit, action or other proceeding instituted in any court or before any administrative tribunal with respect to any Dispute or the provision of the Services by EDS. Nothing in this Article X prevents the Parties from exercising their rights to terminate this Agreement in accordance with Article XI.
- Continued Performance; Escrow Account. Unless (a) EDS has commenced a proceeding or has presented a claim pursuant to this Article X for nonpayment by Franklin Covey of amounts due under this Agreement, and Franklin Covey does not promptly pay all amounts in dispute into an escrow account as required below and does not otherwise continue to make payment to EDS in accordance with this Agreement of all amounts not required to be so escrowed, or (b) this Agreement has been terminated in accordance with Article XI, EDS will continue to provide the Services during any dispute resolution proceedings (whether informal or formal) commenced pursuant to this Article X and Franklin Covey will continue to perform its obligations (including the making of payments) in accordance with this Agreement. Up to the lesser of (i) the maximum amount in dispute and (ii) the extent of EDS' liability under Section 12.2(a) as of the date of escrow (the "Escrow Cap"), any disputed payment will be paid pending resolution of the Dispute into an escrow account that is structured by agreement of the Parties or, if agreement cannot be reached, as directed by the mediator or arbitrator, as the case may be, engaged in accordance with this Agriele X. Any such escrow account will provide for the payment of interest on the amounts deposited therein, and the Parties (if the Dispute is resolved informally) or the mediator or arbitrator, as the case may be (if the Dispute is resolved formally), will make the determination regarding distribution of such deposited amounts plus interest. In addition to the escrow, Franklin Covey will continue to make payment to EDS of all amounts not in dispute that are in excess of the Escrow Cap) in accordance with this Agreement. If Franklin Covey fails to escrow disputed payments up to the Escrow Cap or to otherwise continue to make payment to EDS in accordance with this Agreement of all manumts not required to be so escrowed, EDS may apply to any court of competent jurisdiction to seek injunctive relief for such fa

Article XI. <u>Termination</u>

- Dispute Escalation. If either Party materially defaults in the performance of any of its duties or obligations under this Agreement (except for a default in payments to EDS hereunder, which will be governed by Section 11.2), which default is not substantially cured within 60 days after written notice is given to the defaulting Party specifying such default or, with respect to those defaults that cannot reasonably be cured within 60 days, should the defaulting Party fail to proceed within 60 days to commence curing the default and thereafter to proceed with reasonable diligence to substantially cure the default, the Party not in default may, by giving written notice thereof to the defaulting Party, terminate this Agreement as of the date of receipt by the defaulting Party of such notice or as of a future date specified in such notice of termination. In the event of a termination for cause, the sole rights and obligations of the respective Parties shall be those set forth in Section 11.7.
- 11.2 <u>Termination for Nonpayment</u>. If Franklin Covey defaults in the payment when due of any amount due to EDS pursuant to this Agreement and does not cure such default within 30 days after being given written notice of such default or otherwise does not comply with <u>Section 10.5</u>. EDS may, by giving written notice thereof to Franklin Covey, terminate this Agreement as of the date of receipt by Franklin Covey of such notice or as of a future date specified in such notice of termination.
- 11.3 <u>Termination for Bankruptcy and Related Events.</u> Subject to Title 11, United States Code, if either Party becomes or is declared insolvent or bankrupt, is the subject of any proceedings relating to its liquidation, insolvency or for the appointment of a receiver or similar officer for it, makes an assignment for the benefit of all or substantially all of its obligations, then the other Party may, by giving written notice thereof to such Party, terminate this Agreement as of a date specified in such notice of termination.
- 11.4 <u>Termination for Convenience.</u> Subject to the other provisions of this Agreement, Franklin Covey may terminate this Agreement at any time upon at least 6 months prior written notice to EDS; provided however, that if Franklin Covey terminates this Agreement pursuant to this <u>Section 11.4</u>, Franklin Covey shall pay EDS, as EDS 's sole remedy for termination, upon giving such notice the applicable termination fees set forth on <u>Schedule 11.4</u> Section (A) The foregoing notwithstanding, (i) in the event Franklin Covey terminates this Agreement for convenience due to EDS's failure to take the corrective action required by <u>Section 9.6(d)</u> within the time period set forth therein, then the termination fees shall be those set forth in <u>Schedule 11.4</u> Section (B); and (ii) in the event that Franklin Covey terminates this Agreement for convenience within 12 months of a change of control (other than through one or more public offerings of Franklin Covey securities) where such control is acquired, directly or indirectly, in a single transaction or series of related transactions at any time during the Term (except that such change of control shall be measured during any 6 month period only in respect of any period when Franklin Covey shall be publicly owned), or all or substantially all of the assets of Franklin Covey are acquired by one or more persons, or Franklin Covey is merged or consolidated with or into any other person to form a new entity, then the termination fees shall be those set forth in Schedule 11.4 Section (C).

11.5 Effect of Expiration or Termination.

- (a) Wind-Down. Upon expiration or termination of this Agreement for any reason, EDS will cease to perform the Services, and Franklin Covey will pay to EDS all amounts due to EDS for all Services provided and expenses incurred (including those expenses that, instead of being concurrently billed, have been included in future payments to be made by Franklin Covey) through the effective date of such expiration or termination.
- (b) Transition. In connection with the expiration of this Agreement on the Expiration Date or with the termination of this Agreement by Franklin Covey pursuant to this Article XI. EDS will comply with Franklin Covey's reasonable directions to cause the orderly transition and migration from EDS to Franklin Covey (or a third party services provider undertaking, on behalf of Franklin Covey, to provide the Services (the "Third Party Provider")) of all Services then being performed by EDS (the "Termination Transition"). The Termination Transition will be provided for a reasonable period of time which in no event will exceed 12 months. Franklin Covey will cooperate in good faith with EDS in connection with EDS' obligations under this Section 11.5(b) and will perform its obligations under the Transition Plan. If the Termination Transition extends beyond the Expiration Date or effective date of termination of this Agreement by Franklin Covey, the provisions of this Agreement will remain in effect for the duration of the Termination Transition and will apply to all transition assistance services provided by EDS during such period (subject to the last two sentences of Section 11.5(d). EDS will perform the following obligations (and such other obligations as may be contained in the Transition Plan) at Franklin Covey's expense, unless otherwise stated below or in Schedule 11.5. Franklin Covey acknowledges and agrees that, as indicated above in this Section 11.5(b), EDS will have no obligation to provide any form of Termination Transition if EDS terminates this Agreement pursuant to this Article XI.
 - (i) <u>Transition Plan.</u> EDS and Franklin Covey will work together to develop a transition plan (the "Transition Plan") setting forth the respective tasks to be accomplished by each Party in connection with the Termination Transition and a schedule pursuant to which such tasks are to be completed.
 - (ii) Specifications. EDS will, upon Franklin Covey's request, provide Franklin Covey with reasonably detailed specifications for the hardware and software needed by Franklin Covey (or the Third Party Provider) to properly provide the Services then being performed by EDS.
 - (iii) Acquisition of Equipment. Franklin Covey will have the option to purchase from EDS at its fair market value any equipment owned by EDS and to assume the leases (provided such leases are assumable) for any equipment leased by EDS, in each case only to the extent that such equipment is dedicated solely to providing the Services to Franklin Covey as of the Expiration Date or the effective date of termination, whichever is applicable. Such equipment will be provided on an "AS IS, WHERE IS" basis.
 - (iv) Licensed Programs. The parties will execute a mutually acceptable license agreement, and EDS will deliver a copy of the Licensed Programs and existing documentation to Franklin Covey and will install such copy on the equipment of Franklin Covey (or the Third Party Provider).
 - (v) <u>Vendor Software.</u> EDS will reasonably assist Franklin Covey in Franklin Covey's acquisition of any necessary rights to access any EDS-Vendor Software and existing documentation then being used by EDS in performing the Services.
 - (vi) <u>Training.</u> EDS will provide appropriate training for the employees of Franklin Covey (or the Third Party Provider) who will be assuming responsibility following the Termination Transition for operation of the software then being used by EDS in performing the Services.
 - (vii) Hiring of Employees. Notwithstanding Section 2.4, Franklin Covey may offer employment to any Account Team member who is then assigned to the Franklin Covey account and is not providing Services from a leveraged support center..
- (c) <u>Disclosure of Information</u>. Notwithstanding anything to the contrary in this Agreement, EDS will not be required to disclose any of its proprietary information, whether in the nature of a trade secret, software or otherwise, to the Third Party Provider except to the extent that Franklin Covey is entitled to such information under this Agreement. Prior to providing any termination assistance to the Third Party Provider, Franklin Covey will cause the Third Party Provider to provide EDS with written assurances, in form and substance reasonably satisfactory to EDS, that the Third Party Provider (i) will maintain at all times the confidentiality of any EDS proprietary information disclosed or provided to, or learned by, the Third Party Provider in connection therewith and (ii) will use such information exclusively for the purposes for which Franklin Covey is authorized to use such information pursuant to this Agreement.
- (d) Charges. For so long as this Agreement remains in effect and during the Termination Transition but subject to the last two sentences of this Section 11.5(d), Franklin Covey will pay to EDS the charges set forth in this Agreement. If the Termination Transition provided by EDS under this Section 11.5 requires personnel or other resources in excess of those resources then being provided by EDS under this Agreement, Franklin Covey will pay EDS for such additional resources at the applicable resource rates set forth in Schedule 9.1 on such periodic basis as the Parties agree in writing. Notwithstanding anything to the contrary in this Agreement, all charges to be paid by Franklin Covey to EDS during the Termination Transition (including those set forth in Schedules 9.1 and 11.4) will be paid on a monthly basis in advance.
- (e) <u>Franklin Covey's Right to Extend the Termination Date</u>. Except in the case of a valid termination for cause by EDS under <u>Section 11.2</u> (unless Franklin Covey has paid EDS all amounts then due and payable in respect of the Services and has agreed to pay in advance for the future Services to be provided), Franklin Covey may elect, upon 90 days' prior written notice to EDS, to extend the Termination Date one or more times, in its sole and absolute discretion, provided that the total of all such extensions shall not exceed 12 months following the original Termination Date without EDS 's prior written consent.

Article XII. Indemnities and Liability

2.1 Indemnities

(a) Claims Relating to Space, Taxes and Employees. Subject to Sections 12.1(f) and 12.2, EDS and Franklin Covey each agrees to indemnify and defend the other Party from any and all claims, actions, damages, liabilities, costs and expenses, including reasonable attorneys' fees and expenses (collectively, "Losses"), arising out of, under or in connection with (i) any claim for rent or utilities at any location where the indemnitor is financially responsible under this Agreement for such rent or utilities, (ii) any claim for wages, benefits, third party fees, taxes, assessments, duties, permits or other charges of any nature for which the indemnitor is financially responsible under this Agreement, as well as penalties, interest, fees or other expenses incurred by the indemnitor as a result of such charges not being paid at the time or in the manner required by applicable law, (iii) an act or omission, or misrepresentation of fact or information upon which the indemnitee reasonably relied, which resulted in any erroneous payment from, or any defect in the administration or operation of, any employee benefit plan or employee benefit program maintained by the indemnitee, or which resulted in any fine, penalty or correction imposed, assessed or required by a governmental agency or which is reasonably required by the indemnitee to avoid a fine, penalty or corrective action or (iv) an act or omission of the indemnitor in its capacity as an employee of faree or a Transitioned Employee and arising out of or relating to (A) federal, state or other laws or regulations for the protection of persons who are members of a protected class or category of persons, (B) sexual discrimination or harassment, (C) accrued employment and which, with respect to each of clauses (A) through (D), arose when the person asserting the claim, demand, charge, actions, cause of action or other proceeding was or purported to be an employee of the indemnitor. Franklin Covey acknowledges and agrees that its employee indemnity obligation set forth abo

(b) Claims Relating to Personal Injury and Property Damage.

- (i) General. EDS and Franklin Covey each will be responsible for Losses to their respective tangible personal or real property (whether owned or leased), and each Party agrees to look only to its own insuring arrangements (if any) with respect to such Losses. EDS and Franklin Covey each will be responsible for Losses for the death of or personal injury to any person (including any employee of either Party) and Losses for damages to any third party's tangible personal or real property (whether owned or leased), in accordance with the law of the jurisdiction in which such Loss is alleged to have occurred. Subject to Sections 12.1(f) and 12.2, each Party will indemnify and defend the other Party from any and all Losses arising out of, under or in connection with claims for which the indemnitor is responsible under the preceding sentence.
- (ii) Waiver of Subrogation. EDS and Franklin Covey waive all rights to recover against each other for any Loss to their respective tangible personal property (whether owned or leased) from any cause covered by insurance maintained by each of them, including their respective deductibles or self-insured retentions. EDS and Franklin Covey will cause their respective insurers to issue appropriate waivers of subrogation rights endorsements to all property insurance policies maintained by each Party. Each Party will give the other written notice if a waiver of subrogation is unobtainable or obtainable only at additional expense. If the Party receiving such notice agrees to reimburse the other Party for such additional expense, the other Party will obtain such waiver of subrogation. If a waiver is unobtainable or if a Party elects not to pay the additional expense of a waiver, then neither Party nor their insurers will waive such subrogation rights.

(c) <u>Infringement Claims</u>

- (i) General. Subject to Sections 12.1(f) and 12.2 and the limitations set forth below in this Section 12.1(c), EDS and Franklin Covey each agrees to defend the other Party against any third party action to the extent that such action is based upon a claim that the software (other than third party software) or confidential information provided by the indemnitor, or any part thereof, (i) infringes a copyright perfected under United States statute, (ii) infringes a patent granted under United States law or (iii) constitutes an unlawful disclosure, use or misappropriation of another party's trade secret. The indemnitor will bear the expense of such defense and pay any Losses that are attributable to such claim finally awarded by a court of competent jurisdiction.
- (ii) <u>Exclusions</u>. Neither EDS nor Franklin Covey will be liable to the other for claims of indirect or contributory infringement. The indemnitor will have no liability to the indemnite hereunder if (A) the claim of infringement is based upon the use of software provided by the indemnitor hereunder in connection or in combination with equipment, devices or software not supplied by the indemnitor or used in a manner for which the software was not designed, (B) the indemnite modifies any software provided by the indemnitor hereunder and such infringement would not have occurred but for such modification, or uses the software in the practice of a patented process and there would be no infringement in the absence of such practice, or (C) the claim of infringement arises out of the indemnitor's compliance with specifications provided by the indemnitee and such infringement would not have occurred but for such compliance.
- (iii) Additional Remedy. If software or confidential information becomes the subject of a claim under this Section 12.1(c), or in the indemnitor's opinion is likely to become the subject of such a claim, then, in addition to defending the claim and paying any damages and attorneys' fees as required above in this Section 12.1(c), the indemnitor may, at its option and in its sole discretion, (A) replace or modify the software or confidential information to make it noninfringing or cure any claimed misuse of another's trade secret or (B) procure for the indemnite the right to continue using the software or confidential information pursuant to this Agreement. Any costs associated with implementing either of the above alternatives will be borne by the indemnitor but will be subject to Section 12.2. If neither alternative is pursued by, or (if pursued) available to, the indemnite will return such software or confidential information to the indemnitor and (y) if requested by the indemnitee in good faith, the Parties will negotiate, pursuant to Article X but subject to Section 12.2.(t) or each a written agreement on what, if any, monetary damages (in addition to the indemnitor's obligation to defend the claim and pay any damages and attorneys' fees as required above in this Section 12.1(c)) are reasonably owed by the indemnitor to the indemnitor to implement either of the above alternatives.

(d) <u>Claims Relating to Internet Usage.</u>

(i) Franklin Covey warrants that the publication of any material delivered by or through it hereunder will not violate the copyright laws of the United States or any other jurisdiction, unlawfully infringe or interfere in any way with the literary property or rights of another or contain libelous or indecent matter. Subject to Sections 12.1(f) and 12.2. Franklin Covey will indemnify and defend EDS from any and all Losses, including those associated with claims for indirect or contributory infringement, arising out of, under or in connection with any claims relating to (i) content, whether of an editorial, advertising or other nature, (ii) the provision, use, alteration or distribution thereof, the accessibility thereto or the exchange of information over the Internet in connection therewith, including copyright infringement, libel, indecency, false light, misrepresentation, invasion of privacy or image or personality rights, (iii) statements or other materials made or made available by readers of the content or by persons to whom the content is linked at the request of Franklin Covey or (iv) the conduct of Franklin Covey's business using the Internet.

- (ii) EDS warrants that, except as to materials covered by the terms of Section 12.1(d)(1) above, the publication of any material by EDS in performing Services hereunder will not violate the copyright laws of the United States or any other jurisdiction, unlawfully infringe or interfere in any way with the literary property or rights of another or contain libelous or indecent matter. Subject to Sections 12.1(f), and 12.2. EDS will indemnify and defend Franklin Covey from any and all Losses, including those associated with claims for indirect or contributory infringement, arising out of, under or in connection with any claims relating to (i) connent, whether of an editorial, advertising or other nature, (ii) the provision, use, alteration or distribution thereof, the accessibility thereto or the exchange of information over the Internet in connection therewith, including copyright infringement, libel, indecency, false light, misrepresentation, invasion of privacy or image or personality rights, (iii) statements or other materials made or made available by readers of the content or by persons to whom the content is linked at the request of EDS or (iv) the conduct of EDS' business using the Internet.
- (e) Third Party Indemnification of EDS. Without limiting EDS' liability to Franklin Covey under this Agreement, each of the Parties acknowledges that by entering into and performing its obligations under this Agreement EDS will not assume and should not be exposed to the business and operational risks associated with Franklin Covey's business, and Franklin Covey therefore agrees, subject to Sections 12.1(f) and 12.2, to indemnify and defend EDS from any and all third party Losses arising out of the conduct of Franklin Covey's business, including the use by Franklin Covey of the Services.
- (f) <u>Procedures</u>. The indemnification obligations set forth in this Section 12.1 will not apply unless the Party claiming indemnification:
 - (i) Notice. Notifies the other promptly in writing of any matters in respect of which the indemnity may apply and of which the notifying Party has knowledge in order to allow the indemnitor the opportunity to investigate and defend the matter; provided, however, that the failure to so notify will only relieve the indemnitor of its obligations under this Section 12.1 if and to the extent that the indemnitor is prejudiced thereby; and
 - (ii) Control of Defense. Gives the other Party full opportunity to control the response thereto and the defense thereof, including any agreement relating to the settlement thereof; provided, however, that the indemnitee will have the right to participate in any legal proceeding to contest and defend a claim for indemnification involving a third party and to be represented by legal counsel of its choosing, all at the indemnitee's cost and expense. However, if the indemnitor fails to promptly assume the defense of the claim, the Party entitled to indemnification may assume the defense at the indemnitor's cost and expense.

The indemnitor will not be responsible for any settlement or compromise made without its consent, unless the indemnitee has tendered notice and the indemnitor has then refused to assume and defend the claim and it is later determined that the indemnitor was liable to assume and defend the claim. The indemnitee agrees to cooperate in good faith with the indemnitor at the request and expense of the indemnitor.

2.2 <u>Liability</u>

- (a) General Limitation. The liability of each Party to the other for all damages arising out of or related to this Agreement, regardless of the form of action that imposes liability, whether in contract, equity, negligence, intended conduct, tort or otherwise, will be limited to and will not exceed, in the aggregate for all claims, actions and causes of action of every kind and nature, the sum of \$10,000,000.00.
- (b) <u>Limitation on Other Damages</u>. In no event will the measure of damages payable by either Party include, nor will either Party be liable for, any amounts for loss of income, profit or savings or indirect, incidental, consequential, exemplary, punitive or special damages of any party, including third parties, even if such Party has been advised of the possibility of such damages in advance, and all such damages are expressly disclaimed.
- (c) Exceptions to Limitations. The limitations set forth in Sections 12.2(a) and 12.2(b), will not apply to (i) the liability of either Party to the extent such liability results from (A) that Party's acts of intentional tortious conduct or gross negligence in the performance or nonperformance of its obligations under this Agreement or (B) that Party's nonperformance of its payment obligations to the other expressly set forth in this Agreement (including, with respect to Franklin Covey, Franklin Covey's obligation to make payments to EDS during the original term of this Agreement as required hereby, whether in the form of charges for Services performed hereunder or for payment or reimbursement of taxes, out-of-pocket expenses or pass-through expenses, and EDS' lost profits on such payments) or (ii) the liability of Franklin Covey under Article IX.
- (d) <u>Duty to Mitigate</u>. Each Party has a duty to mitigate the damages that would otherwise be recoverable from the other pursuant to this Agreement by taking appropriate and reasonable actions to reduce or limit the amount of such damages.
- (e) Contractual Statute of Limitations. No claim and demand for mediation or arbitration or cause of action which arose out of an event or events which occurred more than two years prior to the filing of a demand for mediation or arbitration or suit alleging a claim or cause of action may be asserted by either Party against the other.
- (f) Acknowledgment. The Parties expressly acknowledge that the limitations and exclusions set forth in this Section 12.2 have been the subject of active and complete negotiation between the Parties and represent the Parties' agreement taking into account each Party's level of risk associated with the performance or nonperformance of its obligations under this Agreement and the payments and other benefits to be derived by each Party pursuant to this Agreement. The provisions of this Section 12.2 will survive the expiration or termination of this Agreement for any reason.

Article XIII. Miscellaneous

- 13.1 Right to Engage in Other Activities. Subject to Section 2.7 of this Agreement, Franklin Covey acknowledges and agrees that EDS may provide information technology services for third parties at any EDS facility that EDS may utilize from time to time for performing the Services. Subject to the restrictions on the use of data and the disclosure of confidential information set forth in Article VIII, nothing in this Agreement will impair EDS' right to acquire, license, market, distribute, develop for itself or others or have others develop for EDS similar technology performing the same or similar functions as the technology and Services contemplated by this Agreement.
- 13.2 Independent Contractors. The Parties are independent contractors, and this Agreement will not be construed as constituting either Party as partner, joint venturer or fiduciary of the other, as creating any other form of legal association that would impose liability on one Party for the act or failure to act of the other or as providing either Party with the right, power or authority (express or implied) to create any duty or obligation of the other. Except as otherwise expressly provided in this Agreement, each Party has the sole right and obligation to supervise, manage, contract, direct, procure, perform or cause to be performed all work to be performed by it pursuant to this Agreement.
- 13.3 Entire Agreement; Survival. This Agreement (including the Schedules attached hereto, each of which is incorporated into this Agreement by this reference) constitutes the full and complete statement of the agreement of the Parties with respect to the subject matter hereof and supersedes any previous or contemporaneous agreements, understandings or communications, whether written or oral, relating to such subject matter. The expiration or termination of this Agreement for any reason will not release either Party from any liabilities or obligations set forth herein which (a) the Parties have expressly agreed will survive any such expiration or termination or (b) remain to be performed or by their nature would be intended to be applicable following any such expiration or termination.
- 13.4 Amendments; Waiver. Changes or modifications to this Agreement may not be made orally or through a course of dealing, but only by a written amendment or revision signed by the Parties. Any terms and conditions varying from this Agreement on any order, invoice or other notification from either Party are not binding on the other unless specifically accepted by the other. Unless otherwise expressly provided in this Agreement, neither a delay nor omission by either Party to exercise, nor a course of dealing with respect to any right or power under this Agreement will be construed to be a waiver thereof. No waiver of any prior, concurrent or subsequent breach of the same or any other provision hereof.
- Binding Nature; Assignment. This Agreement will be binding on the Parties and their successors and permitted assigns (it being understood and agreed that nothing contained in this Agreement is intended to confer upon any party other than EDS and Franklin Covey any rights, benefits or remedies of any kind or character whatsoever under or by reason of this Agreement). Except as otherwise expressly provided in Article II and except as noted below, neither Party may, nor will it have the power to, assign this Agreement, or any part hereof, without the consent of the other. EDS may assign its rights to Franklin Covey's payments hereunder, including any termination amount payable pursuant to Article XI. to a financial institution or other third party in connection with any transaction entered into to provide financing related to this Agreement or the obligations of EDS hereunder, and any such assignee may further assign its rights hereunder in connection with such financing. The Parties acknowledge that either of them may become a party to one or more transactions in the form of a merger (including a reincorporation merger), consolidation, reorganization, stock sale or exchange, sale of all or substantially all of such Party's assets or some similar or related transaction, with the result being that the affected Party is the surviving entity assumes the rights and obligations under this Agreement or, if the affected Party is not the surviving entity, the surviving entity continues to conduct the business conducted by the affected Party prior to consummation of the transaction, including the assumption of the rights and obligations under this Agreement. No such transaction involving either Party will be deemed to be an assignment of this Agreement requiring the consent of the other, unless (a) in the case of EDS being involved in such a transaction, the transaction materially and adversely affects EDS' ability to continue to perform the Services in accordance with this Agreement or (b) in the case of Franklin Covey's
 - Export Regulations. This Agreement is expressly made subject to any United States government laws, regulations, orders or other restrictions regarding export from the United States of computer hardware, software, technical data or derivatives of such hardware, software or technical data. Notwithstanding anything to the contrary in this Agreement, Franklin Covey will not directly or indirectly export (or reexport) any computer hardware, software, technical data or derivatives of such hardware, software or technical data, or permit the shipment of same: (a) into (or to a national or resident of) Cuba, North Korea, Iran, Iraq, Libya, Syria or any other country to which the United States has embargoed goods; (b) to anyone on the U.S. Treasury Department's List of Specially Designated Narcotics Traffickers, or the U.S. Commerce Department's Denied Parties List; or (c) to any country or destination for which the United States government or a United States governmental agency requires an export license or other approval for export without first having obtained such license or other approval. Each Party will reasonably cooperate with the other and will provide to the other promptly upon request any end-user certificates, affidavits regarding reexport or other certificates or documents as are reasonably requested to obtain approvals, consents, licenses and/or permits required for any payment or any export or import of products or services under this Agreement. The provisions of this Section 13.6 will survive the expiration or termination of this Agreement for any reason.
- 13.7 Approvals and Similar Actions. Except as otherwise expressly provided in this Agreement, where agreement, approval, acceptance, consent or similar action is required of either Party by any provision of this Agreement, such action will not be unreasonably withheld or delayed. An approval or consent given by a Party under this Agreement will not relieve the other Party from responsibility for complying with the requirements of this Agreement, nor will it be construed as a waiver of any rights under this Agreement, except as and to the extent otherwise expressly provided in such approval or consent.
- Notices. Except as otherwise expressly provided in this Agreement, all notices under this Agreement will be in writing and will be deemed to have been duly given if delivered personally or by a nationally recognized courier service, faxed or mailed by registered or certified mail, return receipt requested, postage prepaid, to the Parties at the addresses set forth in Schedule 13.8. All notices under this Agreement that are addressed as provided in this Section 13.8. (a) if delivered personally or by a nationally recognized courier service, will be deemed given upon delivery, (b) if delivered by facsimile, will be deemed given when confirmed and (c) if delivered by mail in the manner described above, will be deemed given on the fifth business day after the day it is deposited in a regular depository of the United States mail. Either Party from time to time may change its address or designee for notification purposes by giving the other Party notice of the new address or designee and the date upon which such change will become effective.
- 13.9 Excused Performance. Neither Party will be deemed to be in default hereunder, or will be liable to the other, for failure to perform any of its non-monetary obligations under this Agreement for any period and to the extent that such failure results from any event or circumstance beyond that Party's reasonable control (each, a Force Majeure Event"), including acts or omissions of the other Party or third parties, natural disasters, riots, war, civil disorder, court orders, acts or regulations of governmental bodies, labor disputes or failures, and which it could not have prevented by reasonable precautions or could not have remedied by the exercise of reasonable efforts, provided that the exercise of such reasonable precautions or reasonable efforts will not require the incurrence of any additional cost or expense. To the extent that disaster recovery services are included in the Services, the foregoing will not limit EDS' obligation to provide such services unless they also are affected by the Force Majeure Event.
- 13.10 <u>Public Relations and Marketing References.</u> Each Party will coordinate with the other regarding any media release, public announcement or similar disclosure relating to this Agreement or its subject matter and will give the other Party a reasonable opportunity to review and comment on the content of such release, announcement or disclosure prior to its release. This provision does not alter the restrictions on the disclosure of confidential information set forth in <u>Section 8.4</u>, will not be construed so as to delay or restrict either Party from disclosing any information required to be disclosed in order to comply with any applicable laws, rules or regulations. Notwithstanding the foregoing but subject to any applicable laws, rules or regulations, each Party will have the right to list the name of the other Party, to make general references to the basic nature of the relationship between the

Parties under this Agreement and to describe generally the type of services being provided by EDS to Franklin Covey under this Agreement in such Party's promotional and marketing materials, in such Party's oral or visual presentations to third parties, in interviews conducted by the news media or securities analysts and in or through any other available media channels, including print, Internet, radio, cable and broadcast mediums.

- 13.11 Construction Rules. The Article and Section headings and table of contents used in this Agreement are for convenience of reference only and will not enter into the interpretation hereof. As used in this Agreement, unless otherwise expressly provided to the contrary, (a) any reference to a "Section", "Article" or "Schedule" is a reference to a Section or Article of this Agreement or a Schedule attached to this Agreement, and (b) all references to days, months or years are references to calendar days, months or years. To the extent that the provisions of this Agreement and the Schedules are inconsistent, to the extent possible such provisions will be interpreted so as to make them consistent, and if that is not possible, the provisions of the Schedules will prevail. If any provision of this Agreement is held to be invalid, illegal or unenforceable, the validity, legality and enforceability of the remaining provisions will be in any way be affected or impaired, and such provision will be deemed to be restated to reflect the original intentions of the Parties as nearly as possible in accordance with applicable law. The Parties agree that this Agreement is an executory contract as contemplated by 11 U.S.C. Section 365. In performing its obligations under this Agreement, neither Party will be required to undertake any activity that would conflict with the requirements of any applicable law, rule, regulation, interpretation, judgment, order or injunction of any governmental authority. This Agreement may be executed in multiple counterparts, each of which will be deemed an original and all of which taken together will constitute one instrument. The Parties acknowledge and agree that each has been represented by legal counsel of its choice throughout the negotiation and drafting of this Agreement, that each has participated in the drafting hereof and that this Agreement will not be construed in favor of or against either Party solely on the basis of a Party's drafting or participation in the drafting of any portio
- 13.12 Accounting Terms. All accounting terms that are not specifically defined herein, and all accounting obligations, duties, procedures and practices that are undertaken pursuant to this Agreement, shall be construed, interpreted and undertaken in accordance with generally accepted accounting principles that exist in the United States of America from time to time.
- 13.13 <u>Governing Law</u>. This Agreement will be governed by and construed in accordance with the substantive laws of the State of Utah, without giving effect to any choice-of-law rules that may require the application of the laws of another jurisdiction.

In Witness Whereof, the Parties have duly executed and delivered this Agreement by their duly authorized representatives as of the Effective Date.

FRANKLIN COVEY CO	ELECTRONIC DATA SYSTEMS
By:	By:
Title:	Title:
Date:	Date:
	EDS INFORMATION SERVICES L.L.C.
	By:
	Title:
	Date:

Schedule of Definitions

Defined Term	Location of Definition
AAA	Section 10.2
Access	Section 5.5
Additional Services	Section 3.6(a)
Agreement	Introduction
Article	Section 13.11
Base Line Assessment	Section 3.3(a)
BLS	Schedule 9.7, Section 1
Collected Taxes	Section 9.9(a)
Copy/Copies	Schedule 7.6 [7.6(b)], Introduction
Developed Software	Section 7.1(e)
Dispute	Section 10.1
ECI	Schedule 9.7, Section 1
ECI Adjustable Charges	Schedule 9.7, Section 1
ECI Base Index	Schedule 9.7, Section 1
ECI Current Index	Schedule 9.7, Section 1
EDS	Introduction
EDS Account Executive	Section 1.1
EDS-Acquired Equipment	Section 5.1
EDS Development Tools	Section 7.1(e)
EDS Services	Section 3.2
EDS Software	Section 7.1(a)
EDS-Vendor Software	Section 7.1(b)
Effective Date	Introduction
EIS	Introduction
Employee Offeree	Section 2.1
Equipment Leases	Section 5.2
Escrow Cap	Section 10.5
Expiration Date	Section 3.1

Force Majeure Event	Section 13.9
Hewitt Index Adjustable Charges	Schedule 9.7, Section 4
Index	Schedule 9.7, Section 4
Items	Section 6.1(e)
Key Positions	Section 2.2(b)
Licensed Programs	Section 7.2
Losses	Section 12.1(a)
Maintenance Contracts	Section 5.4
Nonconformance	Section 6.1(g)
Other Deliverables	Section 7.1(h)
Party/Parties	Introduction
Percent Change	Schedule 9.7, Section 4
Privacy Laws	Section 8.3
Residual Technology	Section 7.1(e)
Schedule	Section 13.11
Section	Section 13.11
Service Conversion Date	Schedule 9.4, Section 3
Service Levels	Section 3.3(c)
Services	Section 3.2
Software	Section 7.1(f)
Software Changes	Section 7.1(f)
Software Licenses	Section 5.3
Steering Committee	Section 1.3
Termination Transition	Section 11.4(b)
Third Party Provider	Section 11.4(b)
Transition Plan	Section 11.4(b)(i)
Transitioned Employees	Section 2.1
Vendor	Schedule 3.5, Section 1(a)
Franklin Covey	Introduction
Franklin Covey Data	Section 8.1
Franklin Covey-Leased Equipment	Section 5.5
Franklin Covey-Owned Equipment	Section 5.5
Franklin Covey Representative	Section 1.2
Franklin Covey Software	Section 7.1(c)
Franklin Covey-Vendor Software	Section 7.1(d)

Schedule 9.1 Charges

INTRODUCTION

- 1. <u>Charges Sections.</u> This <u>Schedule 9.1</u> contains EDS' fees, charges and rates for the Agreement. The Charges Sections contains a Resource Baseline Section, a Monthly Charges Section, a Variable Charges Section, and an Inflation Adjustment Section.
 - (a) Resource Baseline Section. The Resource Baseline Section sets forth the quantity of resources and other items for services contained in the Monthly Charges (as defined below). Such quantity of resources and other items is referred to herein as the "Resource Baseline". The Resource Baseline is set forth for each calendar year during the Initial Term. The Resource Baseline for the last year of the Initial Term shall continue for each month during which EDS provides termination assistance to Franklin Covey pursuant to Section 11.5 of this Agreement.
 - (b) Monthly Charges Section. The Monthly Charges Section sets forth the monthly charges to be paid by Franklin Covey to EDS for the Services applicable to the Resource Baselines. The Monthly Charges Section shows the billing period to which the base charges apply, the date on which such base charges will be invoiced by EDS to Franklin Covey and various components of such charges. The total of all such components are indicated in the last column of the Monthly Charges Section and such total is referred to herein as the "Baseline Monthly Charges". The Baseline Monthly Charges for each month during which EDS provides termination assistance to Franklin Covey pursuant to Section 11.5 of this Agreement shall be equal to the Monthly Charges for the last month of the last year of the Initial Term.
 - (c) Variable Charges Section. The Variable Charges Section sets forth the Additional Resource Charges ("ARCs") and the Reduced Resource Credits ("RRCs") that will be applied to the Baseline Monthly Charges as the quantity of the items included in the Services provided by EDS varies from the Resource Baseline by month. The method by which ARCs and RRCs are applied to the Baseline Monthly Charges is further described in paragraph 4, below. The ARCs and RRCs for the last month of the Initial Term shall continue for each month during which EDS provides termination assistance to Franklin Covey pursuant to Section 11.5 of this Agreement.
 - (d) Inflation Adjustment Section. The Inflation Adjustment Section sets forth the method by which all of the charges, fees and rates set forth in this Schedule will be increased during the term of the Agreement.
- Minimum Monthly Charges. Notwithstanding anything to the contrary in this Agreement or otherwise, Franklin Covey will be obligated to pay EDS at least the Minimum Monthly Charges, which are defined to be eighty percent (80%) of the then-current Baseline Monthly Charges, even if the application of RRCs would reduce the amount payable by Franklin Covey to EDS during a month to below the Minimum Monthly Charges, for two consecutive calendar months (the "Two Months"), either (a) Franklin Covey has paid to EDS the Minimum Monthly Charges even though the application of RRCs would have reduced the amount payable by Franklin Covey to EDS for each of such months to below the Minimum Monthly Charges, or (b) Franklin Covey has paid to EDS one hundred twenty percent (120%) of the then current Baseline Monthly Charges due to the application of ARCs, and Franklin Covey reasonably anticipates that the same will continue for the following two calendar months, then, at the written request of Franklin Covey, Franklin Covey and EDS will meet for the purpose of negotiating in good faith adjustments to the Minimum Monthly Charges, the Baseline Monthly Charges, and the ARCs and RRCs and to any other terms of this Agreement that either party deems appropriate in connection therewith. In conducting such negotiations, the parties will take into account the total charges paid by Franklin Covey to EDS for Services, the reasons for Franklin Covey's reduced or increased demand for the Services, any written commitments made by Franklin Covey to obtain Additional Services from EDS, and EDS' expectation at the time of the Effective Date for the revenue it was to receive for Services at the Resource Baseline. Any adjustments to the Charges and the other terms of this Agreement upon which the parties agree will be set forth in a written amendment to this Agreement executed by the parties and will be retroactive to the beginning of the Two Months. If the parties are unable to reach agreement within sixty

- (60) days following EDS' receipt of Franklin Covey's request for meetings, then the matter shall be submitted to the Steering Committee for resolution. If, within thirty (30) days thereafter, the Steering Committee is unable to resolve such matter, then either party may initiate the dispute resolution process as set forth in Article X.
- 3. Application of Variable Charges (ARCs and RRCs). If the actual quantity of items included in the Services provided by EDS during any calendar month are higher than the Resource Baseline for the applicable resource category designated in the Resource Baseline Section of this Schedule 9.1, then Franklin Covey will pay to EDS an additional amount calculated by multiplying the ARC for the applicable resource category times the number of items in excess of the Resource Baseline for that resource category in the applicable month. If the actual quantity of items included in the Services provided by EDS during any calendar month are less than the Resource Baseline for the applicable resource category designated in the Resource Baseline Section of this Schedule 9.1, then EDS will credit to Franklin Covey an amount calculated by multiplying the RRC for the applicable resource category times the number of items below the Resource Baseline for that resource category in the applicable month. ARCs and RRCs will be itemized in the invoices sent by EDS to Franklin Covey.

Schedule 9.1 Annual Adjustment to Charges

- Adjustment to Charges. Unless otherwise expressly provided in an Authorization Letter with respect to the charges to be paid thereunder, the Parties acknowledge and agree to use the Employment Cost Index for Total Compensation (not seasonally adjusted), Private Industry Workers, White-collar occupations excluding sales, June 1989 = 100 (the "ECI"), as the basis for annual adjustments to charges to be paid by Franklin Covey to EDS under this Agreement for those Services listed in Section 5 of this Schedule 9.1 (the "ECI Adjustable Charges"). The ECI is published by the Bureau of Labor Statistics (the "BLS") of the U.S. Department of Labor. For purposes of this Schedule 9.1, the most recently published ECI as of any anniversary of the Effective Date is the "ECI Current Index", and the "ECI Base Index" is the ECI Current Index from the prior anniversary of the Effective Date (or, for the first anniversary, the ECI adjustable Charges will be made by increasing the ECI Adjustable Charges by the percentage that the ECI Current Index is more than 2% higher than the ECI Base Index, then, effective as of such anniversary, an adjustment to the ECI Adjustable Charges will be made by increasing the ECI Current Index is lower than 102% of the ECI Base Index, no adjustment to the ECI Adjustable Charges will be made to the ECI Current Index is lower than 102% of the ECI Base Index, no adjustment to the ECI Adjustable Charges will be made in the manner indicated in the example set forth in Section 3 of this Sechedule 9.1. If an adjustment to a full year will be made in the manner indicated in the example set forth in Section 3 of this Sechedule 9.1. If an adjustment to a full year will be made in the manner indicated in the example set forth in Section 3 of this Sechedule 9.1. If an adjustment was made, as indicated in the note to the third example set forth in Section 3 of this Sechedule 9.1. If an adjustment was made, as indicated in the note to the third example set forth in Section 3 of this Sechedule 9.1. The ECI is publi
- Adjustment to Charges Example. The following is an example of the adjustments described in Section 1 of this Schedule 9.1. The specific numbers used in the example are for illustration purposes only and are not necessarily reflective of an actual calculation hereunder or the actual ECI.

```
Annual Adjustment on First Anniversary Date:
     Example Charge under this Agreement
                                                                                  $1,500.00
      ECI Current Index
                                                                                       136.0
     ECI Base Index (as of Effective Date)
                                                                                       129.9
     Percentage Change
Charge Increased by (1+ Percentage Change-2%)
Equals Adjusted Charge
                                                           (136.0 - 129.9) / 129.9 = 4.7\%
                                                                             (1 + 4.7%-2%)
$1,540.50
                                                                $1,500.00
Annual Adjustment on Second Anniversary Date:
     Adjusted Charge as of First Anniversary Date
                                                                                  $1.540.50
      ECI Current Index
                                                                                       143.2
     ECI Base Index (as of First Anniversary Date)
                                                                                       136.0
                                                          (143.2 - 136.0) / 136.0 = 5.3%

$1.540.50 * (1 + 5.3%-2%)
      Percentage Change
     Charge Increased by (1+ Percentage Change-2%)
                                                                $1,540.50
                                                                             (1 + 5.3\% - 2\%)
     Equals Adjusted Charge
                                                                                  $1,591.34
Annual Adjustment on Third Anniversary Date:
      Adjusted Charge as of Second Anniversary Date
                                                                                  $1,591.34
     ECI Current Index
                                                                                       144.9
      ECI Base Index (as of Second Anniversary Date)
                                                                                       143.2
     Percentage Change
                                                           (144.9 - 143.2) / 143.2 = 1.2%
     Charge Increased by (1+ Percentage Change-2%)
Equals Adjusted Charge*
                                                                                  No change
                                                                                   No Change
```

- * The ECI Base Index for the fourth anniversary date would be 146.1.
- 3. Changes to Index. In the event that the BLS should stop publishing the ECI or should substantially change the content, format or calculation methodology of the ECI, the Parties will substitute another comparable measure published by a mutually agreeable source, except as noted below. If the change is to redefine the base period for the ECI from one period to some other period, the Parties will continue to use the index but will use the new base period figures for all future adjustments. If the change is to the name of the ECI, the new name will be used instead of the old name so long as the numbers previously published for the index have not changed. If the change is to the publication schedule, the Parties may agree in writing to use a different publication schedule and to adjust any partial year to a full year, if needed. The adjustment to convert a partial year to a full year is shown below.

```
ECI Current Index as of July 19XX

ECI Base Index as of June 19XX-1

Percentage Change (rounded to 3 decimals)

Percent Times 12 Divided # of Months in Period

Equals Percentage Change (rounded to 1 decimal)*

151.0

145.2

3.994% * 12 / 13

3.7%
```

- * This calculation method will be used instead of the Percentage Change calculation shown in Section 2 of this Schedule
- 9.1 if the period between the ECI Base Index and the ECI Current Index is other than 12 months.
- 4. Adjustments Using Hewitt Index. The Parties acknowledge and agree to use the percent change in "Total Cash Compensation" for Systems Integration Job Families (the "Percent Change"), as the basis for annual adjustments to the charges to be paid by Franklin Covey to EDS under this Agreement for those Services listed in Section 5 of this Schedule 9.1 as being subject to this Section 4 (the "Hewitt Index Adjustable Charges"), as the Percent Change is either reported in the Hewitt Associates Index for Total Cash Compensation (the "Index") or as such Systems Integration Job Families information is otherwise made available by the management consulting firm of Hewitt Associates LLC (or another comparable measure published or made available by a mutually agreeable source should the Index no longer be published, the content or format of the Index substantially change or Hewitt Associates LLC no longer make comparable Systems Integration Job Families information available). If, on any anniversary of the Effective Date during the term of this Agreement, the most recently published or available Percent Change is greater than 4%, an adjustment to the Hewitt Index Adjustable Charges by such Percent Change in excess of four percent (4%). If an adjustment is not made on a anniversary of the Effective Date on which no adjustment was made. The Parties acknowledge and agree that EDS will adjust the Hewitt Index Adjustable Charges and will advise Franklin Covey of such adjustment in writing so that the new charges will amend this Agreement and become effective on the applicable anniversary of the Effective Date on anniversary of the Effective Date for any reason, EDS will advise Franklin Covey in writing of such fact.
- 5. Service Categories. The Parties acknowledge and agree that the following Services will be subject to adjustment in accordance with Section 1 of this Schedule 9.1: Midrange; Communications Management; Distributed Systems Management; Web Hosting; and Leadership Support. The Parties acknowledge and agree that the following Services will be subject to adjustment in accordance with Section 4 of this Schedule 9.1: Applications and Technical Architects.

Revised Schedule 9.1 (a) Variable Monthly Charges

	2001	2002	2003	2004	2005	2006	2007	2008	2009	2010	2011
Midrange: (1) Small / medium Large	2,838.31 case-by-case	-,	3,244.02	3,308.90	3,375.08	3,442.58	3,511.43	3,581.66	3,653.29	3,726.36	3,800.88
Communications Mgmt: (1)											
Voice Definity Ports	9.35	10.62	10.83	11.05	11.27	11.49	11.72	11.96	12.20	12.44	12.69
Mailboxes	8.18	9.28	9.46	9.65	9.84	10.04	10.24	10.45	10.66	10.87	11.09
Rockwell ACD	0.66	0.75	0.77	0.78	0.80	0.81	0.83	0.85	0.86	0.88	0.90
Routers											
Small	640	726.74	741.27	756.10	771.22	786.65	802.38	818.43	834.80	851.49	868.52
Medium	880	999.78	1,019.77	1,040.17	1,060.97	1,082.19	1,103.83	1,125.91	1,148.43	1,171.40	1,194.82
Large	1850	2101.05	2,143.07	2,185.93	2,229.65	2,274.24	2,319.73	2,366.12	2,413.45	2,461.71	2,510.95

Distributed Systems Mgmt: (1)											
Per Desktop	87.71	97.12	99.06	101.04	103.06	105.13	107.23	109.37	111.56	113.79	116.07
Incremental Email Box	9.26	10.35	10.56	10.77	10.98	11.20	11.43	11.66	11.89	12.13	12.37
Information Systems Resource Ra	tes: (2),	(3)									
Long-term:(5), (7)											
Information Associate	9,400	9,776	10,167	10,574	10,997	11,437	11,894	12,370	12,865		13,914
Information Analyst	12,000	12,480	12,979	13,498	14,038	14,600	15,184	15,791	16,423	17,080	17,763
Information Specialist	14,100	14,664	15,251	15,861	16,495	17,155	17,841	18,555	19,297	20,069	20,871
Information Specialist - Senior	16,400	17,056	17,738	18,448	19,186	19,953	20,751	21,581	22,445	23,342	24,276
System Architect	17,800	18,512	19,252	20,023	20,823	21,656	22,523	23,424	24,361	25,335	26,348
DBA	17,800	18,512	19,252	20,023	20,823	21,656	22,523	23,424	24,361	25,335	26,348
Project Control Manager	17,800	18,512	19,252	20,023	20,823	21,656	22,523	23,424	24,361	25,335	26,348
Short-term: (4), (6)											
Information Associate	10,900	11,336	11,789		12,751	13,262	13,792	14,344	14,917		16,135
Information Analyst	13,900	14,456	15,034		16,261	16,911	17,588	18,291	19,023	19,784	20,575
Information Specialist	16,400	17,056	17,738	18,448	19,186	19,953	20,751	21,581	22,445	23,342	24,276
Information Specialist - Senior	19,100	19,864	20,659	21,485	22,344	23,238	24,168	25,134	26,140	27,185	28,273
System Architect	20,600	21,424	22,281	23,172	24,099	25,063	26,066	27,108	28,193	29,320	30,493
DBA	20,600	21,424	22,281	23,172	24,099	25,063	26,066	27,108	28,193	29,320	30,493
Project Control Manager	20,600	21,424	22,281	23,172	24,099	25,063	26,066	27,108	28,193	29,320	30,493
	(=)										
E-Solutions Resource Rates: (2)	, (3)										
Long-term:(5), (7)											
Consultant Associate	14,535	15,116	15,721		17,004	17,684	18,391	19,127	19,892	20,688	21,515
Consultant	18,900	19,656	20,442	21,260	22,110	22,995	23,915	24,871	25,866	26,901	27,977
Consultant Specialist	23,985	24,944	25,942	26,980	28,059	29,181	30,349	31,563	32,825	34,138	35,504
Consultant Senior	28,345	29,479	30,658	31,884	33,160	34,486	35,865	37,300	38,792		41,958
Consultant Architect	32,700	34,008	35,368	36,783	38,254	39,785	41,376	43,031	44,752	46,542	48,404
Program Client Executive	36,340	37,794	39,305	40,878	42,513	44,213	45,982	47,821	49,734	51,723	53,792
05											
Short-term: (4), (6), (8)	400	404	100	440		100	407	100	407	4.40	4.40
	100	104	108	112	117	122	127	132	137	142	148
	130	135	141	146	152	158	164	171	178	185	192
	165	172	178	186	193	201	209	217	226	235	244
	195	203	211	219	228	237	247	257	267	278	289
	225	234	243	253	263	274	285	296	308	320	333
Program Client Executive	250	260	270	281	292	304	316	329	342	356	370

- (1) Subject to ECI inflation index pursuant to Section 9.1 (d) of this Schedule
 (2) Subject to Hewitt inflation index pursuant to Section 9.1 (d) of this Schedule
 (3) Person-month is defined as 130 hours / month and the rates do not include travel-related expenses
 (4) Short-term rates are for resources less than 6 months in duration
 (5) Long-term rates are for resources greater than 6 months in duration
 (6) Additional job classifications will be discounted 10% off EDS' Commercial Billing Rates
 (7) Additional job classification will be discounted 25% off EDS' Commercial Billing Rates
 (8) Short-term E.Solutions rates are hourly

Web Hosting (1)(2)	2001	2002	2003	2004	2005	2006	2007	2008	2009	2010	2011
Sun level 2 Basic Monthly Start Up	2,049.30 3,081.60	2,322.54 3,492.48	2,368.99 3,562.33			2,513.99 3,780.37	2,564.27 3,855.98		2,667.87 4,011.76	2,721.23 4,092.00	2,775.65 4,173.84
Sun level 3 Enhanced Monthly Start Up	3,039.30 4,274.00	- /				3,728.48 5,243.28	3,803.05 5,348.15			4,035.83 5,675.49	4,116.54 5,789.00
Sun Level 4 Enhanced Monthly Start Up	3,934.80 4,333.80	4,459.44 4,911.64	4,548.63 5,009.87		0 4,732.39 7 5,212.27	4,827.04 5,316.52			5,122.49 5,641.93	5,224.94 5,754.77	5,329.44 5,869.86
Sun Level 3 Basic Monthly Start Up	2,277.90 3,676.50		2,633.25 4,250.03			2,794.43 4,510.17	2,850.32 4,600.37	2,907.32 4,692.38		3,024.78 4,881.95	-,
Sun Level 4 Basic Monthly Start Up	3,223.80 3,081.60	3,653.64 3,492.48				3,954.82 3,780.37		4,114.59 3,933.10	4,196.88 4,011.76	4,280.82 4,092.00	4,366.44 4,173.84
Custom Servers	Case-by-0	Case basis									
Additional momitoring events	25.00	25.50	26.01	26.53	27.06	27.60	28.15	28.72	29.29	29.88	30.47

- Subject to ECI inflation index pursuant to Section 9.1 (d) of this Schedule Any change in monthly plus or minus 10% triggers a renegotiation

Franklin Covey will pay a one-time charge of \$5,500 per 500-user-test instance. A test instance will be defined as 90 consecutive calendar days. After 90 days, Franklin Covey may pay \$2500 per additional day for testing services, or may pay an additional \$5,500 to establish a new 90-day test instance. Franklin Covey will pay \$85 per hour for testing labor, set up and testing activity performed by EDS.

Each day the tests are run at EDS, Franklin Covey will pay \$500 per day for EDS provided hardware and software infrastructure

Web Hosting Internet Bandwidth Charges -

5 MB bandwidth 3200 3264 3,329.28 3,395.87 3,463.78 3,533.06 3,603.72 3,675.79 3,749.31 3,824.30 3,900.78

- 1. The usage of such variable capacity above the 1Mbps will be tracked on a monthly basis
- 2. EDS will monitor the usage of the EDS Services by recording the actual usage every five minutes.

 3. EDS will review the overall usage each calendar month and will discard the top five percent (5%) statistical usage.

 4. The highest remaining value is then used as the basis for the connection usage for the calendar month and is referred to as the
- 4. The highest is... "95th percentile."

Such usage will be rounded upward to the next highest Mbps for the calculation of the usage based fee.

Revised Schedule 9.1 (b) **Baseline Monthly Charges**

	Start-up	Midrange	Network Management	Systems Management	Applications	Web Hosting	Technical Architects	Leadership Support	Aggregate Monthly Charge
April-01	975,702	114,498	125,197	139,431	195,827	84,416	35,600	83,251	1,753,922
May-01	•	114,498	125,197	139,431	195,827	84,416	35,600	83,251	778,220
June-01		114,498	111,097	139,431	195,827	84,416	35,600	83,251	764,120
July-01		114,498	111,097	139,431	195,827	84,416	35,600	83,251	764,120
August-01		114,498	111,097	139,431	195,827	84,416	35,600	83,251	764,120
September-01	826,998	137,870	120,097	169,043	246,936	84,416	35,600	83,251	1,704,211
October-01		137,870	120,097	169,043	246,936	84,416	35,600	83,251	877,213
November-01		137,870	120,097	169,043	246,936	84,416	35,600	83,251	877,213
December-01		137,870	120,097	169,043	246,936	84,416	35,600	83,251	877,213
January-02		137,870	123,334	169,043	246,936	83,327	35,600	83,251	879,361

February-02	137,870	123,334	169,043	246,936	83,327	35,600	83,251	879,361
March-02	137,870	123,334	169,043	246,936	83,327	35,600	83,251	879,361
April-02	140,627	125,801	172,424	256,813	84,994	37,024	84,916	902,599
May-02	140,627	125,801	172,424	256,813	84,994	37,024	84,916	902,599
June-02	140,627	125,801	172,424	256,813	84,994	37,024	84,916	902,599
July-02	140,627	125,801	172,424	256,813	84,994	37,024	84,916	902,599
August-02	140,627	125,801	172,424	256,813	84,994	37,024	84,916	902,599
September-02	140,627	125,801	172,424	256,813	84,994	37,024	84,916	902,599
October-02	140,627	125,801	172,424	256,813	84,994	37,024	84,916	902,599
November-02	140,627	125,801	172,424	256,813	84,994	37,024	84,916	902,599
December-02	140,627	125,801	172,424	256,813	84,994	37,024	84,916	902,599
January-03	140,627	125,161	170,658	246,764	84,014	37,024	84,915	889,163
February-03	140,627	125,161	170,658	246,764	84,014	37,024	84,916	889,164
March-03	140,627	125,161	170,658	246,764	84,014	37,024	84,916	889,164
				256,634		38,505	86,614	
April-03	143,440	127,665	174,071		85,694			912,623
May-03	143,440	127,665	174,071	256,634	85,694	38,505	86,614	912,623
June-03	143,440	127,665	174,071	256,634	85,694	38,505	86,614	912,623
July-03	143,440	127,665	174,071	256,634	85,694	38,505	86,614	912,623
August-03	143,440	127,665	174,071	256,634	85,694	38,505	86,614	912,623
September-03	143,440	127,665	174,071	256,634	85,694	38,505	86,614	912,623
October-03	143,440	127,665	174,071	256,634	85,694	38,505	86,614	912,623
November-03	143,440	127,665	174,071	256,634	85,694	38,505	86,614	912,623
December-03	143,440	127,665	174,071	256,634	85,694	38,505	86,614	912,623
January-04	143,440	127,023	172,388	246,183	84,812	38,505	86,614	898,965
February-04	143,440	127,023	172,388	246, 183	84,812	38,505	86,614	898,965
March-04	143,440	127,023	172,388	246,183	84,812	38,505	86,614	898, 965
April-04	146,309	129,563	175,836	256,030	86,508	40,045	88,347	922,638
May-04	146,309	129,563	175,836	256,030	86,508	40,045	88,347	922,638
June-04	146,309	129,563	175,836	256,030	86,508	40,045	88,347	922,638
July-04	146,309	129,563	175,836	256,030	86,508	40,045	88,347	922,638
,		129,563	175,836		86,508	40,045	88,347	922,638
August-04	146,309			256,030				
September-04	146,309	129,563	175,836	256,030	86,508	40,045	88,347	922,638
October-04	146,309	129,563	175,836	256,030	86,508	40,045	88,347	922,638
November - 04	146,309	129,563	175,836	256,030	86,508	40,045	88,347	922,638
December-04	146,309	129,563	175,836	256,030	86,508	40,045	88,347	922,638
January-05	146,309	128,928	175,803	256,030	85,714	40,045	88,347	921,176
February-05	146,309	128,928	175,803	256,030	85,714	40,045	88,347	921,176
March-05	146,309	128,928	175,803	256,030	85,714	40,045	88,347	921,176
April-05	149,235	131,507	179,319	266,272	87,428	41,647	90,113	945,521
May-05	149,235	131,507	179,319	266,272	87,428	41,647	90,113	945,521
June-05	149,235	131,507	179,319	266,272	87,428	41,647	90,113	945,521
July-05	149,235	131,507	179,319	266,272	87,428	41,647	90,113	945,521
August-05	149,235	131,507	179,319	266,272	87,428	41,647	90,113	945,521
September-05	149,235	131,507	179,319	266,272	87,428	41,647	90,113	945,521
October-05	149,235	131,507	179,319	266,272	87,428	41,647	90,113	945,521
November-05	149,235	131,507	179,319	266,272	87,428	41,647	90,113	945,521
December-05	149,235	131,507	179,319	266,272	87,428	41,647	90,113	945,521
January-06	149,235	130,879	179,319	266,272	86,713	41,647	90,113	944,177
February-06	149,235	130,879	179,319	266,272	86,713	41,647	90,113	944,177
March-06	149,235	130,879	179,319	266,272	86,713	41,647	90,113	944,177
April-06	152,220	133,496	182,905	276,922	88,447	43,313	91,916	969,219
May-06	152,220	133,496	182,905	276,922	88,447	43,313	91,916	969,219
June-06	152,220	133,496	182,905	276,922	88,447	43,313	91,916	969,219
July-06	152,220	133,496	182,905	276,922	88,447	43,313	91,916	969,219
August-06	152,220	133,496	182,905	276,922	88,447	43,313	91,916	969,219
September-06	152,220	133,496	182,905	276,922	88,447	43,313	91,916	969,219
October-06	152,220	133,496	182,905	276,922	88,447	43,313	91,916	969,219
November-06	152,220	133,496	182,905	276,922	88,447	43,313	91,916	969,219
December-06	152,220	133,496	182,905	276,922	88,447	43,313	91,916	969,219
	152,220	132,874	182,905		87,803		91,916	
January-07 February-07			182,905	276,922		43,313		967,953
	152,220	132,874		276,922	87,803	43,313	91,916	967,953
March-07	152,220	132,874	182,905	276,922	87,803	43,313	91,916	967,953
April-07	155,264	135,531	186,563	287,999	89,559	45,045	93,754	993,716
May-07	155,264	135,531	186,563	287,999	89,559	45,045	93,754	993,716
June-07	155,264	135,531	186,563	287,999	89,559	45,045	93,754	993,716
July-07	155,264	135,531	186,563	287,999	89,559	45,045	93,754	993,716
August-07	155,264	135,531	186,563	287,999	89,559	45,045	93,754	993,716
September-07	155,264	135,531	186,563	287,999	89,559	45,045	93,754	993,716
October-07	155,264	135,531	186,563	287,999	89,559	45,045	93,754	993,716
November-07	155,264	135,531	186,563	287,999	89,559	45,045	93,754	993,716
December-07	155,264	135,531	186,563	287,999	89,559	45,045	93,754	993,716
January-08	155,264	134,914	186,563	287,999	88,979	45,045	93,754	992,519
February-08	155,264	134,914	186,563	287,999	88,979	45,045	93,754	992,519
March-08	155,264	134,914	186,563	287,999	88,979	45,045	93,754	992,519
April-08	158,369	137,612	190,295	299,519	90,759	46,847	95,629	1,019,031
May-08	158,369	137,612	190,295	299,519	90,759	46,847	95,629	1,019,031
June-08	158,369	137,612	190,295	299,519	90,759	46,847	95,629	1,019,031
July-08	158,369	137,612	190,295	299,519	90,759	46,847	95,629	1,019,031
August-08	158,369	137,612	190,295	299,519	90,759	46,847	95,629	1,019,031
September-08	158,369	137,612	190,295	299,519	90,759	46,847	95,629	1,019,031
October-08	158,369	137,612	190,295	299,519	90,759	46,847	95,629	1,019,031
November-08	158,369	137,612	190,295	299,519	90,759	46,847	95,629	1,019,031
December-08	158,369	137,612	190,295	299,519	90,759	46,847	95,629	1,019,031
January-09	158,369	137,002	190,295 190,295	299,519	90,237	46,847	95,629	1,017,899
February-09	158,369	137,002		299,519	90,237	46,847	95,629	1,017,899
March-09	158,369	137,002	190,295	299,519	90,237	46,847	95,629	1,017,899
April-09	161,537	139,742	194,100	311,500	92,042	48,721	97,542	1,045,184
May-09	161,537	139,742	194,100	311,500	92,042	48,721	97,542	1,045,184
June-09	161,537	139,742	194,100	311,500	92,042	48,721	97,542	1,045,184
July-09	161,537	139,742	194,100	311,500	92,042	48,721	97,542	1,045,184
August-09	161,537	139,742	194,100	311,500	92,042	48,721	97,542	1,045,184
September-09	161,537	139,742	194,100	311,500	92,042	48,721	97,542	1,045,184
October-09	161,537	139,742	194,100	311,500	92,042	48,721	97,542	1,045,184
November-09	161,537	139,742	194,100	311,500	92,042	48,721	97,542	1,045,184
December-09	161,537	139,742	194,100	311,500	92,042	48,721	97,542	1,045,184
January-10	161,537	139,139	194,100	311,500	91,572	48,721	97,542	1,044,111
February-10	161,537	139,139	194,100	311,500	91,572	48,721	97,542	1,044,111
March-10	161,537	139,139	194,100	311,500	91,572	48,721	97,542	1,044,111
April-10	164,767	141,921	197,982	323,960	93,403	50,670	99,493	1,072,197
May-10	164,767	141,921	197,982	323,960	93,403	50,670	99,493	1,072,197
June-10	164,767	141,921	197,982	323,960	93,403	50,670	99,493	1,072,197
July-10	164,767	141,921	197,982	323,960	93,403	50,670 50,670	99,493	1,072,197
August-10	164,767	141,921	197,982	323,960	93,403	50,670	99,493	1,072,197
September-10	164,767	141,921	197,982	323,960	93,403	50,670	99,493	1,072,197
October-10	164,767	141,921	197,982	323,960	93,403	50,670	99,493	1,072,197
November-10	164,767	141,921	197,982	323,960	93,403	50,670	99,493	1,072,197
December-10	164,767	141,921	197,982	323,960	93,403	50,670	99,493	1,072,197
January-11	164,767	141,330	197,982	323,960	92,980	50,670	99,493	1,071,183
February-11	164,767	141,330	197,982	323,960	92,980	50,670	99,493	1,071,183
March-11	164,767	141,330	197,982	323,960	92,980	50,670	99,493	1,071,183

Schedule 11.4 Termination for Convenience Charges

The Parties agree that if the Agreement is terminated on other than an anniversary of the Effective Date, and the termination fee as of the anniversary of the Effective Date preceding the effective termination date is greater than the termination fee as of the anniversary of the Effective Date proceding the effective termination date is greater than the termination fee as of the anniversary of the Effective Date proceding the effective termination date, then Franklin Covey will pay to EDS a prorated share of such difference, such proration to be based upon the number of days elapsed in the contract year prior to the effective termination date.

April 1	., 2	002	-	March	31,	2003
April 1	L, 2	003	-	March	31,	2004
April 1	L, 2	004	-	March	31,	2005
April 1	L, 2	005	-	March	31,	2006
April 1	L, 2	006	-	March	31,	2007
April 1	L, 2	007	-	March	31,	2008
April 1	L, 2	800	-	March	31,	2009
April 1	L, 2	009	-	March	31,	2010
April 1	L, 2	010	-	March	31,	2011

Effective Date of Termination

Termination for Convenience Fees

The total Fees paid or payable by Franklin Covey to EDS for the twelve (12) calendar months immediately preceding the date on which EDS received Franklin Covey's notice of termination for convenience

Fees paid or payable by Franklin Covey to EDS for the ten (10) calendar months

immediately preceding the date on which EDS received Franklin Covey's notice of termination for convenience.

The total Fees paid or payable by Franklin Covey to EDS for the eight (8) calendar months immediately preceding the date on which EDS received Franklin Covey's notice of termination for convenience.

The total Fees paid or payable by Franklin Covey to EDS for the six (6) calendar months immediately preceding the date on which EDS received Franklin Covey's notice of termination for convenience.

The total Fees paid or payable by Franklin Covey to EDS for the three (3) calendar months immediately preceding the date on which EDS received Franklin Covey's notice of termination for convenience

The total Fees paid or payable by Franklin Covey to EDS for the three (3) calendar months immediately preceding the date on which EDS received Franklin Covey's notice of termination for convenience

total Fees paid or payable by Franklin Covey to EDS for the three (3) calendar months immediately preceding the date on which EDS received Franklin Covey's notice of termination for convenience

The total Fees paid or payable by Franklin Covey to EDS for the three (3) calendar months immediately preceding the date on which EDS received Franklin Covey's notice of termination for convenience.

The total Fees paid or payable by Franklin Covey to EDS for the three (3) calendar months immediately preceding the date on which EDS received Franklin Covey's notice of termination for convenience.

Section 11.4 Section (B)

ELLECTIV	e Date	от тетштия	LIUII
April 1,	2002 -	March 31,	2003
April 1,	2003 -	March 31,	2004
April 1,	2004 -	March 31,	2005
April 1,	2005 -	March 31,	2006
April 1,	2006 -	March 31,	2007
April 1,	2007 -	March 31,	2008
April 1,	2008 -	March 31,	2009
April 1,	2009 -	March 31,	2010
April 1,	2010 -	March 31,	2011

Effective Date of Termination

Termination for Convenience Fees

The total Fees paid or payable by Franklin Covey to EDS for the three (3) calendar months immediately preceding the date on which EDS received Franklin Covey's notice of termination for convenience.

The total Fees paid or payable by Franklin Covey to EDS for the three (3) calendar months immediately preceding the date on which EDS received Franklin Covey's notice termination for convenience

The total Fees paid or payable by Franklin Covey to EDS for the three (3) calendar months immediately preceding the date on which EDS received Franklin Covey's notice of termination for convenience

The total Fees paid or payable by Franklin Covey to EDS for the three (3) calendar months immediately preceding the date on which EDS received Franklin Covey's notice of termination for convenience.

The total Fees paid or payable by Franklin Covey to EDS for the three (3) calendar

months immediately preceding the date on which EDS received Franklin Covey's notice of termination for convenience. The total Fees paid or payable by Franklin Covey to EDS for the three (3) calendar months immediately preceding the date on which EDS received Franklin Covey's notice of

termination for convenience.

The total Fees paid or payable by Franklin Covey to EDS for the three (3) calendar months immediately preceding the date on which EDS received Franklin Covey's notice termination for convenience

The total Fees paid or payable by Franklin Covey to EDS for the three (3) calendar months immediately preceding the date on which EDS received Franklin Covey's notice of

months immediately proceeding the date on which EDS received Franklin Covey's notice of termination for convenience.

Section 11.4 Section (C)

Effective Date of Termination April 1, 2002 - March 31, 2003 April 1, 2003 - March 31, 2004 April 1, 2004 - March 31, 2005 April 1, 2005 - March 31, 2006 April 1, 2006 - March 31, 2007 April 1, 2007 - March 31, 2008 April 1, 2008 - March 31, 2009 April 1, 2009 - March 31, 2010 April 1, 2010 - March 31, 2011

Termination for Convenience Fees

The total Fees paid or payable by Franklin Covey to EDS for the twelve (12) calendar months immediately preceding the date on which EDS received Franklin Covey's notice of termination for convenience.

The total Fees paid or payable by Franklin Covey to EDS for the ten (10) calendar months immediately preceding the date on which EDS received Franklin Covey's notice of termination for convenience.

The total Fees paid or payable by Franklin Covey to EDS for the three (3) calendar months immediately preceding the date on which EDS received Franklin Covey's notice of termination for convenience

The total Fees paid or payable by Franklin Covey to EDS for the three (3) calendar months immediately preceding the date on which EDS received Franklin Covey's notice of

termination for convenience.
The total Fees paid or payable by Franklin Covey to EDS for the three (3) calendar months immediately preceding the date on which EDS received Franklin Covey's notice of termination for convenience

The total Fees paid or payable by Franklin Covey to EDS for the three (3) calendar months immediately preceding the date on which EDS received Franklin Covey's notice of termination for convenience.

The total Fees paid or payable by Franklin Covey to EDS for the three (3) calendar months immediately preceding the date on which EDS received Franklin Covey's notice of

termination for convenience The total Fees paid or payable by Franklin Covey to EDS for the three (3) calendar months immediately preceding the date on which EDS received Franklin Covey's notice of

termination for convenience

The total Fees paid or payable by Franklin Covey to EDS for the three (3) calendar months immediately preceding the date on which EDS received Franklin Covey's notice of termination for convenience

Additional Services Addendum #1

This Additional Services Addendum #1 ("Addendum #1") dated as of June 30, 2001 (the "Addendum #1 Effective Date"), is to that certain agreement for Information Technology Services (the "Agreement") dated April 1, 2001 (the "Agreement Effective Date") between Electronic Data Systems Corporation and EDS Information Services, L.L.C. ("EDS"), and Franklin Covey Company ("Franklin Covey"). Defined terms utilized herein, to the extent not defined herein, shall have the meaning ascribed to such terms in the Agreement.

RECITALS

WHEREAS, pursuant to the Section 3.5(a) of the Agreement, Franklin Covey desires to obtain from EDS certain Additional Services as described in this Addendum #1; and

WHEREAS, EDS desires to undertake the performance of such Additional Services as set forth in this Addendum #1;

NOW, THEREFORE, in consideration of the promises and mutual agreements of the parties herein expressed, and intending to be legally bound hereby, the parties agree as follows:

- 1. The term of this Addendum #1 will commence on the Addendum #1 Effective Date and, unless earlier terminated as provided in the Agreement or this Addendum #1, will expire on the fifteenth (15th) anniversary of the Addendum #1 Effective Date. This Addendum #1 may be extended on one or more occasions by mutual written agreement of the Parties. The provisions of the Agreement will continue to apply to this Addendum #1 notwithstanding the earlier expiration or other termination of the Agreement.
- 2. During the term of this Addendum #1, EDS will provide to Franklin Covey the Services described in the attached Exhibit A-2 (Distribution Warehouse Services Statement of Work). In connection with the Services provided by EDS under this Addendum #1, Franklin Covey will provide to EDS the support and resources described in the attached Exhibits A-1 and A-2. EDS will perform the Addendum #1
 Services in accordance with the Service Levels described in Exhibits A-1 and A-2.
- 3. For the Services provided by EDS under this Addendum #1, Franklin Covey will pay to EDS the amounts specified in the attached Exhibit B, plus any and all Collected Taxes (as defined in Section 9.4 of the Agreement) and other applicable amounts described in Article IX of the Agreement. The charges hereunder will be subject to an annual adjustment according to annual changes in the ECI in accordance with the provisions of Schedule 9.1 to the Agreement.
- The Services provided by EDS under this Addendum #1 will be subject to the additional provisions set forth in the attached <u>Exhibit C</u>.
- 5. Except as may be contrary to the express terms of this Addendum #1, performance of this Addendum #1 by both parties shall be subject to the terms and conditions of the Agreement.

IN WITNESS WHEREOF, the parties have executed this Addendum #1 by their authorized representatives as of the date first set forth above.

Electronic Data Systems Corporation		Franklin Covey Co.
name		name
title		title
date		date
EDS Information Services, L.L.C.	_	Franklin Covey Products Sales, Inc.
name	_	name
title	_	title
date		date
		Franklin Covey Client Sales, Inc.
		name
		title
		date

Exhibit B to Addendum #1

Charges

This Exhibit B contains EDS' fees, charges and rates for the Services being provided pursuant to Exhibit A-1 and Exhibit A-2 to Addendum #1.

- Charges Specific to Exhibit A-1.
 - a) This Section sets forth the charges which will be applied based on volumes of Associate Hours in the Services provided by EDS for the Customer Care Center (the "CCC Transaction Charges"). The CCC Transaction Charges will take effect beginning month 13 following the Addendum #1 Effective Date; provided that on each anniversary of the Addendum #1 Effective Date, the charges set forth in Table 1 below (expressly excluding Telecommunications and Freight fees, rates and charges) will be increased by the sum of (i) 2% plus (ii) the percentage adjustment, if any, resulting from the application of the adjustment mechanism set forth in Section 1 of Schedule 9.1 to the Agreement ("Annual Adjustment to Charges"). The following sets forth the tiered pricing schedule as it relates to CCC Transaction Charges.

Table 1: CCC Transaction Charges:

	A	В
	Month 13 thru Month 38*	Month 39 thru Month 180*
Up to 300,000 Associate Hours per contract year	\$31.65 per Associate Hour	\$31.15 per Associate Hour
Next 50,000 Associate Hours per contract year	\$31.65 per Associate Hour	\$30.15 per Associate Hour

Incremental Associate \$31.65 per Associate Hour \$29.15 per Associate Hour contract year

* EDS will bill Franklin Covey a minimum of 80% of the estimated 277,280 Associate Hours (Associate Hour is defined as all scheduled work hours for each individual that delivers the services defined in the Franklin Covey Customer Care Center SOW, Section 2.0 EDS General Responsibilities - Services Description Table) per year according to the following schedule:

July - September	October - December	January - March	April - June
58,560 Associate Hours			45,475 Associate Hours

EDS will reconcile billing quarterly based on the actual volumes experienced. Should volumes fall below the minimum Associate Hours stated above for the quarter indicated, Franklin Covey will be billed for the minimum Associate Hours stated.

b) <u>Telecommunications Usage Charges:</u> The Telecommunication Usage Section sets forth the method by which all charges for toll free usage services related to the Customer Care Center Services will be applied. EDS will provide a rate chart at the beginning of each year outlining a Telecommunications Usage rate on which EDS will base toll free usage charges to Franklin Covey. If Franklin Covey is able to obtain lower rates for a particular locality, and EDS is able to take advantage of such lower rates, EDS will provide services in that locality at or below said rates. The following rates apply for the first 6 months of the Initial term:

Telecommunication Usage Rate \$0.055 per minute

c) The CCC Transaction Charges for the last month of the Initial Term shall continue for each month during which EDS continues to perform Services under Exhibit A-2 while performing termination assistance for Franklin Covey pursuant to Section 11.5 of the Agreement.

II. Charges Specific to Exhibit A-2.

This Section sets forth the charges, which will be applied, based on volumes of boxes shipped in the Services provided by EDS pursuant to Exhibit A-2 (the "W&D Transaction Charges"). The W&D Transaction Charges will take effect beginning month 13 after the Addendum #1 Effective Date; provided that on each anniversary of the Addendum #1 Effective Date, the charges set forth in Table 1 below (expressly excluding Telecommunication and Freight fees, rates and charges) will be increased by the sum of (i) 2% plus (ii) the percentage adjustment, if any, resulting from the application of the adjustment mechanism set forth in Section 1 of Schedule 9.1 to the Agreement ("Annual Adjustment to Charges"). The following sets forth the tiered pricing schedule as it relates to W&D Transaction Charges:

Table 2: W&D Transaction Charges:

				Α		В		С
			Month 13	3 thru	Month 38*	Month 39 thru	Month 50**	Month 51 - Month 180***
First 2.0M	Boxes Shipped	\$3.53 per l	box shippe	ed	\$3.49 per	box shipped	\$3.49 per bo	ox shipped
	Next 0.3M Boxes	Shipped	\$3.43 pe	er box	shipped	\$3.39 per box	shipped	\$3.39 per box shipped
	Incremental Boxe	s Shipped	\$3.33 pe	er box	shipped	\$3.29 per box	shipped	\$3.29 per box shipped

*During months 13 through 38, EDS will bill Franklin Covey a minimum of 2.2 M boxes shipped annually, calculated and paid quarterly based on the quarterly volume table below.

** During months 39 through 50, EDS will bill Franklin Covey a minimum of 2.2M boxes shipped annually, calculated and paid quarterly based on the quarterly volume table below.

*** During months 51 through 180, EDS will bill Franklin Covey a minimum of 1.75M boxes shipped annually, calculated and paid quarterly based on the quarterly volume table below.

EDS has based the 2.2M boxes shipped minimum on the following schedule:

July - September October- December January - March April - June 520,000 boxes shipped 761,000 boxes shipped 495,000 boxes shipped 424,000 boxes shipped

EDS has based the 1.75M boxes shipped minimum on the following schedule:

 July - September
 October - December
 January - March
 April - June

 425,250 boxes shipped
 623,000 boxes shipped
 404,250 boxes shipped
 297,500 boxes shipped

EDS will reconcile billing quarterly based on the above volume estimates. Should volumes fall below the stated minimum boxes for a particular quarter, Franklin Covey will be billed for the minimum boxes stated for

- b) The Freight Fee Section sets forth the method by which all charges related to freight of boxes shipped will be billed to Franklin Covey. EDS will initially use the EDS applied rates as they relate to next day and second day shipping. These rates are updated on a 3-month cycle and will be forwarded to Franklin Covey, as they are made available to EDS. New rates will become effective on the first day of the month following the date on which Franklin Covey is notified of such new rates. Attachment B-2 and Attachment B-2 identify the current rates related to next day and second day air shipments. For all other shipping modes (i.e. ground, residential, commercial), EDS will use the Franklin Covey negotiated rates. Should EDS obtain more favorable rates than the current Franklin Covey negotiated rates. EDS will apply the more favorable rate to the freight fee applied to the monthly invoice. Should Franklin Covey obtain more favorable rates than the EDS rate schedule, EDS will apply Franklin Covey's contracted rates.
- c) The Transaction Charges for the last month of the Initial Term shall continue for each month during which EDS continues to perform Services under Exhibit A-2 while performing termination assistance for Franklin Covey pursuant to Section 11.5 of the Agreement
- II. Charges, which apply to both Customer Care Center and Warehouse & Distribution Services:
 - a) One Time License Fee Section. The One Time License Fee Section sets forth the payment schedule of the \$4,500,000.00 license fee for the use of EDS' integrated Order Management and Warehouse Management System, including the Retek and McHugh systems, as implemented. Franklin Covey will pay EDS the One Time License Fee as follows:

Month 3 following the Addendum #1 Effective \$1,500,000

Date \$1,500,000

Month 6 following the Addendum #1 Effective \$1,500,000

Date \$1,500,000

Upon acceptance and implementation of the 0MS/WMS, scheduled for March 23, 2002

Addendum #1 to the Agreement. Listed below is the payment schedule for said twelve-month period:

Month	Minimum Payment
July 2001	\$1,416,057
August 2001	\$1,490,210
September 2001	\$1,541,241
October 2001	\$1,793,741
November 2001	\$1,454,067
December 2001	\$1,655,367
January 2002	\$1,470,967
February 2002	\$1,226,667
March 2002	\$1,281,467
April 2002	\$1,126,167
May 2002	\$1,153,967
June 2002	\$1,396,167

These payments exclude telecommunications charges and freight charges. Telecommunications charges and freight charges will be applied based on the rates identified in Section I Exhibit A-1 and Section I Exhibit A-2.

The above charges are based on an estimated volume of 277,280 Associate Hours and 2.0M boxes shipped.

EDS will reconcile billing after the first 12 months. Should volumes exceed 110% of volumes identified above, EDS will bill the additional volumes based on the transaction charges outlined in Exhibit B, Section I, Table 1, Column A and Exhibit B, Section II, Table 2, Column A.

<u>Exhibit C</u> to Addendum #1

Additional Provisions

- 1. <u>Procedures Manual</u>. EDS shall complete and deliver to Franklin Covey an additional Procedures Manual covering the activities to be provided hereunder as described in Exhibit A1 and A2 and this paragraph. Within 60 days of the Addendum #1 Effective Date, EDS will produce an Interim Operations Plan, which plan will be updated within 4 months after completion of implementation of the Order Management System / Warehouse Management System ("OMS/WMS").
- Transition of Employees. EDS will extend offers of at-will employment to each Franklin Covey employee listed in Attachment 1 to this Exhibit C (each, an "Addendum #1 Employee Offeree") in accordance with EDS' normal employment policies, subject to the allowances and provisions set forth in Article II of the Agreement, at a base salary at least equal to the base salary such employee received from Franklin Covey immediately prior to the Addendum #1 Effective Date. Each Addendum #1 Employee Offeree who accepts EDS' offer of employment pursuant to EDS' normal employment policies, this provision, and Article II of the Agreement will become an EDS employee (each, an "Addendum #1 Transitioned Employee") as of July 1, 2001 (the "Addendum #1 Employment Date") and will be eligible to participate in all employee benefit plans or employment policies and programs available to similarly situated EDS employees. For purposes of this Addendum #1, (i) the reference to the Addendum #1 Employment Date shall be the date set forth in this paragraph 2, and not the date set forth in Section 2.1 (e) of the Agreement; (ii) the reference to Schedule 11.4 in Section 2.1(e) of the Agreement shall be deemed to be to the transition plans set forth in Exhibits A-1 and A-2 to this Addendum #1.
- 3. <u>Use of Subsidiaries and Subcontractors.</u> Due to the cyclical nature of the Addendum #1 Services, the second sentence of Section 2.4 of the Agreement shall not apply to subcontracts for non-managerial resources used to perform the Addendum #1 Services.
- 4. Shared Cost Savings. Franklin Covey agrees that, with respect to changes proposed by EDS which were originated outside of the Franklin Covey account, EDS will have the right to bear the entire cost of the applicable change or new service, and thereby be entitled to all of the savings associated therewith.
- 5. <u>Cost and Quality Benchmarking</u>. The parties agree that the circumstances described in <u>Section 9.6</u> of the Agreement for determining comparability of offerings will include, without limitation, the geographic location where such Services are being performed, and the specific service levels at which the comparable services are being performed.
- 6. <u>Termination for Convenience</u>. Franklin Covey shall not have the right to terminate this Addendum #1 for convenience without EDS' written consent, which consent may be withheld in EDS' reasonable discretion.
- 7. <u>License to OMS/WMS Software</u>. In connection with the completion of implementation of EDS' OMS/WMS software, EDS will grant to Franklin Covey a non-exclusive, non-transferable, non-refundable license to use and access the OMS/WMS software to the extent reasonably necessary to receive Services under this Addendum #1 during the term of this Addendum #1. For such license, Franklin Covey shall pay EDS the license fee as set forth in Exhibit B to Addendum #1. Franklin Covey shall have no continuing rights to the OMS/WMS software following the termination or expiration of this Agreement or any extension hereof and the completion of any Termination Transition services being performed pursuant to Section 11.5 of the Agreement.
- Termination for Significant Business Change. Subject to the other provisions of this Agreement, and in the event that Franklin Covey encounters a significant change in its business such that Franklin Covey no longer requires the delivery of Services under Exhibit A-1, Call Center Services SOW (a "Significant Call Center Business Change"), Franklin Covey may terminate the Call Center Services at any time upon at least 6 months prior written notice to EDS, provided however, that if Franklin Covey terminates this Agreement, Franklin Covey shall pay EDS, as EDS' sole remedy for termination, upon providing such notice, the applicable termination fees set forth in Table 1 below. In the event that Franklin Covey encounters a significant change in its business such that Franklin Covey no longer requires the delivery of Services under Exhibit A-2, Distribution Warehouse SOW (a "Significant Warehouse Business Change"), EDS and Franklin Covey agree to negotiate in good faith an alternate services agreement, for alternate services sufficient to replace the minimum spend levels of this Addendum #1. In the event that EDS and Franklin Covey are unable to negotiate an alternate services agreement as described above, EDS and Franklin Covey agree to negotiate a termination fee that covers EDS unamortized costs and shutdown expenses related to this Addendum #1 and takes into account, among other factors, EDS' lost profits, and remaining lease and sublease expenses for the Salt Lake City distribution facility.

Table 1: Call Center Termination Fees

Calendar Year of Termination	Termination Fees
2002	150% of Minimum Annual Fee for CCC Services
2003	133% of Minimum Annual Fee for CCC Services
2004	117% of Minimum Annual Fee for CCC Services
2005	100% of Minimum Annual Fee for CCC Services

2006	75% of Minimum Annual Fee for CCC Services
2007	75% of Minimum Annual Fee for CCC Services
2008	75% of Minimum Annual Fee for CCC Services
2009	50% of Minimum Annual Fee for CCC Services
2010	50% of Minimum Annual Fee for CCC Services
2011	33% of Minimum Annual Fee for CCC Services
2012	25% of Minimum Annual Fee for CCC Services
2013	20% of Minimum Annual Fee for CCC Services
2014	15% of Minimum Annual Fee for CCC Services
2015	8.5% of Minimum Annual Fee for CCC Services
2016	5% of Minimum Annual Fee for CCC Services

- 9. Intent to Award Additional Scope to EDS. Franklin Covey hereby represents that to the extent EDS provides Franklin Covey, within 18 months of the Addendum #1 Effective Date, a compelling business case such that Franklin Covey would receive substantial cost savings and other significant benefits through consolidation and outsourcing to EDS of all of its out-of-scope call centers and fulfillment distribution (Chandler/Government, Bellingham/Education, O.S.G., Public Programs, and Technical Support), it is Franklin Covey's intent to award the said out of scope call center support to EDS.
- 10. Oracle 11i Upgrade. EDS agrees to provide Franklin Covey by August 3, 2001 ("the Deadline"), a proposal to upgrade Franklin Covey's current financial and human resources systems to Oracle 11i (including, but not limited to, the Accounts Receivable module and considering the time frames necessary to implement the WMS/OMS system) (the "Oracle Proposal"). If EDS fails to provide the Oracle Proposal by the Deadline, and implementation of the WMS/OMS system is delayed thereby, and not as a result of any act or omission of Franklin Covey, then, subject to Section 12.2 of the Agreement, EDS shall be liable for damages related to any delay in implementation of the WMS/OMS system.

Further, in creating the Oracle Proposal, EDS will perform a "Needs Assessment" related to the upgrade between the Franklin Covey legacy systems and Oracle 11i. If Franklin Covey does not accept the Oracle Proposal, then EDS shall bill Franklin Covey for the Needs Assessment in an amount not to exceed \$60,000.00.

- <u>Liability General Limitation.</u> Section 12.2(a) of the Agreement shall not apply to this Addendum #1. Section 11.a describes the general limitation of liability with respect to the Services being provided pursuant to the Call Center Services Statement of Work, and Section 11.b below describes the general limitation of liability with respect to the Services being provided pursuant to the Distribution Warehouse Services Statement of Work.
 - a. <u>Call Center Services Statement of Work.</u> The liability of each Party to the other for all damages arising out of or related to the services provided by EDS under this Addendum #1 <u>Exhibit A-1 (Call Center Services Statement of Work)</u>, regardless of the form of action that imposes liability, whether in contract, equity, negligence, intended conduct, tort or otherwise, will be limited to and will not exceed, in the aggregate for all claims, actions and causes of action of every kind and nature, the sum of \$10,000,000.00.
 - b. <u>Distribution Warehouse Services Statement of Work.</u> The liability of each Party to the other for all damages arising out of or related to the services provided by EDS under this Addendum #1 Exhibit A-2 (Distribution Warehouse Statement of Work), regardless of the form of action that imposes liability, whether in contract, equity, negligence, intended conduct, tort or otherwise, will be limited to and will not exceed, in the aggregate for all claims, actions and causes of action of every kind and nature, the sum of \$10,000,000.00.

AMENDMENT #2

This Amendment #2 ("Amendment #2") dated as of June 30, 2001 (the "Amendment #2 Effective Date"), is to that certain agreement for Information Technology Services dated April 1, 2001 (the "Agreement"), as amended by that certain Amendment and Re-Incorporation Agreement effective as of May 14, 2001 (collectively, the "Amended Agreement") between Electronic Data Systems Corporation, EDS Information Services, L.L.C., and Franklin Covey Company. Defined terms utilized herein, to the extent not defined herein, shall have the meaning ascribed to such terms in the Amended Agreement.

RECITALS

WHEREAS, the parties desire to extend the Term of the Amended Agreement and otherwise modify the Amended Agreement as described Amendment #2;

NOW, THEREFORE, in consideration of the promises and mutual agreements of the parties herein expressed, and intending to be legally bound hereby, the parties agree to amend the Amended Agreement as follows:

- 1. The Term of the Amended Agreement shall expire on June 30, 2016, and Section 3.1 of the Amended Agreement is amended by changing the reference to March 31, 2011 to June 30, 2016.
- 2. Schedule 9.1(a) to the Amended Agreement, Resource Baselines, is hereby amended by adding resource baseline information pertaining to the five years and three months added to the Term through this Amendment #2, which information is set forth in Section 1 of Attachment 1 to this Amendment #2.
- 3. Schedule 9.1(b) to the Amended Agreement, Baseline Monthly Charges, is hereby amended by adding baseline monthly charges pertaining to the five years and three months added to the Term through this Amendment #2, which baseline monthly charges are set forth in Section 2 of Attachment 1 to this Amendment #2.
- 4. Schedule 9.1(a) to the Amended Agreement, Variable Monthly Charges, was unintentionally designated Schedule 9.1(a), instead of Schedule 9.1(c). This schedule is hereby amended by redesignating it as Schedule 9.1(c), and by adding variable monthly charges pertaining to the five years and three months added to the Term through this Amendment #2, which variable monthly charges are set forth in Section 3 of Attachment 1 to this Amendment #2.
- 5. Schedule 11.4 to the Amended Agreement is hereby amended by adding to each of Sections A, B and C of Schedule 11.4 termination for convenience charges pertaining to the five years and three months added to the Term through this Amendment #2, which termination for convenience charges are set forth in Attachment 2 to this Amendment #2.
- 6. Except as may be contrary to the express terms of this Amendment #2, all other terms of the Amended Agreement shall remain in effect.

IN WITNESS WHEREOF, the parties have executed this Amendment #2 by their authorized representatives as of the date first set forth above.

Electronic Data Systems Corporation		Franklin Covey Co.
signature	-	signature
name	-	name
title	-	title
date	-	date
EDS Information Services, L.L.C.		
signature	-	
name	-	
title	=	
	_	

Attachment 1, Section 1 to Amendment 2

Resource Baselines for Years 11 - 15 of Agreement

Schedule 9.1 (a) Resource Baselines

	2001	2002	2003	2004	2005	2006	2007	2008	2009	2010	2011
IS Application Resources											
Maintenance:											
Information Associate	7	5	4	3	3	3	3	3	3	3	3
Information Analyst	3	2	2	2	2	2	2	2	2	2	2
Information Specialist	2	3	3	3	3	3	3	3	3	3	3
Development:											
Information Associate	4	4	4	4	4	4	4	4	4	4	4
Information Analyst	4	4	4	4	4	4	4	4	4	4	4
Information Specialist	0	0	0	0	0	0	0	0	0	0	0
Total:	20	18	17	16	16	16	16	16	16	16	16
ES Application Resources											
Information Associate	0	0	0	0	0	0	0	0	0	0	0
Information Analyst	0	0	0	0	0	0	0	0	0	0	0
Information Specialist	3	3	3	3	3	3	3	3	3	3	3 3
Total:	3	3	3	3	3	3	3	3	3	3	3
Midrange:											
Small / medium	44	44	44	44	44	44	44	44	44	44	44
Large	0	Θ	0	0	Θ	0	0	0	0	0	0
Communications Mgmt: Voice											
Definity Ports	4386	4386	4386	4386	4386	4386	4386	4386	4386	4386	4386
Mailboxes	2600	2600	2600	2600	2600	2600	2600	2600	2600	2600	2600
Rockwell ACD	500	500	500	500	500	500	500	500	500	500	500
MOCKWETT ACD	300	300	300	300	300	300	300	300	300	300	300
Routers											
Small	11	11	11	11	11	11	11	11	11	11	11

Medium	6	6	6	6	6	6	6	6	6	6	6
Large	5	5	5	5	5	5	5	5	5	5	5
Distributed Systems Mgmt: Desktops / Laptops File/Print/Email Incremental e-mail boxes (over the desktop baseline)	1650 47 850										

Attachment 1, Section 2 to Amendment 2

6 5

Baseline Monthly Charges for Years 11 - 15 of Agreement

Schedule 9.1 (b) Baseline Monthly Charges

		Ва	Seline Monthly Cr	iar ges					
				Distributed					Aggregate
			Communications				Technical	Leadership	Monthly
	Start-up	Midrange	Management	Management	Applications	Web Hosting	Architects	Support	Charge
April-01	975,702	114,498	125,197	139,431	195,827	89,417	35,600		1,758,923
May-01		114,498	125,197	139,431	195,827	89,417	35,600	83,251	783,221
June-01 July-01		114,498 114,498	111,097 111,097	139,431 139,431	195,827 195,827	89,417 89,417	35,600 35,600	83,251 83,251	769,121 769,121
August-01		114,498	111,007	139,431	195,827	89,417	35,600	83,251	769,121
September-01	826,998	137,870	120,097	169,043	246,936	89,417	35,600	83,251	1,709,212
October-01		137,870	120,097	169,043	246,936	89,417	35,600	83,251	882,214
November-01		137,870	120,097	169,043	246,936	89,417	35,600	83,251	882,214
December-01 January-02		137,870 137,870	120,097 123,334	169,043 169,043	246,936 246,936	89,417 87,239	35,600 35,600	83,251 83,251	882,214 883,273
February-02		137,870	123,334	169,043	246,936	87,239	35,600	83,251	883,273
March-02		137,870	123,334	169,043	246,936	87,239	35,600	83,251	883,273
April-02		140,627	125,801	172,424	256,813	88,984	37,024	84,916	906,589
May-02		140,627	125,801	172,424	256,813	88,984	37,024	84,916	906,589
June-02 July-02		140,627 140,627	125,801 125,801	172,424 172,424	256,813 256,813	88,984 88,984	37,024 37,024	84,916 84,916	906,589 906,589
August-02		140,627	125,801	172,424	256,813	88,984	37,024	84,916	906,589
September-02		140,627	125,801	172,424	256,813	88,984	37,024	84,916	906,589
October-02		140,627	125,801	172,424	256,813	88,984	37,024	84,916	906,589
November-02		140,627	125,801	172,424	256,813	88,984	37,024	84,916	906,589
December-02		140,627	125,801	172,424	256,813	88,984 87,033	37,024 37,024	84,916	906,589
January-03 February-03		140,627 140,627	125,161 125,161	170,658 170,658	246,764 246,764	87,033	37,024	84,915 84,916	892,182 892,183
March-03		140,627	125, 161	170,658	246,764	87,033	37,024	84,916	892,183
April-03		143,440	127,665	174,071	256,634	88,773	38,505	86,614	915,702
May-03		143,440	127,665	174,071	256,634	88,773	38,505	86,614	915,702
June-03		143,440	127,665	174,071	256,634	88,773	38,505	86,614	915,702
July-03		143,440 143,440	127,665 127,665	174,071 174,071	256,634 256,634	88,773 88,773	38,505 38,505	86,614 86,614	915,702 915,702
August-03 September-03		143,440	127,665	174,071	256,634	88,773	38,505	86,614	915,702
October-03		143,440	127,665	174,071	256,634	88,773	38,505	86,614	915,702
November-03		143,440	127,665	174,071	256,634	88,773	38,505	86,614	915,702
December-03		143,440	127,665	174,071	256,634	88,773	38,505	86,614	915,702
January-04		143,440	127,023	172,388	246,183	87,206	38,505	86,614	901,359
February-04 March-04		143,440 143,440	127,023 127,023	172,388 172,388	246,183 246,183	87,206 87,206	38,505 38,505	86,614 86,614	901,359 901,359
April-04		146,309	129,563	175,836	256,030	88,950	40,045	88,347	925,080
May-04		146,309	129,563	175,836	256,030	88,950	40,045	88,347	925,080
June-04		146,309	129,563	175,836	256,030	88,950	40,045	88,347	925,080
July-04		146,309	129,563	175,836	256,030	88,950	40,045	88,347	925,080
August-04		146,309	129,563	175,836	256,030	88,950	40,045	88,347	925,080
September-04 October-04		146,309 146,309	129,563 129,563	175,836 175,836	256,030 256,030	88,950 88,950	40,045 40,045	88,347 88,347	925,080 925,080
November-04		146,309	129,563	175,836	256,030	88,950	40,045	88,347	925,080
December-04		146,309	129,563	175,836	256,030	88,950	40,045	88,347	925,080
January-05		146,309	128,928	175,803	256,030	87,513	40,045	88,347	922,974
February-05		146,309	128,928	175,803	256,030	87,513	40,045	88,347	922,974
March-05 April-05		146,309 149,235	128,928 131,507	175,803 179,319	256,030 266,272	87,513 89,263	40,045 41,647	88,347 90,113	922,974 947,355
May-05		149,235	131,507	179,319	266,272	89,263	41,647	90,113	947,355
June-05		149,235	131,507	179,319	266,272	89,263	41,647	90,113	947,355
July-05		149,235	131,507	179,319	266,272	89,263	41,647	90,113	947,355
August-05		149,235	131,507	179,319	266,272	89,263	41,647	90,113	947,355
September-05 October-05		149,235 149,235	131,507 131,507	179,319 179,319	266,272 266,272	89,263 89,263	41,647 41,647	90,113 90,113	947,355 947,355
November-05		149,235	131,507	179,319	266,272	89,263	41,647	90,113	947,355
December-05		149,235	131,507	179,319	266,272	89,263	41,647	90,113	947,355
January-06		149,235	130,879	179,319	266,272	88,947	41,647	90,113	946,411
February-06		149,235	130,879	179,319	266,272	88,947	41,647	90,113	946,411
March-06 April-06		149,235 152,220	130,879 133,496	179,319 182,905	266,272 276,922	88,947 88,947	41,647 43,313	90,113 91,916	946,411 969,719
May-06		152,220	133,496	182,905	276,922	88,947	43,313	91,916	969,719
June-06		152,220	133,496	182,905	276,922	88,947	43,313	91,916	969,719
July-06		152,220	133,496	182,905	276,922	88,947	43,313	91,916	969,719
August-06		152,220	133,496	182,905	276,922	88,947	43,313	91,916	969,719
September-06 October-06		152,220	133,496	182,905	276,922	88,947	43,313	91,916	969,719
November-06		152,220 152,220	133,496 133,496	182,905 182,905	276,922 276,922	88,947 88,947	43,313 43,313	91,916 91,916	969,719 969,719
December-06		152,220	133,496	182,905	276,922	88,947	43,313	91,916	969,719
January-07		152,220	132,874	182,905	276,922	88,947	43,313	91,916	969,097
February-07		152,220	132,874	182,905	276,922	88,947	43,313	91,916	969,097
March-07		152,220	132,874	182,905	276,922	88,947	43,313	91,916	969,097
April-07 May-07		155,264 155,264	135,531 135,531	186,563 186,563	287,999 287,999	89,317 89,317	45,045 45,045	93,754 93,754	993,475 993,475
June-07		155, 264	135,531	186,563	287,999	89,317	45,045	93,754	993,475
July-07		155,264	135,531	186,563	287,999	89,317	45,045	93,754	993,475
August-07		155,264	135,531	186,563	287,999	89,317	45,045	93,754	993,475
September-07		155,264	135,531	186,563	287,999	89,317	45,045	93,754	993,475
October-07		155,264	135,531	186,563	287,999	89,317	45,045	93,754	993,475
November-07 December-07		155,264 155,264	135,531 135,531	186,563 186,563	287,999 287,999	89,317 89,317	45,045 45,045	93,754 93,754	993,475 993,475
January-08		155,264	134,914	186,563	287,999	89,173	45,045	93,754	992,713
February-08		155,264	134,914	186,563	287,999	89,173	45,045	93,754	992,713
March-08		155,264	134,914	186,563	287,999	89,173	45,045	93,754	992,713
April-08		158,369	137,612	190,295	299,519	90,956	46,847	95,629	1,019,228
May-08 June-08		158,369 158,369	137,612 137,612	190,295 190,295	299,519 299,519	90,956 90,956	46,847 46,847	95,629 95,629	1,019,228 1,019,228
July-08		158,369	137,612	190,295	299,519	90,956	46,847	95,629	1,019,228
-		•	•	•	•	•	•	•	

August-08	158,369	137,612	190,295	299,519	90,956	46,847	95,629	1,019,228
September-08	158,369	137,612	190,295	299,519	90,956	46,847	95,629	1,019,228
October-08	158,369	137,612	190,295	299,519	90,956	46,847	95,629	1,019,228
November - 08	158,369	137,612	190,295	299,519	90,956	46,847	95,629	1,019,228
December-08	158,369	137,612	190,295	299,519	90,956	46,847	95,629	1,019,228
January-09	158,369	137,002	190,295	299,519	90,812	46,847	95,629	1,018,474
February-09	158,369	137,002	190,295	299,519	90,812	46,847	95,629	1,018,474
March-09	158,369	137,002	190,295	299,519	90,812	46,847	95,629	1,018,474
April-09	161,537	139,742	194,100	311,500	92,628	48,721	97,542	1,045,771
	161,537	139,742			92,628	48,721	97,542	
May-09			194,100	311,500				1,045,771
June-09	161,537	139,742	194,100	311,500	92,628	48,721	97,542	1,045,771
July-09	161,537	139,742	194,100	311,500	92,628	48,721	97,542	1,045,771
August-09	161,537	139,742	194,100	311,500	92,628	48,721	97,542	1,045,771
September-09	161,537	139,742	194,100	311,500	92,628	48,721	97,542	1,045,771
October-09	161,537	139,742	194,100	311,500	92,628	48,721	97,542	1,045,771
November-09	161,537	139,742		311,500	92,628	48,721	97,542	1,045,771
			194,100					
December-09	161,537	139,742	194,100	311,500	92,628	48,721	97,542	1,045,771
January-10	161,537	139,139	194,100	311,500	92,483	48,721	97,542	1,045,022
February-10	161,537	139,139	194,100	311,500	92,483	48,721	97,542	1,045,022
March-10	161,537	139,139	194,100	311,500	92,483	48,721	97,542	1,045,022
April-10	164,767	141,921	197,982	323,960	94,333	50,670	99,493	1,073,126
May-10	164,767	141,921	197,982	323,960	94,333	50,670	99,493	1,073,126
June-10	164,767	141,921	197,982	323,960	94,333	50,670	99,493	1,073,126
July-10	164,767	141,921	197,982	323,960	94,333	50,670	99,493	1,073,126
August-10	164,767	141,921	197,982	323,960	94,333	50,670	99,493	1,073,126
September-10	164,767	141,921	197,982	323,960	94,333	50,670	99,493	1,073,126
October-10	164,767	141,921	197,982	323,960	94,333	50,670	99,493	1,073,126
November-10	164,767	141,921	197,982	323,960	94,333	50,670	99,493	1,073,126
June-10 July-10 August-10 September-10 October-10 November-10 December-10								
	164,767	141,921	197,982	323,960	94,333	50,670	99,493	1,073,126
January-11	164,767	141,330	197,982	323,960	92,904	50,670	99,493	1,071,107
February-11	164,767	141,330	197,982	323,960	92,904	50,670	99,493	1,071,107
March-11	164,767	141,330	197,982	323,960	92,904	50,670	99,493	1,071,107
April-11	168,063	144,157	201,942	336,918	94,762	52,697	101,482	1,100,021
	168,063					52,697		1,100,021
May-11		144,157	201,942	336,918	94,762		101,482	
June-11	168,063	144,157	201,942	336,918	94,762	52,697	101,482	1,100,021
July-11	168,063	144,157	201,942	336,918	94,762	52,697	101,482	1,100,021
August-11	168,063	144,157	201,942	336,918	94,762	52,697	101,482	1,100,021
September-11	168,063	144,157	201,942	336,918	94,762	52,697	101,482	1,100,021
October-11	168,063	144,157	201,942	336,918	94,762	52,697	101,482	1,100,021
			201,942				101,482	
November-11	168,063	144,157		336,918	94,762	52,697		1,100,021
December-11	168,063	144,157	201,942	336,918	94,762	52,697	101,482	1,100,021
January-12	168,063	143,576	201,942	336,918	93,369	52,697	101,482	1,098,047
February-12	168,063	143,576	201,942	336,918	93,369	52,697	101,482	1,098,047
March-12	168,063	143,576	201,942	336,918	93,369	52,697	101,482	1,098,047
April-12	171,424	146,447	205,981	350,395	95,236	54,805	103,512	1,127,800
May-12	171,424	146,447	205,981	350,395	95,236	54,805	103,512	1,127,800
June-12	171,424	146,447	205,981	350,395	95,236	54,805	103,512	1,127,800
July-12	171,424	146,447	205,981	350,395	95,236	54,805	103,512	1,127,800
August-12	171,424	146,447	205,981	350,395	95,236	54,805	103,512	1,127,800
September-12	171,424	146,447	205,981	350,395	95,236	54,805	103,512	1,127,800
October-12	171,424	146,447	205,981	350,395	95,236	54,805	103,512	1,127,800
November - 12	171,424	146,447	205,981	350,395	95,236	54,805	103,512	1,127,800
December-12	171,424	146,447	205,981	350,395	95,236	54,805	103,512	1,127,800
January-13	171,424	145,863	205,981	350,395	93,836	54,805	103,512	1,125,816
February-13	171,424	145,863	205,981	350,395	93,836	54,805	103,512	1,125,816
March-13	171,424	145,863	205,981	350,395	93,836	54,805	103,512	1,125,816
April-13	174,852	148,780	210,101	364,411	95,713	56,997	105,582	1,156,436
	174,852	148,780	210,101	364,411	95,713	56,997	105,582	1,156,436
May-13								
June-13	174,852	148,780	210,101	364,411	95,713	56,997	105,582	1,156,436
July-13	174,852	148,780	210,101	364,411	95,713	56,997	105,582	1,156,436
August-13	174,852	148,780	210,101	364,411	95,713	56,997	105,582	1,156,436
September-13	174,852	148,780	210,101	364,411	95,713	56,997	105,582	1,156,436
October-13	174,852	148,780	210,101	364,411	95,713	56,997	105,582	1,156,436
November-13	174,852	148,780	210,101	364,411	95,713	56,997	105,582	1,156,436
December-13	174,852	148,780	210,101	364,411	95,713	56,997	105,582	1,156,436
January-14	174,852	148,193	210,101	364,411	94,305	56,997	105,582	1,154,441
February-14	174,852	148,193	210,101	364,411	94,305	56,997	105,582	1,154,441
March-14	174,852	148,193	210,101	364,411	94,305	56,997	105,582	1,154,441
April-14	178,350	151,157	214,303	378,987	96,191	59,277	107,694	1,185,958
May-14	178,350	151,157	214,303	378,987	96,191	59, 277	107,694	1,185,958
June-14	178,350	151,157	214,303	378,987	96,191	59, 277	107,694	1,185,958
July-14	178,350	151, 157	214,303	378,987	96,191	59,277	107,694	1,185,958
August-14	178,350	151,157	214,303	378,987	96,191	59,277	107,694	1,185,958
September-14	178,350	151,157	214,303	378,987	96,191	59,277	107,694	1,185,958
October-14	178,350	151,157	214,303	378,987	96,191	59,277	107,694	1,185,958
November-14	178,350	151,157	214,303	378,987	96,191	59, 277	107,694	1,185,958
December - 14	178,350	151, 157	214,303	378,987	96,191	59,277	107,694	1,185,958
January-15	178,350	150,566	214,303	378,987	94,777	59,277	107,694	1,183,954
February-15	178,350	150,566	214,303	378,987	94,777	59,277	107,694	1,183,954
March-15	178,350	150,566	214,303	378,987	94,777	59,277	107,694	1,183,954
April-15	181,917	153,578	218,589	394,147	96,673	61,648	109,848	1,216,398
May-15	181,917	153,578	218,589	394,147	96,673	61,648	109,848	1,216,398
June-15	181,917	153,578	218,589	394,147	96,673	61,648	109,848	1,216,398
July-15		153,576	218,589		96,673	61,648	109,848	
	181,917			394,147				1,216,398
August-15	181,917	153,578	218,589	394,147	96,673	61,648	109,848	1,216,398
September-15	181,917	153,578	218,589	394,147	96,673	61,648	109,848	1,216,398
October-15	181,917	153,578	218,589	394,147	96,673	61,648	109,848	1,216,398
November-15	181,917	153,578	218,589	394,147	96,673	61,648	109,848	1,216,398
December - 15	181,917	153,578	218,589	394,147	96,673	61,648	109,848	1,216,398
January-16	181,917	152,983	218,589	394,147	95,251	61,648	109,848	1,214,382
February-16	181,917	152,983	218,589	394,147	95,251	61,648	109,848	1,214,382
March-16	181,917	152,983	218,589	394,147	95,251	61,648	109,848	1,214,382
April-16	185,555	156,043	222,960	409,913	97,156	64,114	112,045	1,247,785
May-16	185,555	156,043	222,960	409,913	97,156	64,114	112,045	1,247,785
June-16	185,555	156,043	222,960	409,913	97,156	64,114	112,045	1,247,785
	-,	,0	,	,	,	,	_,	, ,

Attachment 1, Section 3 to Amendment 2

Variable Monthly Charges for Years 11 - 15 of Agreement

Schedule 9.1 (c) Variable Monthly Charges

2001 2002 2003 2004 2005 2006 2007 2008 2009 2010 2011 2012 2013 2014

Large	case-by-cas		0,202	0,000.00	0,0.0.00	0,	0,011.10	0,002.00	0,000.20	0,.20.00	0,000.00	0,0.0.00	0,00	., 000.00	.,.
Communications Mgmt:	(1)														
Voice	(1)														
Definity Ports	9.35	10.62	10.83	11.05	11.27	11.49	11.72	11.96	12.20	12.44	12.69	12.94	13.20	13.46	
Mailboxes	8.18	9.28	9.46	9.65	9.84	10.04	10.24	10.45	10.66	10.87	11.09	11.31	11.53	11.77	
Rockwell ACD	0.66	0.75	0.77	0.78	0.80	0.81	0.83	0.85	0.86	0.88		0.92	0.93	0.95	
Bautana															
Routers Small	640	726.74	741.27	756.10	771.22	786.65	802.38	818.43	834.80	851.49	868.52	885.89	903.61	921.68	Ç
Medium	880	999.78	1,019.77		1,060.97		1,103.83	1,125.91			1,194.82		1,243.09	1,267.96	1,2
Large	1850	2101.05	2,143.07		2,229.65		2,319.73				2,510.95	2,561.17	2,612.39	2,664.64	2,1
-			_,	_,	-,	-,	_,	_,	_,	_,	_,	_, -,	-,	_,	-, -
Distributed Systems M															
Per Desktop	87.71	97.12	99.06	101.04	103.06	105.13	107.23	109.37	111.56	113.79		118.39	120.76	123.17	1
Incremental Email Box	9.26	10.35	10.56	10.77	10.98	11.20	11.43	11.66	11.89	12.13	12.37	12.62	12.87	13.13	
Information Systems R	esource Rate	es: (2), (3)												
Long-term:(5), (7)		. ,,													
Information Associate	9,400	9,776	10,167	10,574	10,997	11,437	11,894	12,370	12,865	13,379	13,914	14,471	15,050	15,652	1
Information Analyst	12,000	12,480	12,979	13,498	14,038	14,600	15,184	15,791	16,423	17,080		18,473	19,212	19,981	2
Information Specialis		14,664	15,251	15,861	16,495	17,155	17,841	18,555	19,297	20,069		21,706	22,575	23,478	2
Information Specialis		17,056	17,738	18,448	19,186	19,953	20,751	21,581	22,445	23,342		25,247	26,257	27,307	2
System Architect	17,800	18,512	19,252	20,023	20,823	21,656	22,523	23,424	24,361	25,335		27,402	28,498	29,638	3
DBA	17,800	18,512	19,252	20,023	20,823	21,656	22,523	23,424	24,361	25,335		27,402	28,498	29,638	3
Project Control Manag	er 17,800	18,512	19,252	20,023	20,823	21,656	22,523	23,424	24,361	25,335	26,348	27,402	28,498	29,638	5
Short-term: (4), (6)															
Information Associate	10,900	11,336	11,789	12,261	12,751	13,262	13,792	14,344	14,917	15,514	16,135	16,780	17,451	18,149	1
Information Analyst	13,900	14,456	15,034	15,636	16,261	16,911	17,588	18,291	19,023	19,784		21,398	22,254	23, 145	2
Information Specialis	t 16,400	17,056	17,738	18,448	19,186	19,953	20,751	21,581	22,445	23,342	24,276	25,247	26,257	27,307	2
Information Specialis	t Sr 19,100	19,864	20,659	21,485	22,344	23,238	24,168	25,134	26,140	27,185	28,273	29,404	30,580	31,803	
System Architect	20,600	21,424	22,281	23,172	24,099	25,063	26,066	27,108	28,193	29,320	30,493	31,713	32,981	34,301	1
DBA	20,600	21,424	22,281	23,172	24,099	25,063	26,066	27,108	28,193	29,320		31,713	32,981	34,301	3
Project Control Manag	er 20,600	21,424	22,281	23,172	24,099	25,063	26,066	27,108	28,193	29,320	30,493	31,713	32,981	34,301	3
E-Solutions Resource	Rates: (2)	(3)													
Long-term:(5), (7)	(=//	(-)													
Consultant Associate	14,535	15,116	15,721	16,350	17,004	17,684	18,391	19,127	19,892	20,688	21,515	22,376	23,271	24,202	2
Consultant	18,900	19,656	20,442	21,260	22,110	22,995	23,915	24,871	25,866	26,901		29,096	30,260	31,470	1
Consultant Specialist	23,985	24,944	25,942	26,980	28,059	29,181	30,349	31,563	32,825	34,138	35,504	36,924	38,401	39,937	4
Consultant Senior	28,345	29,479	30,658	31,884	33,160	34,486	35,865	37,300	38,792	40,344	41,958	43,636	45,381	47,197	4
Consultant Architect	32,700	34,008	35,368	36,783	38,254	39,785	41,376	43,031	44,752	46,542		50,340	52,354	54,448	É
Program Client Execut	ive 36,340	37,794	39,305	40,878	42,513	44,213	45,982	47,821	49,734	51,723	53,792	55,944	58,182	60,509	(
Short-term: (4), (6),	(8)														
Consultant Associate	100	104	108	112	117	122	127	132	137	142	148	154	160	167	
Consultant	130	135	141	146	152	158	164	171	178	185		200	208	216	
Consultant Specialist		172	178	186	193	201	209	217	226	235		254	264	275	
Consultant Senior	195	203	211	219	228	237	247	257	267	278	289	300	312	325	
Consultant Architect	225	234	243	253	263	274	285	296	308	320		346	360	375	
Program Client Execut	ive 250	260	270	281	292	304	316	329	342	356	370	385	400	416	

- (1) Subject to ECI inflation index pursuant to Section 9.1 (d) of this Schedule
 (2) Subject to Hewitt inflation index pursuant to Section 9.1 (d) of this Schedule
 (3) Person-month is defined as 130 hours / month and the rates do not include travel-related expenses
 (4) Short-term rates are for resources less than 6 months in duration
 (5) Long-term rates are for resources greater than 6 months in duration
 (6) Additional job classifications will be discounted 10% off EDS' Commercial Billing Rates
 (7) Additional job classification will be discounted 25% off EDS' Commercial Billing Rates
 (8) Short-term E.Solutions rates are hourly

Small / medium

Attachment 2 to Amendment 2

Termination for Convenience Charges for Years 11 - 15 of Agreement

Dates	Termination Fees
April 1, 2011 - March 31, 2012	The total Fees paid or payable by Franklin Covey to EDS for the three (3) calendar months immediately preceding the date on which EDS received Franklin Covey's notice of termination for convenience.
April 1, 2012 - March 31, 2013	The total Fees paid or payable by Franklin Covey to EDS for the three (3) calendar months immediately preceding the date on which EDS received Franklin Covey's notice of termination for convenience.
April 1, 2013 - March 31, 2014	The total Fees paid or payable by Franklin Covey to EDS for the two (2) calendar months immediately preceding the date on which EDS received Franklin Covey's notice of termination for convenience.
April 1, 2014 - March 31, 2015	The total Fees paid or payable by Franklin Covey to EDS for the one (1) calendar months immediately preceding the date on which EDS received Franklin Covey's notice of termination for convenience.
April 1, 2015 - June 30, 2016	The total Fees paid or payable by Franklin Covey to EDS for the one (1) calendar months immediately preceding the date on which EDS received Franklin Covey's notice of termination for convenience.

INTEREST TRANSFER AGREEMENT

THIS INTEREST TRANSFER AGREEMENT (this "Agreement") dated as of May 3, 2001, is by and between BANK ONE, NA ("Bank One") and FRANKLIN COVEY CO., a Utah corporation (the "Company").

RECITALS

- A. The Company and Bank One, individually and as Agent, (the "Agent") are party to that certain Facility Agreement dated as of March 27, 2000 (as amended, the "Facility Agreement"). Unless otherwise specified herein, capitalized terms used in this Agreement shall have the meanings ascribed to them by the Facility Agreement.
- B. It is the desire of the Company and Bank One to enter into an agreement whereby, in exchange for certain payments from the Company, Bank One will transfer to the Company, upon (and subject to) receipt, certain interest payments and, if applicable, Early Payment Fees received from the Borrowers. Such arrangement is intended to be an arrangement solely among the parties hereto and shall in no way affect the obligations of the Borrowers under the Loan Documents.

NOW THEREFORE, in consideration of the mutual covenants and undertakings herein contained, and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto hereby agree as follows:

ARTICLE I

DEFINITIONS

Capitalized terms used herein without definition shall have the meanings given them in the Facility Agreement and the following terms shall have the following meanings:

- 1.1 "Alternate Base Rate" means a fluctuating rate of interest equal to the higher of (a) the Corporate Base Rate and (b) the sum of the Federal Funds Effective Rate most recently determined by Bank One plus 1/2% per annum.
- 1.2 "Corporate Base Rate" means a rate per annum equal to the corporate base rate or prime rate of interest announced by Bank One or by its parent, Bank One Corporation, from time to time, changing when and as said corporate base rate or prime rate changes. The Corporate Base Rate is a reference rate and does not necessarily represent the lowest or best rate of interest at, above or below the Corporate Base Rate.
- 1.3 "Federal Funds Effective Rate" means, for any day, an interest rate per annum equal to the weighted average of the rates on overnight Federal funds transactions with members of the Federal Reserve System arranged by Federal funds brokers, as published for such day by the Federal Reserve Bank of New York, or if such rate is not so published for such day, the average of the quotations for such day on such transactions received by Bank One from three Federal funds brokers of recognized standing selected by it.
 - 1.4 "Fixed Rate" means a rate per annum equal to 10.30%.

ARTICLE II

TRANSFER OF INTEREST AND FEES

- 2.1 Payment for Transfer of Certain Interest and Fees. The Company shall pay to Bank One (i) upon execution of this Agreement a lump sum in the amount of \$1,435,652.39 and (ii) thereafter, an amount equal to the product of the Fixed Rate times the outstanding principal amount of each Loan for each day from May 8, 2001 to the date such Loan is paid in full. Such amounts set forth in (ii) above shall be payable quarterly in arrears on each Quarterly Date commencing June 30, 2001 and on the date such Loan is paid in whole or part and shall be calculated for actual days elapsed on the basis of a 360-day year. Such payments shall not discharge any amounts owing under any Note, all of which shall remain payable by the applicable Borrower in accordance with his or her Note.
- 2.2 <u>Transfer of Certain Interest and Fees.</u> In consideration of the foregoing payments, Bank One will transfer to the Company upon (and subject to) receipt thereof (i) interest on all Loans calculated at the Interest Rate pursuant to Section 1(a) of the Notes (and expressly excluding any interest on the Loans in excess of the Interest Rate, including, but not limited to, default rate interest paid on Loans pursuant to Section 1(b) of the Notes) and (ii) to the extent not duplicative of the amount paid pursuant to subsection (i), Early Payment Fees in respect of the Loans; <u>provided, however</u>, such transfers shall be net of any funding losses or costs incurred by the Lenders as a result of any voluntary or mandatory prepayment or acceleration of all or any portion of such Loans, including, without limitation, the amount of the Funding Indemnity (as defined in Section 2.5 herein). Since the Notes provide for no payment or interest until prepayment, acceleration or maturity, no payments will be made by Bank One until such time (i.e., absent prepayment or acceleration under a Note, it is anticipated that Bank One's payment to the Company hereunder will be due on March 30, 2005).
- 2.3 Nature of Transfer. The agreement of Bank One to transfer interest payments and Early Payment Fees to the Company under the Interest Transfer Agreement is not intended to and does not constitute a sale or assignment of Bank One's rights to receive such interest payments and Early Payment Fees under the Notes. The Company will acquire no interest in, and will have no rights to enforce any Note, Pledge Agreement or other Loan Documents against the applicable Borrower unless and until Bank One is paid in full thereunder, at which time Bank One will sell or assign its rights against such Borrower under the Loan Documents (excluding the Pledge Agreement of such Borrower) to the Company. The amounts paid by the Company pursuant to Section 2.1 are nonrefundable. The Company assumes all the risk of non-payment by the Borrowers and shall have no recourse against Bank One if any Borrower fails to pay interest or Early Termination Fees for any reason.

2.4 Method of Payment.

- (a) All payments owing to Bank One from the Company hereunder shall be made without setoff, deduction or counterclaim, in funds which are available to Bank One at Bank One's address at One Bank One Plaza, Chicago, Illinois 60670, or at any other office of Bank One specified in writing by Bank One to the Company, by 12:00 noon (Chicago time) on the date when due. Whenever any payment to be made by the Company hereunder shall be stated to be due on a day other than a Business Day, such payment shall be made on the next succeeding Business Day, and such extension of time shall in such case be included in the computation of payment of the amounts due under Section 2.1(ii).
- (b) All payments owing to the Company from Bank One hereunder shall be made in funds which are available to the Company at the Company's address at 2200 West Parkway Boulevard, Salt Lake City, Utah, 84119, or at any other place specified in writing by the Company to Bank One, by 12:00 noon (Chicago time) on the Business Day immediately following the day on which Bank One receives applicable payments from a Borrower.
- 2.5 <u>Funding Indemnity</u>. To the extent not set off against amounts transferred to the Company pursuant to Section 2.2, the Company shall reimburse Bank One on demand for any losses or expenses (including, without limitation, any loss incurred in obtaining, liquidating or employing deposits from third parties or in terminating or unwinding any interest rate exchange or similar arrangement) incurred by the Lenders as a result of any voluntary or mandatory prepayment, or upon or after the acceleration, of all or any portion of the Loans. Each amount payable pursuant to this Section is referred to as a "<u>Funding Indemnity</u>". With respect to the payment of a Funding Indemnity to Bank One only (and not with respect to any other Lender), the amount of such Funding Indemnity shall be calculated pursuant to the methodology set forth in Section 2.5 and Schedules 2.05(A) and 2.05(B) to the Facility Agreement, treating all amounts which have been accelerated as having been prepaid on the date of acceleration; <u>provided</u>, <u>however</u>, for purposes of this Agreement only, the "<u>Reference Fixed Rate</u>" will be assumed to be 10.30%.

ARTICLE III

COMPANY SUBORDINATION

3.1 Company Subordination. Notwithstanding Section 2.2 above, all interest payments and Early Payment Fees received by Bank One from a Borrower will be first applied to satisfy any amounts due and owing to Bank One from such Borrower under its Note and any other Loan Documents and, unless and until such amounts have been paid in full with respect to any Loan, Bank One shall not be obligated to make any payment to the Company in respect thereof pursuant to Section 2.2. Further, Bank One may set off amounts owing by Bank One hereunder against amounts then owing by the Company under the Facility Agreement.

ARTICLE IV

RATABLE COLLATERAL SECURITY

4.1 <u>Ratable Security.</u> In the event the obligations under the Existing Credit Agreement or the Obligations under the Facility Agreement become secured by assets or property (which collateral security is anticipated by the parties hereto to occur on a date following the date hereof) or are guaranteed by any Person, then the Company will cause the obligations under the Interest Transfer Agreement to be equally and ratably secured by such assets or property and equally and ratably guaranteed by such Person pursuant to documentation satisfactory to Bank One.

ARTICLE V

EVENTS OF DEFAULT

- $5.1\,$ Events of Default. Each of the following events shall constitute an "Event of Default":
- $(a) \quad \text{The Company shall fail to perform or observe any term, covenant or agreement set forth in Sections 2.1, 2.4 or 4.1;}\\$
- (b) The Company shall fail to perform or observe any other term, covenant or agreement contained in this Agreement (other than those referenced in Section 5.1(a)) if the failure to perform or observe such covenant or agreement set forth therein shall remain unremedied for five (5) days after written notice thereof shall have been given to the Company by Bank One;
- (c) Failure of the Company or any of its Subsidiaries to pay when due any other Indebtedness, including, without limitation, the Existing Credit Agreement or the Facility Agreement, or the default by the Company or any of its Subsidiaries in the performance (beyond the applicable grace period with respect thereto, if any) of any term, provision or condition contained in any agreement under which any such Indebtedness was created or is governed, or any other event shall occur or condition exist, the effect of which default or event is to cause, or to permit the holder or holders of such Indebtedness to become due prior to its stated maturity; or any Indebtedness of the Company or any of its Subsidiaries shall be declared to be due and payable or required to be prepaid or repurchased (other than by a regularly scheduled payment) prior to the stated maturity thereof; or the Company or any of its Subsidiaries shall not pay, or admit in writing its inability to pay, its debts generally as they become due;

- (d) Any representation or warranty made by or on behalf of the Company to Bank One under or in connection with this Agreement shall prove to have been incorrect in any material respect when made
- (e) The Company or any of its Subsidiaries shall (i) have an order for relief entered with respect to it under the Federal bankruptcy laws as now or hereafter in effect, (ii) make an assignment for the benefit of creditors, (iii) apply for, seek, consent to, or acquiesce in, the appointment of a receiver, custodian, trustee, examiner, liquidator or similar official for it or any Substantial Portion of its Property, (iv) institute any proceeding seeking an order for relief under the Federal bankruptcy laws as now or hereafter in effect or seeking to adjudicate it a bankrupt or insolvent, or seeking dissolution, winding up, liquidation, reorganization, arrangement, adjustment or composition of it or its debts under any law relating to bankruptcy, insolvency or reorganization or relief of debtors or fail to file an answer or other pleading denying the material allegations of any such proceeding filed against it, (v) take any corporate or partnership action to authorize or effect any of the foregoing actions set forth in this Section 5.1(e); or (vi) fail to contest in good faith any appointment or proceeding described in Section 5.1(f); or
- (f) Without the application, approval or consent of the Company or any of its Subsidiaries, a receiver, trustee, examiner, liquidator or similar official shall be appointed for the Company or any of its Subsidiaries or any Substantial Portion of its Property, or a proceeding described in Section 5.1(e)(iv) shall be instituted against the Company or any of its Subsidiaries and such appointment continues undischarged or such proceeding continues undismissed or unstayed for a period of 60 consecutive days.
 - 5.2 Remedies.
- (a) If any Event of Default described in Section 5.1 occurs and is continuing, Bank One may (i) by notice to the Company, terminate this Agreement (whereupon Bank One shall be relieved of its obligations under Section 2.2); provided, however, such notice shall not be required and this Agreement shall automatically terminate if an Event of Default described in Section 5.1(e) or (f) occurs, and/or (ii) enforce any and all rights and remedies existing under this Agreement (including without limitation, all rights of setoff) and under applicable law.
- (b) Following an Event of Default, the Company shall continue to owe and pay its obligations under this Agreement and shall pay Bank One simple interest on any past due amounts at the higher of (i) the Fixed Rate plus two percent (2%) per annum and (ii) the Alternate Base Rate plus two percent (2%) per annum.

ARTICLE VI

NOTICES

- 6.1 Giving Notice. All notices and other communications provided to any party hereto under this Agreement shall be in writing, by facsimile, first class U.S. mail or courier and addressed or delivered to such party at its address set forth below its signature hereto or at such other address as may be designated by such party in a notice to the other parties. Any notice, if mailed and properly addressed with first class postage prepaid, return receipt requested, shall be deemed given three Business Days after deposit in the U.S. mail; any notice, if transmitted by facsimile, shall be deemed given when transmitted if a confirmation of transmission to the addressee is then generated by the sender's fax machine (and a copy thereof is simultaneously posted in first class U.S. mail); and any notice given by courier shall be deemed given when received by the addressee.
 - 6.2 Change of Address. The Company and Bank One may each change the address for service of notice upon it by a notice in writing to the other party hereto.

ARTICLE VII

REPRESENTATIONS AND WARRANTIES OF THE COMPANY

The Company represents and warrants that:

- 7.1 <u>Validity and Enforceability</u>. The execution, delivery and performance by the Company of this Agreement have been duly authorized by all necessary corporate action and that this Agreement is a legal, valid and binding obligation of the Company, enforceable against the Company in accordance with its terms, except as the enforcement thereof may be subject to the effect of any applicable bankruptcy, insolvency, reorganization, moratorium or similar law affecting creditors' rights generally:
- 7.2 No Conflicts. The execution, delivery and performance by the Company of this Agreement do not and will not violate any law, rule, regulation, order, writ, judgment, decree or award applicable to the Company or any contractual provision to which the Company is a party or to which the Company or any of its properties are subject.

ARTICLE VIII

MISCELLANEOUS

- 8.1 Term. This Agreement shall become effective upon the effective date of Amendment No. 4 to Facility Agreement dated the date hereof by and among the Company, Bank One and Bank One as Agent and shall, unless earlier terminated by Bank One in accordance with Article V, terminate on the date Bank One has been paid in full all amounts due under the Loan Documents and all payments owing pursuant to Section 2.1 or 2.2 have been paid. The Company shall have no right to terminate this Agreement without the consent of Bank One.
- 8.2 <u>Further Assurances</u>. Upon any assignment of the Notes by Bank One in connection with the syndication of the Facility Agreement and Loans, the Company shall enter into such further agreements, if any, as Bank One may deem necessary to give continuing effect to the economics of this Agreement.
 - 8.3 <u>Amendments</u>. This Agreement may not be amended orally but only in writing signed by the parties hereto.
- 8.4 <u>Preservation of Rights: Survival</u>. No delay or omission of any party hereto to exercise any right under this Agreement shall impair such right or be construed to be a waiver of any default or an acquiescence therein. Any single or partial exercise of any such right shall not preclude other or further exercise thereof or the exercise of any other right.
 - 8.5 Headings: Entire Agreement. Section headings in this Agreement are for convenience of reference only, and shall not govern the interpretation of any of the provisions of this Agreement.
 - 8.6 Benefits of this Agreement. This Agreement shall inure to the benefit of Bank One and its respective successors and assigns. The Company shall not have the right to assign its rights or obligations hereunder.
- 8.7 Expenses; Indemnification. The Company agrees to reimburse Bank One for any costs, internal charges and out-of-pocket expenses (including reasonable attorneys' fees and time charges of attorneys for Bank One, which attorneys may be employees of Bank One) paid or incurred by Bank One in connection with the collection or enforcement of this Agreement. The Company further agrees to indemnify Bank One, its directors, officers and employees against all losses, claims, damages, penalties, judgments, liabilities and expenses (including, without limitation, all expenses of litigation or preparation therefor whether or not Bank One is a party thereto) which any of them may pay or incur arising out of or relating to this Agreement, or the transactions contemplated hereby except to the extent that they arise out of the gross negligence or willful misconduct of the party seeking indemnification. The obligations of the Company under this Section shall survive the termination of this Agreement.
- 8.8 <u>Severability of Provisions</u>. Any provision in this Agreement that is held to be inoperative, unenforceable, or invalid in any jurisdiction shall, as to that jurisdiction, be inoperative, unenforceable, or invalid without affecting the remaining provisions in that jurisdiction or the operation, enforceability, or validity of that provision in any other jurisdiction, and to this end the provisions of this Agreement are declared to be severable.
- 8.9 CHOICE OF LAW. THIS AGREEMENT SHALL BE CONSTRUED IN ACCORDANCE WITH THE INTERNAL LAWS (INCLUDING 735 ILCS 105/5-1 ET SEQ., BUT OTHERWISE, WITHOUT REGARD TO CONFLICT OF LAWS PROVISIONS) OF THE STATE OF ILLINOIS, BUT GIVING EFFECT TO FEDERAL LAWS APPLICABLE TO NATIONAL BANKING ASSOCIATIONS, FEDERAL AGENCIES, BRANCHES OF FOREIGN BANKS AND OTHER FINANCIAL INSTITUTIONS.
- 8.10 CONSENT TO JURISDICTION. THE COMPANY AND BANK ONE HEREBY IRREVOCABLY SUBMIT TO THE NON-EXCLUSIVE JURISDICTION OF ANY UNITED STATES FEDERAL OR ILLINOIS STATE COURT SITTING IN CHICAGO IN ANY ACTION OR PROCEEDING ARISING OUT OF OR RELATING TO THIS AGREEMENT AND THE PARTIES HEREBY IRREVOCABLY AGREE THAT ALL CLAIMS IN RESPECT OF SUCH ACTION OR PROCEEDING MAY BE HEARD AND DETERMINED IN ANY SUCH COURT AND IRREVOCABLY WAIVE ANY OBJECTION THEY MAY NOW OR HEREAFTER HAVE AS TO THE VENUE OF ANY SUCH SUIT, ACTION OR PROCEEDING BROUGHT IN SUCH A COURT OR THAT SUCH COURT IS AN INCONVENIENT FORUM. NOTHING HEREIN SHALL LIMIT THE RIGHT OF BANK ONE TO BRING PROCEEDINGS AGAINST THE COMPANY IN THE COURTS OF ANY OTHER JURISDICTION. ANY JUDICIAL PROCEEDING BY THE COMPANY AGAINST BANK ONE OR ANY AFFILIATE OF BANK ONE INVOLVING, DIRECTLY, OR INDIRECTLY, ANY MATTER IN ANY WAY ARISING OUT OF, RELATED TO, OR CONNECTED WITH THIS AGREEMENT SHALL BE BROUGHT ONLY IN A COURT IN CHICAGO, ILLINOIS.
- 8.11 WAIVER OF JURY TRIAL, THE COMPANY AND BANK ONE HEREBY WAIVE TRIAL BY JURY IN ANY JUDICIAL PROCEEDING INVOLVING, DIRECTLY OR INDIRECTLY, ANY MATTER (WHETHER SOUNDING IN TORT, CONTRACT OR OTHERWISE) IN ANY WAY ARISING OUT OF, RELATED TO, OR CONNECTED WITH THIS NOTE.

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be executed by their respective officers thereunto duly authorized, as of the date first above written.

FRANKLIN COVEY CO.		
Ву:		
Name:		
Title:		

Address: 2200 West Parkway Boulevard Salt Lake City, UT 84119 Attention: Val Christensen Telecopy: (801) 817-4291 Telephone: (801) 817-7102 Name: Title:

Address: 777 South Figueroa Street,4th Floor, IL1-4001
Los Angeles, CA 90017
Attention: Stephen M. Flynn
Telecopy: (213) 683-4999
Telephone: (213) 683-4932

BANK ONE, NA, individually

with a copy to:

Address: 1 Bank One Plaza Chicago, Illinois 60670 Attention: Chapin Bates Telecopy: (312) 732-5645 Telephone: (312) 732-8762

AMENDMENT NO. 4 TO FACILITY AGREEMENT

This Amendment (the "Amendment") is entered into as of May 3, 2001 by and among FRANKLIN COVEY CO., a Utah corporation (the "Company"), BANK ONE, NA, a national banking association with its principal office in Chicago, Illinois, individually ("Bank One") and as agent (the "Agent").

RECITALS

- A. The Company, the Agent and Bank One are party to that certain Facility and Guaranty Agreement dated as of March 27, 2000 (as previously amended the "Facility Agreement"). Unless otherwise specified herein, capitalized terms used in this Amendment shall have the meanings ascribed to them by the Facility Agreement.
- B. The Company, the Agent and Bank One (being the sole "Lender" under the Facility Agreement) wish to amend the Facility Agreement on the terms and conditions set forth below.
- C. The Company and Bank One are entering into an agreement (the "Interest Transfer Agreement") in connection with the Amendment whereby, in exchange for certain payments to Bank One by the Company, Bank One will transfer to the Company, upon (and subject to) receipt, certain interest payments and, if applicable, Early Payment Fees received from the Borrowers.

Now, therefore, in consideration of the mutual execution hereof and other good and valuable consideration, the parties hereto agree as follows:

- 1. Amendment to Facility Agreement. Upon the Effective Date (as defined below), the Facility Agreement shall be amended as follows:
 - (a) Section 1.01. Section 1.01 is amended by adding the following definitions in appropriate alphabetical order:
 - "Interest Transfer Agreement" means the Agreement between the Company and Bank One dated May 3, 2001 wherby, in exchange for certain payments to Bank One by the Company, B ank One will transfer to the Company, upon (and subject to) receipt, certain interest payments and Early Payment Fees received from the Borrowers, as such agreement may be from time to time amended.
 - (b) Section 6.01. Section 6.01 is amended by adding the following new subsection to the end of such Section:
 - "(m) The Company shall fail to pay amounts due under or shall default int he performance of any term, provision, or condition contained in the Interest Transfer Agreement or The Interest Transfer Agreement shall fail to remain in full force and effect or to give Bank One the rights, powers and privileges purported to be created thereby."
- 2. Reaffirmation of Guaranty. The Company hereby reaffirms its Obligations under the Facility Agreement, including its absolute, irrevocable and unconditional guaranty of prompt, full and complete payment when due, whether at stated maturity, upon acceleration or otherwise, and at all times thereafter, of the Guaranteed Debt which shall in any event include any and all principal, interest or any other amount which may be payable by any Borrower pursuant to the terms of his or her Note, as amended, without respect to the enforceability of such terms against any Borrower. The Company further agrees that if for any reason any acceleration of any Loan by the Lenders in accordance with the terms of the applicable Note is unenforceable against any Borrower, for purposes hereof the related Guaranteed Debt shall never the less be deemed due and payable by the Company and shall be forthwith paid by the Company and that, upon any acceleration of any Loan in accordance with the terms of the applicable Note (without respect to its enforceability against the applicable Borrower), the Early Payment Fee payable to Bank One with respect thereto shall, for purposes of this Guaranty be deemed Guaranteed Debt which is then immediately due and payable by the Company under the Guaranty, calculated as if the entire principal amount of such Loan had been paid as of the date of such acceleration. Moreover, the Company hereby reaffirms and agrees that (a) under no circumstance shall the Agent or the Lenders be obligated to foreclose upon the Pledged Shares of any Borrower or any other collateral from time to time securing the Guaranteed Debt prior to demanding payment under and enforcing the Guaranty and (b) the Agent and the Required Lenders may at any time in their sole discretion release any of the Pledged Shares of any Borrower, and no action or inaction of the Agent or the Lenders with respect to foreclosure upon, or release of, any Pledged Shares shall negate, release, or otherwise affect the Guaranty.
- 3. Representations and Warranties of the Company. The Company represents and warrants that:
 - (a) The execution, delivery and performance by the Company of this Amendment have been duly authorized by all necessary corporate action and that this Amendment is a legal, valid and binding obligation of the Company enforceable against the Company in accordance with its terms, except as the enforcement thereof may be subject to the effect of any applicable bankruptcy, insolvency, reorganization, moratorium or similar law affecting creditors' rights generally;
 - (b) Each of the representations and warranties contained in the Facility Agreement is true and correct in all material respects on and as of the date hereof as if made on the date hereof; and
 - (c) After giving effect to this Amendment, no Program Event of Default or Unmatured Default has occurred and is continuing.
- 4. Effective Date. This Amendment shall become effective upon (i) the execution and delivery hereof by the Company, the Agent and the Required Lenders and (ii) the execution and delivery by the Company and Bank One of the Interest Transfer Agreement; provided that the Company has delivered to Bank One and the Agent a certificate, executed by the Secretary or Assistant Secretary of the Company, which shall attach and certify copies of the Company's Board of Directors' resolutions authorizing the execution, deliver and performance of the Interest Transfer Agreement and the Facility Agreement as amended hereby.
- Reference to and Effect Upon the Facility Agreement.
 - (a) Except as specifically amended above, the Facility Agreement and the other Loan Documents shall remain in full force and effect and are hereby ratified and confirmed. The Company hereby reaffirms its obligations under the Facility Agreement and all of the other Loan Documents to which it is a party.
 - (b) The execution, delivery and effectiveness of this Amendment shall not operate as a waiver of any right, power or remedy of the Agent or any Lender under the Facility Agreement or any Loan Document, nor constitute a waiver of any provision of the Facility Agreement or any Loan Document, except as specifically set forth herein. Upon the effectiveness of this Amendment, each reference in the Facility Agreement to "this Agreement", "hereunder", "herein" or words of similar import shall mean and be a reference to the Facility Agreement as amended hereby.
- 6. GOVERNING LAW. THIS AMENDMENT SHALL BE GOVERNED BY AND CONSTRUED IN ACCORDANCE WITH THE INTERNAL LAWS (AS OPPOSED TO CONFLICTS OF LAWS PROVISIONS) OF THE STATE OF ILLINOIS BUT GIVING EFFECT TO FEDERAL LAWS APPLICABLE TO NATIONAL BANKS.
- 7. Headings. Section headings in this Amendment are included herein for convenience of reference only and shall not constitute a part of this Amendment for any other purposes.
- 7. Counterparts. This Amendment may be executed in any number of counterparts, each of which when so executed shall be deemed an original but all such counterparts shall constitute one and the same instrument. Delivery of an executed counterpart to this Amendment by facsimile transmission shall constitute delivery of a manually executed counterpart hereof.

[signature page follows]

 $IN\ WITNESS\ WHEREOF, the\ parties\ have\ executed\ this\ Amendment\ as\ of\ the\ date\ and\ year\ first\ above\ written.$

FRANKLIN COVEY CO.
By: J. SCOTT NIELSEN
Title: Senior Vice President - Finance
Date: May 3, 2001
BANK ONE, NA, individually and as Agent
By: STEPHEN M. FLYNN
Title: First Vice President
Date: May 3, 2001